

FRANCHISE DISCLOSURE DOCUMENT

USL SUPER LEAGUE, LLC

A Florida Limited Liability Company 1715 N. Westshore Blvd. Suite 825 Tampa, Florida 33607 (813) 963-3909 www.uslsoccer.com/usl-super-league

You will operate a team in the women's professional soccer league known as "USL Super League" under a Franchise Agreement with us.

The total initial investment necessary to begin operation of a Club is set forth below, and the amounts set forth adjacent to the total initial investment below must be paid to franchisor or its affiliates:

<u>Total Initial Investment</u>	<u>Portion of Initial Investment Paid to</u> <u>Franchisor or its Affiliates</u>	<u>Franchise Fee</u>
\$13,677,450-\$18,771,900	\$10,377,700-\$10,461,900	\$10,000,000

This Franchise Disclosure Document summarizes certain provisions of your franchise agreement and other information in plain English. Read this document and all accompanying agreements carefully. You must receive this Franchise Disclosure Document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Note, however, that no governmental agency has certified the information contained in this document.

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact USL Legal Department legal@uslsoccer.com, 1715 N. Westshore Blvd., Suite 825, Tampa, FL 33607, (813) 963-3909.

The terms of your contract will govern your franchise relationship. Do not rely on the Franchise Disclosure Document alone to understand your contract. Read the entire contract carefully. Show your contract and this Franchise Disclosure Document to an advisor, like a lawyer or accountant.

Buying a franchise is a complex investment. The information in this Franchise Disclosure Document can help you make up your mind. More information on franchising, such as "<u>A Consumer's Guide to Buying a Franchise</u>," which can help you understand how to use this Franchise Disclosure Document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information of franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: February 15, 2023

STATE COVER PAGE

Your state may have a franchise law that requires a franchisor to register or file with a state franchise administrator before offering or selling in your state. REGISTRATION OF A FRANCHISE BY A STATE DOES NOT MEAN THAT THE STATE RECOMMENDS THE FRANCHISE OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT.

MANY FRANCHISE AGREEMENTS DO NOT ALLOW YOU TO RENEW UNCONDITIONALLY AFTER THE INITIAL TERM EXPIRES. YOU MAY HAVE TO SIGN A NEW AGREEMENT WITH DIFFERENT TERMS AND CONDITIONS IN ORDER TO CONTINUE TO OPERATE YOUR BUSINESS. BEFORE YOU BUY, CONSIDER WHAT RIGHTS YOU HAVE TO RENEW YOUR FRANCHISE, IF ANY, AND WHAT TERMS YOU MIGHT HAVE TO ACCEPT IN ORDER TO RENEW.

Call the state franchise administrator listed in <u>Exhibit B</u> for information about the franchisor, or about franchising in your state.

Please consider the following RISK FACTORS before you buy this franchise:

- 1. THE FRANCHISE AGREEMENT REQUIRES YOU TO RESOLVE DISPUTES WITH US BY ARBITRATION IN TAMPA, FLORIDA. OUT OF STATE ARBITRATION MAY FORCE YOU TO ACCEPT A LESS FAVORABLE SETTLEMENT FOR DISPUTES. IT ALSO MAY COST MORE TO ARBITRATE IN TAMPA, FLORIDA, THAN IN YOUR HOME STATE.
- 2. THE FRANCHISE AGREEMENT STATES THAT FLORIDA LAW GOVERNS THE AGREEMENT, AND THIS LAW MAY NOT PROVIDE THE SAME PROTECTION AND BENEFITS OF OTHER CHOICES OF LAW. YOU MAY WANT TO REVIEW THESE LAWS AND COMPARE THEM TO YOUR LOCAL AND OTHER SOURCES OF LAW.
- 3. YOU SHALL BE GRANTED PROTECTED TERRITORIAL RIGHTS COVERING A SPECIFIC GEOGRAPHIC AREA WITHIN WHICH ONLY YOU MAY PLAY HOME MATCHES. HOWEVER, YOU MAY FACE COMPETITION IN YOUR PROTECTED TERRITORY FROM OTHER FRANCHISES, TEAMS WE MAY OWN, OR OTHER LEAGUES WE MAY CONTROL.
- 4. THE FRANCHISE AGREEMENT REQUIRES YOU TO WAIVE YOUR RIGHT TO A JURY TRIAL. YOU MAY WANT TO CONSIDER THIS WHEN MAKING A DECISION TO PURCHASE THIS FRANCHISE OPPORTUNITY.
- 5. THERE MAY BE OTHER RISKS CONCERNING THIS FRANCHISE. PLEASE CONSULT AN ATTORNEY AND OTHER RELEVANT ADVISORS TO DISCUSS OTHER POTENTIAL RISKS ASSOCIATED WITH THIS INVESTMENT.
- 6. YOU MUST MAINTAIN MINIMUM SALES PERFORMANCE LEVELS. YOUR INABILITY TO MAINTAIN THESE LEVELS MAY RESULT IN LOSS OF ANY TERRITORIAL RIGHTS YOU ARE GRANTED, TERMINATION OF YOUR FRANCHISE, AND LOSS OF YOUR INVESTMENT.

STATE EFFECTIVE DATES

The following states require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This Franchise Disclosure Document is registered, on file or exempt from registration in the following states having franchise registration and disclosure laws, with the following effective dates:

STATE	EFFECTIVE DATE
California	
Illinois	
Indiana	
Maryland	
Michigan	
Minnesota	
New York	
Rhode Island	
Virginia	
Washington	
Wisconsin	

|--|

ITEM	PAGE
Item 1 – The Franchisor and any Parents, Predecessors, and Affiliates	1
Item 2 – Business Experience	4
Item 3 – Litigation	5
Item 4 – Bankruptcy	5
Item 5 – Initial Fees	7
Item 6 – Other Fees	8
Item 7 – Estimated Initial Investment	12
Item 8 – Restrictions on Sources of Products and Services	19
Item 9 – Franchisee's Obligations	21
Item 10 – Financing	23
Item 11 – Franchisor's Assistance, Advertising, Computer Systems, and Training	24
Item 12 – Territory	
Item 13 – Trademarks	
Item 14 – Patents, Copyrights, and Proprietary Information	
Item 15 – Obligation to Participate in the Actual Operation of the Franchise Business	
Item 16 – Restrictions on What the Franchise May Sell	34
Item 17 – Renewal, Termination, Transfer and Dispute Resolution	
Item 18 – Public Figures	41
Item 19 – Financial Performance Representations	42
Item 20 – Outlets and Franchisee Information	43
Item 21 – Financial Statements	49
Item 22 – Contracts	
Item 23 – Receipts	51

EXHIBITS

- A-1. Form of Franchise Agreement
- A-2. Form of Owner Guaranty
- A-3. Form of Lease Addendum
- B. Director of State Administrators & Agencies
- C. Agent for Service of Process
- D. List of Current Franchisees
- E. List of Franchisees Who Have Left the System
- F-1. Financial Statements
- F-2. Guaranty
- G. State Addenda to Franchise Disclosure Document
 - G.1 California
 - G.2 Illinois
 - G.3 Minnesota
 - G.4 New York
 - G.5 North Dakota
 - G.6 Rhode Island
 - G.7 Washington
- H. Manuals Tables of Contents
- I. Receipt

ITEM 1 – THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

The franchisor is USL Super League, LLC, a Florida limited liability company. The name used to conduct business is USL Super League. To simplify the language in this Franchise Disclosure Document, USL Super League, LLC d/b/a USL Super League is referred to as "Franchisor," "we" or "us." United Soccer Leagues, LLC, a Florida limited liability company, is our sole owner. To simplify the language in this Franchise Disclosure Document, United Soccer Leagues, LLC is referred to as our "parent" or our "parent company." Our and our parent's principal business address is 1715 N. Westshore Blvd., Suite 825, Tampa, Florida 33607.

This Disclosure Document will refer to the person or entity that buys the franchise from us as "you" or "your", and the term includes your partners if you are a partnership, your members if you are a limited liability company, or your shareholders if you are a corporation. If you are a corporation, partnership or limited liability company, certain of your owners will have to guarantee your obligations and be obligated to comply with the terms of the franchise agreement and ancillary documents described in this Disclosure Document.

Franchisor and its Parent

Our parent company was originally formed as a Delaware corporation known as USISL, Inc. on April 1, 1996 to operate a soccer league known as the USISL. On July 1, 1997, Umbro International, Inc. acquired a majority of the shares of capital stock in USISL, Inc. By April 11, 2008, Umbro International, Inc. had acquired all of the issued and outstanding capital stock of USISL, Inc. Nike, Inc. acquired Umbro International, Inc. as of May 31, 2008. As a result of that transaction, USISL, Inc. was a wholly-owned subsidiary of NIKE International Holding, Inc., which is a wholly-owned subsidiary of Nike, Inc. On August 27, 2009, NuRock Soccer Holdings LLC, a Georgia limited liability company ("<u>NuRock</u>"), acquired all of the issued and outstanding capital stock of USISL, Inc. Was converted into a Georgia limited liability company on October 1, 2009 and simultaneously changed its legal name to United Soccer Leagues, LLC. Our parent company is the same legal entity as USISL, Inc. NuRock remains the majority owner of our parent company and its principal business address is 800 North Point Parkway, Suite 12, Alpharetta, Georgia 30005.

We operate the women's professional league known as USL Super League (the "League"), and our parent company has licensed to us any assets used (or to be used) in the League's operation. Our parent began offering professional soccer franchises generally on August 19, 1996 and our parent and/or its subsidiaries have been continuously been engaged in such line of business since that time. Prior to August 19, 1996, neither we nor our parent offered franchises for the same or similar type of business that is to be operated by you. We do not offer franchises in any other lines of business; however, we reserve the right to do so in the future.

Predecessors and Subsidiaries

We have subsidiaries that currently operate multiple soccer leagues, from the youth to the professional level, and related businesses. Our parent created these subsidiaries to operate various portions of the business previously operated directly by our parent. Specifically, these subsidiaries are:

- USL Pro, LLC, which owns, operates, coordinates, and sells franchises for teams to participate in the men's professional soccer league known as "USL Championship."
- USL Pro-2, LLC, which owns, operates, coordinates, and sells franchises for teams to participate in the men's professional soccer league known as "USL League One."

- Premier Development League, LLC, which owns, operates, coordinates, and sells franchises for teams to participate in, the men's elite amateur soccer League currently known as the "Premier Development League" or "PDL."
- USL W League, LLC, which owns, operates, coordinates, and sells franchises for teams to participate in, the women's elite amateur soccer League currently known as the "USL W League."
- Super Y League, LLC, which owns, operates, and coordinates an elite youth soccer membership league currently known as the "Super Y League." This company does not sell franchises.
- USL Productions, LLC, which owns the rights to, produces, and disseminates the broadcasts of all League matches and ancillary programming in any form of media and the related content. This company does not sell franchises.

Except as described above, we do not have any predecessors or affiliates that currently offer, or offered in the past, franchises for the same type of business that is to be operated by you or in any other line of business. We do not have any predecessors or affiliates that provide products or services to our franchisees; however, we reserve the right to have them do so in the future.

All leagues (including the League) owned and operated by our affiliates are referred to in this document as the "System."

Agent for Service of Process

Our agent for service of process for your state, if any, is identified in <u>Exhibit C</u> to this Franchise Disclosure Document.

The Franchise

We sell and grant franchises to people like you to field a team (your "<u>Club</u>") that participates in the League. We do not currently own any team that participates in the League, but we may in the future. Our primary purpose is to operate and coordinate the League. We do not engage in business activities that are unrelated to the League.

As of the Issuance Date of this Franchise Disclosure Document, the League has not been sanctioned as a Women's Professional Outdoor Division 1 league by the United States Soccer Federation ("<u>USSF</u>"). The League will apply for sanctioning for the 2024-25 season. USSF may elect not to sanction or approve the League at any time without our input or approval. In addition, we may determine not to continue the foregoing affiliation in the future, to seek affiliations with other organizations, and/or to seek sanctioning of a different level or division within those organizations.

The League's current season for matches is from approximately September through May of each year, with a break in December and early January, and your Club would begin participating in the League in the 2024-25 season (or later, if applicable).

In operating your Club, you must comply with our Minimum Standards for matches and our league rules, which are contained in our manuals (the "<u>Manuals</u>"), copies of which we will loan to you following your purchase of a franchise. You must also comply with the standards, rules and regulations of USSF, as the governing body for soccer in the United States, and any other organization responsible for governing or overseeing the League, which standards may change without notice or approval from us. USSF's standards

currently relate to such topics as: ownership composition; ownership net worth; stadium seating capacity requirements; financial reporting; field dimensions; field surfaces; and personnel. If your Club competes outside the United States, it may be subject to the rules and regulations of governing bodies for professional or amateur soccer leagues in those jurisdictions.

You will be required to meet the Minimum Performance Standard for Local Revenues and turnstile attendance. You will be considered not to have met the Minimum Performance Standard if your (a) Local Revenues or (b) turnstile attendance at League Matches do not exceed half of the median local revenues or median turnstile attendance of the other team operators in the League. If you do not meet the Minimum Performance Standard for a period of two consecutive Fiscal Years, you will be required to create and present to us a plan to meet the Minimum Performance Standard. Throughout the remainder of the Term, if you do not meet the Minimum Performance Standard in any two consecutive years, we may require you to transfer your franchise. If you are unable to transfer your franchise, we may terminate the franchise agreement.

Market and Competition

You will compete in the general market for, among other things, live viewers, attendees of your franchise's events, merchandise, and sponsorships. Your competitors in these markets include other sporting organizations and forms of live entertainment in your geographic area. Specifically within the sport of soccer, your competitors may include Major League Soccer (MLS), the National Women's Soccer League (NWSL), the North American Soccer League (NASL), the National Independent Soccer League (NISA), USL League Two, operated by our affiliate as described above, the USL Championship , operated by our affiliate as described above, the USL Championship , operated by our affiliate as described above, the National Premier Soccer League (NPSL), and various other regional and local amateur soccer leagues, and any professional leagues that may exist now or in the future.

Board of Governors

With certain exceptions specified in your franchise agreement, the Board of Governors is made up of one representative from each Club that is unconditionally obligated to play in an upcoming season. Your franchise agreement sets forth the authority of the Board of Governors and requires you to abide by the Board of Governors' decisions.

ITEM 2 – BUSINESS EXPERIENCE

Chairman of the Board: Robert Hoskins

Mr. Hoskins has served as the chairman of our parent's board of directors since September 2, 2009 and our chairman since January 1, 2017. Mr. Hoskins is also one of our parent's minority owners and the part owner and Chairman of NuRock, our parent's majority owner. Mr. Hoskins is also the co-founder of The NuRock Companies, a family of companies that focuses on development, management and sports. Mr. Hoskins serves in his current capacity in Alpharetta, Georgia.

Director and Chief Executive Officer: Alec Papadakis

Mr. Papadakis has served as a director of our parent since September 2, 2009, our parent's Chief Executive Officer since September 2, 2009 and our Chief Executive Officer since January 1, 2017. He has previously served on the Appeal Committee of the United States Soccer Federation and has also served on its Board of Directors from February 2015 to February 2017. Mr. Papadakis serves in his current capacity in Tampa, Florida.

Director, Chief Operating Officer, and Chief Real Estate Officer: Justin Papadakis

Papadakis has served as a director, Chief Operating Officer, and Chief Real Estate Officer of our parent since February 2018. Mr. Papadakis serves in his current capacity in Tampa, Florida.

President: Jake Edwards

Jake Edwards has acted as our parent's President since March of 2015 and our President since January of 2020. Mr. Edwards worked as our parent's Vice President, USL Championship, and subsequently as our parent's Executive Vice President, since June of 2013.

ITEM 3 – LITIGATION

1. Ricky Hill v. SLSG Pro, LLC (f/d/b/a St. Louis FC) and United Soccer Leagues, LLC, United States District Court for the N.D. Illinois, Case No. 1:21-CV-06278

On or around January 10, 2022, United Soccer Leagues, LLC ("USL") was served with the lawsuit listed aboveHill alleged in the charge that (i) he had applied for a head coaching position with Saint Louis FC during various occasions and (ii) that Saint Louis FC discriminated against him by hiring other white and non-black applicants for those positions. USL has denied the allegations, on the grounds that it does not make hiring/firing decisions for its clubs, including Saint Louis FC, and has no knowledge about the specific allegations.

2. Ricky Hill v. AMB Sports & Entertainment, LLC, et al., United States District Court for the N.D. Illinois, Case No. 1:22-cv-02961

On or around July 26, 2022, United Soccer Leagues, LLC ("USL") was served with the lawsuit listed above. Similar to Case No. 1:21-CV-06278 (referenced above), Mr. Hill alleged in the charge that; (i) he had applied for a head coaching position at various clubs within the USL Championship and Major League Soccer (ii) the clubs discriminated against him by hiring other white and non-black applicants for those positions; and (iii) that USL and Major League Soccer have tacitly fostered or otherwise failed to prevent discriminatory conduct. As of the date of this update, USL has filed a Motion to Dismiss (currently pending) but has not yet filed an Answer.

3. Ricky Hill v. Las Vegas Lights and United Soccer Leagues, LLC

On or around November 16, 2021, United Soccer Leagues, LLC ("USL") received notice that an EEOC charge had been filed against it by a man named Ricky Hill. Mr. Hill alleged in the charge that (i) he had applied for a head coaching position with the Las Vegas Lights during various occasions; (ii) that Las Vegas discriminated against him by hiring "other white and non-black applicants for those positions, each of whom has been objectively less qualified than Mr. Hill." USL denied the allegations, on the grounds that it does not make hiring/firing decisions for its clubs, including the Las Vegas Lights, and has no knowledge about the specific allegations. On or around March 7, 2022, the EEOC terminated its processing of the above-referenced charges.

ITEM 4 – BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5 – INITIAL FEES

To obtain a franchise, you must pay a one-time expansion fee of \$10,000,000. The expansion fee is not refundable.

You are required to secure your payments and other obligations by providing us with a cash deposit or letter of credit (the "Letter of Credit") in the amount of \$150,000. The Letter of Credit is generally due on or before June 15th preceding the applicable season. However, if a governing body (like the United States Soccer Federation) requires the Letter of Credit to be in a different amount than set forth above, you are required to abide by the requirements of that organization. As a result, the actual amount of the Letter of Credit you are required to provide may be different than as described above. In such case that a governing body increases these standards, you will have to provide to us the amount of the revised Letter of Credit within fifteen (15) days of written notice of the new amount, at which time we will return the original Letter of Credit to you.

Assuming the team's first scheduled season was in 2024-25, you would also pay a participation fee of 200,000 and between \$177,700 and \$261,900 to Franchisor in pass-through costs, as further set forth in Item 7 of this FDD.

Additionally, we reserve the right at the end of the term of your franchise agreement, or if we reasonably decide it is necessary to protect us or the League, to change the Letter of Credit amount to one we deem acceptable. If you renew your franchise agreement, you will be required to acquire a new Letter of Credit in the amount we designate for the required period.

We may draw on the Letter of Credit upon your failure to pay any obligations incurred in connection with the operation of your Club, including: (a) any amounts owed to us under your franchise agreement or any fees, fines or other amounts set forth in the Manual and/or with respect to violations of our rules or minimum standards; and (b) any amounts owed to any other team, League vendors, professional players, or landlords of your stadium.

If we draw on the Letter of Credit, we will take only the amount necessary to satisfy the delinquent obligation (which may include any associated interest and late fees and our costs and expenses incurred in satisfying such obligation), and you will replenish the amount of the Letter of Credit to the amount specified above within thirty (30) days of the date of our draw. We will return the Letter of Credit (or whatever portion of the Letter of Credit is remaining) to you within thirty (30) days following the expiration or termination of your franchise agreement.

All of your direct and indirect owners of greater than 5% are required to execute a guaranty, indemnification and acknowledgment of the franchisee's obligations under the franchise agreement in the form provided in Exhibit A-2.

ITEM 6 – OTHER FEES

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS (Note 1)
Participation Fee	\$200,000, (if starting in 2024-25), subject to a maximum annual increase of 15% or 12% plus CPI	October 15 and January 15 [split evenly] preceding each season during the Term of the applicable franchise agreement	Notes 2 and 3
Change of Control Fee	10% of then-current published expansion fee for the League	Upon closing of the transfer	Fee not applicable to transfers to an entity wholly-owned by a franchisee who is an individual, to family members, other owners of franchisee or in connection with the death of franchisee or its owners
Annual USSF Membership Fee	Variable	On demand	The League's annual membership fee to USSF is passed through equally to each team. As of 2022, the Division 1 league membership fee is \$350,000 (which is divided equally amongst participating teams)
USSF Player Registration Fee	\$1,333	On demand	Each team is required to pay an annual player registration fee to USSF. We pay this to USSF and require you to reimburse us. As of 2022, the Team Fee for Player Registration charged by USSF is \$1,333
Referees	Variable	On demand	Each team is required to pay an equal amount towards referees

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS (Note 1)
Reporting Software	Variable	On demand	You are required to use certain software to report operational and financial information. We pay this to providers and you reimburse us.
Website Hosting and Maintenance	Variable	On demand	You are required to pay us an annual fee for hosting and maintaining the team website
Website Re-Design	Variable	On demand	Only if you request the redesign
Insurance Premiums	Variable	On demand	We may require you to purchase certain forms of insurance, which may be as part of a group plan with other teams in the League.
Product Purchases	See ITEM 8	See ITEM 8	We may require you to buy or lease products or services that (i) meet our standards and specifications and (ii) are purchased from suppliers designated or approved by us. We reserve the right to designate ourselves or affiliates as the exclusive supplier for certain products. If we do so, then you must purchase them from us.
Pool Payments	Variable	On demand	We may require you to make payments that are pooled together amongst the teams for a particular purpose (e.g., travel pool), and we would distribute such funds in accordance with the parameters established by the Board of Governors.

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS (Note 1)
Board of Governor Approved Payments	Variable	On demand	You may be required to make certain payments required by the Board of Governors, even if such payments would otherwise not be permissible under your agreement
Operational Fines	Variable	As incurred	You must pay us any fines we assess for non- compliance with the franchise agreement or the Manuals. We reserve all other rights and remedies
Relocation Fee	10% of then-current published expansion fee for the League	Upon approval of relocation	This amount would be payable to us and any other franchises affected by such relocation
Indemnification	Variable	As incurred	You must reimburse us for damages or other relief related to the operation of your franchise
Costs and Attorney's Fees	Variable	As incurred	Excluding indemnification, payable to the prevailing party in any litigation
Interest and Late Fees	1.5% per month, or the maximum rate permitted by law, whichever is less (Note 2), plus a late fee of 5% of overdue amount	On demand	Payable on any amounts not paid when due. Note 4.

Note: (1) All fees are payable to us and, except as described in Note (2) below, are non-refundable.

(2) Should we disband the League prior to or during a season (as further described in <u>Item 17</u> below and Section 11.2.4 of your franchise agreement), you will receive a refund of any participation fees paid for any season in which your Club did not have an opportunity to participate.

(3) Subject to the authority of the Board of Governors (see ITEM 1), the maximum participation fee increase from year to year is generally limited to 15% or 12% plus CPI of the prior year's participation fee.

(4) For franchises in California, the maximum interest rate allowed is ten percent (10%) annually.

ITEM 7 – ESTIMATED INITIAL INVESTMENT

TYPE OF	AMOUNT (Note 1)		METHOD OF		то whom
EXPENDITURE	Low End	High End	PAYMENT	WHEN DUE	PAYMENT IS TO BE MADE
Expansion Fee	\$10,000,000	\$10,000,000	Wire transfer	At Signing of Franchise Agreement	Franchisor
Participation Fee (Note 1)	\$200,000	\$200,000	Wire transfer or check	January 15, 2024 and May 15, 2024 [split into equal installments]	Franchisor
Letter of Credit (Note 2)	\$12,500	\$150,000	Letter of credit, cash or cashier's check	No later than June 15, 2024	Bank (if letter of credit) or Franchisor (if cash to be held)
Stadium Lease (Note 3)	\$100,000	\$300,000	As required in Lease	As required in Lease	Landlord
Stadium and Field Improvements	\$500,000	\$1,500,000	As required by Suppliers	As required by Suppliers	Suppliers
Practice Field Rental (Note 4)	\$0	\$35,000	As required in rental agreement	As required in rental agreement	Landlord
Office Lease	\$50,000	\$100,000	As required in Lease	As required in Lease	Landlord
Office Expenses	\$25,000	\$75,000	As required by Suppliers	As required by Suppliers	Suppliers
Soccer Equipment/Field Related Supplies (Note 5)	\$25,000	\$50,000	As required by Suppliers	As required by Suppliers	Suppliers
Player Wages (Note 6)	\$400,000	\$800,000	As required in the player contracts	As required in the player contracts	Players
Player Housing (Note 7)	\$85,000	\$250,000	As required by Suppliers	As incurred	Apartment Lessors, Players
Front Office Staff (Note 8)	\$600,000	\$950,000	As required in staff contracts	As required in staff contracts	Staff

TYPE OF	AMOUNT (Note 1)		METHOD OF		TO WHOM
EXPENDITURE	Low End	High End	PAYMENT	WHEN DUE	PAYMENT IS TO BE MADE
Coaching Staff (Note 9)	\$175,000	\$350,000	As required in staff contracts	As required in staff contracts	Staff
Worker's Compensation Insurance (Note 10)	\$125,000	\$300,000	As required by insurance carrier	As required by insurance carrier	Insurance carriers
Benefits (Note 11)	\$116,250	\$390,000	As required in staff contracts	As required in staff contracts	Staff / Gov't
Referees (Note 12)	\$125,000	\$175,000	Wire transfer or check	Prior to start of season	Franchisor (pass- through)
Home Game Costs (Operations) (Note 13)	\$250,000	\$500,000	As required by Suppliers	At or before game	Suppliers
Away Game Costs (Note 14)	\$100,000	\$300,000	As required by Suppliers	Prior to or during travel	Suppliers
Marketing/Sales Costs (Note 15)	\$250,000	\$500,000	As required by Suppliers	As incurred	Suppliers
Uniforms and Supplies (Note 16)	\$20,000	\$35,000	As required by Suppliers	As incurred	Suppliers
USSF Membership Fee (Note 17)	\$7,100	\$17,800	As required by USSF	Prior to start of season	Franchisor (pass- through)
USSF Player Registration Fee (Note 17)	\$1,000	\$2,000	As required by USSF	Prior to start of season	Franchisor (pass- through)
U.S. Open Cup Application Fee (Note 17)	N/A	N/A	As required by USSF	Prior to start of season	Franchisor (pass- through)
Insurance (Note 18)	\$5,000	\$10,000	As required by insurance carrier	As required by insurance carrier	Franchisor (pass- through)
Travel Pool - Playoffs (Note 19)	\$5,000	\$10,000	Wire transfer or check	Prior to Start of Season	Franchisor (pass- through)

TYPE OF	AMOUNT (Note 1)		METHOD OF		TO WHOM
EXPENDITURE	Low End	High End	PAYMENT	WHEN DUE	PAYMENT IS TO BE MADE
Reporting Costs (Note 20)	\$1,500	\$3,000	Wire transfer or check	Prior to Start of Season	Franchisor
BMI / ASCAP Costs	\$500	\$1,500	As required by Suppliers	Prior to Start of Season	Franchisor (pass- through)
Local Broadcast Production Costs (Note 21)	\$100,000	\$200,000	As required by Suppliers	As incurred	Suppliers
Broadcast Transmission and Archival	\$25,000	\$35,000	Wire transfer or check	As incurred	Franchisor (pass- through)
Attendance at AGM and Training (Note 22)	\$1,000	\$5,000	As required by Suppliers	As required by Suppliers	Suppliers
Team Name and Trademark Design and Registration	\$5,000	\$30,000	As required by Suppliers	As incurred	Suppliers
Signage System	\$25,000	\$250,000	As required by Suppliers	As incurred	Suppliers
Professional Fees	\$50,000	\$150,000	As required by Suppliers	As incurred	Suppliers
Initial Merchandise Inventory	\$25,000	\$75,000	As required by Suppliers	As incurred	Suppliers
Website Design and Setup (Note 23)	\$6,000	\$6,000	Wire transfer or check	As incurred	Franchisor (pass- through)
Website Hosting and Maintenance (Note 24)	\$1,600	\$1,600	Wire transfer or check	April 1 each season	Franchisor (pass- through)
Dues and Subscriptions	\$10,000	\$15,000	As required by Suppliers	As incurred	Suppliers
Additional Funds for Year 1 (Note 25)	\$250,000	\$1,000,000	As required by Suppliers	As incurred	Suppliers

TYPE OF	AMOUNT (Note 1)		ΜΕΤΗΟΡ ΟΕ		то whom
EXPENDITURE	Low End	High End	- METHOD OF PAYMENT	WHEN DUE	PAYMENT IS TO BE MADE
TOTAL FIRST YEAR EXPENSES	\$13,677,450	\$18,771,900			

Notes:

- (1) All fees described in this <u>Item 7</u> that are payable to us are non-refundable, other than (a) the Participation Fee for a particular season, which may be refunded to you if there are not at least 6 teams participating in your League during that season or we disband your League after the 2024-25 season (as described in <u>Item 6</u> (Note 2) above), and (b) the Letter of Credit, which we are holding to secure your obligations under the franchise agreement and which will be returned to you upon expiration or termination of your franchise agreement (unless it is terminated as described in <u>Item 5</u> above).
- (2) This amount may change based upon the requirements of any governing body (namely, the United States Soccer Federation and/or the Canadian Soccer Association, CONCACAF, and FIFA), as described in <u>Item 5</u> above. In 2023, the Letter of Credit amount is one hundred and fifty thousand dollars (\$150,000). However, we do accept letters of credit, and the cost of obtaining a letter of credit is typically between one and eight percent (1% 8%) annually of the face amount of the letter of credit. The Letter of Credit is used to secure your payments and other obligations under your franchise agreement and other agreements entered into in connection with the operation of your Club. We may draw upon the Letter of Credit at such times and in such amounts as described in <u>Item 5</u> above and in Section 4.8 of your franchise agreement. If we draw upon your Letter of Credit, you must replenish it to the required amount. We will return the Letter of Credit (or whatever portion of the Letter of Credit is remaining) to you within thirty (30) days following the expiration or termination of your franchise agreement.
- (3) You must lease or own a stadium, which must be within your protected territory and approved by us in advance. The stadium must comply with all League Rules. Some of the main requirements currently included in League Rules are: (i) the stadium must be enclosed; (ii) the playing field must be at least 110x70 yards; (iii) the playing field must either be grass or artificial turf certified as FIFA Quality; (iv) stadium capacity of 5,000; (v) field lighting of at least 125 foot candles; (vi) locker rooms with showers within a one (1) minute walk of the field; and (vii) a press box. The current set of League Rules relating to stadium requirements will be made available to you upon request. We may inspect your stadium from time to time to ensure compliance with League Rules. Stadium rent varies depending on a number of factors such as size, amenities, location, whether renovations are necessary, included services, and any revenue-sharing arrangements.
- (4) You are required to have access to a training field for use by the visiting team upon request. The surface of the training field must match the surface of the game field. The training field is viewed as an extension of your Home Stadium, and you must make security available upon request of the visiting team.

- (5) This amount does not include the cost of a field-level signage system, which is included separately.
- (6) You determine the amount of payment to players in accordance with League regulations, but they typically range from \$20,000 to \$80,000 per player per season.
- (7) Teams typically, and may be required to, provide housing for their players while those players are training with the team. These costs vary by team, market, quality, availability, and other factors.
- (8) You are required to employ at all times, at minimum, a President, Director of Marketing, Director of Ticket Sales, Director of Corporate Sales, Director of Operations, Director of Communications/Media Relations, Director of Promotions/Community Relations, Finance Director, Ticketing Manager, and four (4) Ticket Sales Account Executives. This estimate does not include any health insurance, retirement contributions, or other benefits that may be offered by you to your employees.
- (9) You are required to employ at all times, at minimum, a Head Coach, Assistant Coach, and trainer. This estimate does not include any health insurance, retirement contributions, or other benefits that may be offered by you to your employees.
- (10) Workers compensation costs vary depending upon the state and other factors, such as the loss history of other professional soccer teams and other similarly classified professional teams (e.g., lacrosse). However, as a rule of thumb, workers compensation costs are typically between 20%-30% of player and technical staff salaries.
- (11) Benefits include anything paid to or on behalf of your employees in addition to their base salary and may include, for example, FICA, health insurance, sales commissions, and bonuses. However, this amount does not include player housing, which is included separately.
- (12) The cost of referees is dependent upon a number of factors, including number of teams, availability and quality of referees, etc. However, the cost of referees is typically negotiated by us with the Professional Referee Organization in advance of each season and then passed through equally to the teams in the League.
- (13) Home game costs vary depending on a number of factors, including the terms of the stadium lease and the expected attendance. These costs may include: (a) security; (b) ticket-takers; (c) ushers; (d) parking lot attendants; (e) concessions; (f) clean-up; (g) public address announcers; (h) scoreboard operators; (i) merchandise and other booth attendants; (j) pre-game and half-time promotions; (k) set-up and take-down or advertising materials; and (l) others. You may also be required to pay for hotel accommodations for the visiting team.

These costs do not include the cost of referees, nor do they include any broadcast production costs, each of which are included separately in this <u>Item 7</u>.

These costs also presume that you play in a stadium whose primary purpose is for the playing of soccer games. To the extent it is not (for example, if it also serves as a baseball or football stadium), there could be substantial costs associated with converting the field from baseball

or football to soccer for each game.

- (14) You will pay for transportation for your Club's travel to away games. You will also provide players with meals or *per diem* allowance while they are away from home.
- (15) Marketing costs associated with the club include the creation and distribution of advertising and advertising materials in any form (print, media, other) to generate interest in the club. These typically include television and radio commercials, promotional events, print or digital signage, and other similar advertising. These costs do not include any pre-game, halftime, or other in-game marketing or advertising, which is included separately in the "Home Game Costs."
- (16) Assumes that the Club is unable to enter into a sponsorship agreement for a competition apparel sponsor.
- (17) The USSF Membership Fee for the League is an annual fee passed through and split equally amongst the clubs. The USSF player registration fees are per-club fees issued by USSF which we pay and then seek to be reimbursed from you.
- (18) This amount reflects each team's share of a league-wide policy. For 2022, approximately \$10,000 was charged to the USL Championship and League One teams. This amount represented General Liability, Excess, Media, and Inland Marine insurance coverage under one policy.
- (19) Typically, the teams in the League agree to each pay into a pool which is then used to reimburse playoff teams for certain travel expenses they incur during the playoffs.
- (20) Reporting costs currently include the costs associated with the use of certain mandated reporting and communication platforms, including Teamworks (used to report operational information).
- (21) This amount is an estimate of the costs to put on a live stream of all of your home matches for a season. Currently, home matches must be live streamed for every home game. The home team is responsible for incurring these costs.
- (22) Estimates assume attendance of two representatives. The cost of the continuing education at the Annual General Meeting (instruction and required materials only) shall be borne by us. You are responsible for all other expenses incurred in connection with attending the Annual General Meeting (including without limitation, the costs of transportation, lodging, meals, and applicable wages).
- (23) This is a one-time fee relating to the setup and design of your website, which will be integrated into our website platform. This fee is a passthrough from our existing website host/developer. To the extent you wish to redesign your website at a later date, you will be responsible for the costs incurred in connection with such re-design.
- (24) This is a recurring, annual fee that is passed through from the service provider that hosts the League and team websites.

- (25) This amount is an estimate for additional funds which may be needed to cover the expenses of operating your franchise during one playing season. These funds which may be used to cover unexpected costs incidental to your franchise's operation.
- (26) This estimates your initial start-up expenses. We have relied on our experience in the sport of soccer to compile these estimates. You should review these figures before making any decision to purchase the franchise.

ITEM 8 – RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Your obligations to purchase or lease goods and services including without limitation, supplies, fixtures, equipment, inventory, real estate, computer hardware or software, or other items related to establishing or operating the franchised business will be limited by your franchise agreement. As discussed below, certain products or services may only be purchased from suppliers we have approved, while others may only be purchased in accordance with our specifications and standards. For clarity, we do negotiate purchase agreements with suppliers (including, as reiterated below, price terms) for the benefit of the League and franchisees.

Approved Suppliers

We may limit the number of approved suppliers with whom you may deal, designate sources that you must use (including for sale to the general public) and/or refuse any request for alternative suppliers for any reason, including that we have already designated an exclusive source (which may be us or our affiliates) for any particular item or service if we believe doing so is in the best interest of the League. Neither we nor any officer of ours owns any interest in any approved supplier, although we reserve the right to. At present, we do not disclose our criteria for approving suppliers.

We may require that you, at your expense, enter into agreements with suppliers approved by us. In those instances, we have negotiated purchase arrangements and price terms for the benefit of the franchisees. Currently, we have seven required suppliers: (1) one for soccer balls; (2) one for League patches; (3) one for broadcast production and transmission services; (4) one for the website; (5) one for mobile applications; (6) one for video clips; and (7) one for ticketing.

We will provide you with a current list of approved suppliers (including required and recommended suppliers) through updates to the Manuals or other forms of communication. For certain products or services, we may designate a single supplier and require you to purchase exclusively from such supplier. Neither we nor any of our affiliates are currently approved suppliers of any products or services, although we reserve the right to be.

If you desire to purchase goods or services from suppliers other than those previously designated as exclusive suppliers or otherwise approved by us, you must first submit to us a written request for authorization to purchase such items, and you shall not purchase from any supplier unless and until it has been approved. There are no additional fees required to be paid to us to secure the right to purchase from alternative suppliers, although we do not have to approve your request. Currently, there are no prescribed procedures relating to approval of alternative suppliers. Likewise, there are no stated procedures for revocation of approvals of alternative suppliers.

We (and our affiliates) may receive payments or other compensation from suppliers on account of your (and other franchisees') dealings with them, and we may use all amounts so received for any purpose we deem appropriate. Based on our 2022 financial statements, we did not derive any revenue, rebates or material consideration from any of the required purchases or leases of goods, services or real estate described in this Franchise Disclosure Document in 2022. Our parent's total revenue, based on the 2022 audited financial statements, was \$29,498,125. Our parent's revenue from all required purchases was \$423,359 and, based on our parent's 2022 financial statements, the percentage of our total revenues from required purchases or leases was approximately 1.4%. Currently, no affiliates of ours sell or lease products or services to franchisees.

For 2022, the estimated proportion of required purchases and leases from suppliers by franchisees relative to franchisees' total purchases and leases of goods and services required to establish and operate a franchise

is less than one percent (1%).

Standards and Specifications

We may require you to purchase certain goods or services in accordance with standards and specifications provided to you in the Manuals, provided that such goods or services are available on commercially reasonable terms and reasonably required for your Club's Activities.

Club Sponsorships

Each year, we will provide you with a listing of categories of goods or services for which we have or are pursuing league sponsorships (reserved or restricted "Product Categories"). You may not enter into Club sponsorships, or your Club sponsorships may be restricted, with respect to those Product Categories. As of the date of this disclosure document, the only reserved Product Category we have is for soccer balls (although we also prohibit and/or restrict certain advertising with respect to Product Categories that we view as not in the best interest of the league (e.g., tobacco products, firearms, pornography, etc.)). League sponsorships may require that you purchase exclusively from the League sponsor. In the event that we do not enter into a League sponsorship with respect to a reserved Product Category, it will be released back to you on an annual basis.

We do not provide material benefits to franchisees based on a franchisee's purchase of particular products, services, or use of particular suppliers.

ITEM 9 – FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this Franchise Disclosure Document.

	OBLIGATIONS	SECTION IN AGREEMENT	ITEM IN DISCLOSURE DOCUMENT
a.	Site selection and acquisition/lease	Sections 4.1, 4.2, and 4.3	Items 7 and 12
b.	Pre-opening purchases/leases	Sections 4.1, 4.2, and 4.3	Items 7, 8 and 12
c.	Site development and other pre-opening requirements	Sections 4.1, 4.2, 4.3, and 4.7	Items 7, 11 and 12
d.	Initial and ongoing training	Section 3.5	Item 7 and 11
e.	Opening	Not applicable	Items 6 and 7
f.	Fees	Sections 2.2.5, 4.2.3, 4.8.3, 4.23, 5, and 10.1	Items 5, 6, 7 and 12
g.	Compliance with standards and policies/operating manual	Sections 2.2.4, 2.2.6, 3.3, 4.2, 4.5, 4.7, 4.8, 4.12, 4.13, 4.14, 4.15, 4.17, 4.21, 5.4, 6.1, 6.3.2, 6.4, 6.6, 7.2, 9.7, 10.1, 10.2, and 10.2.1	Item 1, 8, 11, 13, 14, and 15
h.	Trademarks and proprietary information	Sections 3.3, 4.14, 4.22, 6.1, 6.2, 7, 14.1	Items 13 and 14
i.	Restrictions on products/services offered	Sections 3.3, 4.3, 6.3, 6.4, 6.5, 6.6, 6.7	Items 8 and 16
j.	Warranty and customer service requirements	Section 3.7	Not applicable
k.	Territorial development and sales quotas	Sections 3.4, 4.1, 4.2, 4.3, and 4.7	Item 12
1.	Ongoing product/service purchases	Sections 3.3, 5.3.1 and 6.7	Items 8 and 15
m.	Maintenance, appearance and remodeling requirements	Section 3.3, 4.1, 4.3.1, 4.5, 4.9, 4.13, and 4.17	Item 7
n.	Insurance	Sections 3.3.1, 4.16, and 5.3.1	Item 7 and 8
0.	Advertising	Sections 4.14 and 6.3	Item 7 and 11
p.	Indemnification	Section 11	Item 6
q.	Owner's participation/ management/staffing	Sections 4.6, 4,18, and 8	Items 1, 7, 11 and 15

r.	Records/reports	Sections 4.21	Item 7	
s.	Inspections and audits	Section 4.17, 4.21,	Item 7	
t.	Transfer	Section 4.7, 8, 10	Item 17	
u.	Renewal	Section 2.2	Item 17	
V.	Post-termination obligations	Sections 4.22, 4.23, 11, 12.5, 12.6, 12.7, 12.8, 12.9, 12.10, 12.11, 13, 14.1, and 14.2	Item 6 and 17	
w.	Non-competition covenants	Section 4.22	Item 17	
х.	Dispute resolution	Section 13	Item 17	
у.	Other: Sponsorships	Section 6.3	Item 8 and 11	
Z.	Other: Guaranty	Section 8.3	Item 15	
aa.	Other: Compliance with Governing Bodies.	Section 4.5	Item 1 and 15	
bb.	Other: Letter of Credit	Section 4.8	Item 5	
cc.	Other: Board of Governors	Section 9	Item 1	
dd.	Other: Local Broadcast Production	Section 4.9	Item 7	
ee.	Other: Minimum Performance Standards	Section 4.7	Item 1 and 17	

ITEM 10 – FINANCING

Neither we nor any affiliate offers, directly or indirectly, any financing to you. We do not guarantee any of your notes, leases or obligations. We are unable to estimate whether you will be able to obtain financing for any or all of your investment or the terms of any financing.

ITEM 11 – FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as disclosed below, we are not required to provide you with any assistance.

Pre-Opening Assistance

If you purchase a franchise described in this Franchise Disclosure Document, you will own your Club immediately upon the execution of your franchise agreement and the payment of the expansion fee. However, your Club will not begin to participate until the beginning of the next season for matches. As a result, the length of time between the signing of your franchise agreement, or the payment of the expansion fee, and the participation of your Club in the applicable League will vary depending on the effective date of your ownership of the Club and the beginning of the next season for matches for your League.

Before you open your business:

(1) Site Selection. Before you sign your franchise agreement, we will work with you to determine a territory for your franchise that is mutually acceptable. Within that territory, you must obtain access to a stadium that we approve and that meets the requirements for the League and USSF (described in <u>Item 7</u> above). Any time limit for selection of a stadium is based on evaluation of a variety of factors relevant to a franchisee's territory, stadium, and financing options. We may approve the stadium in our discretion. We do not generally own stadia for lease to franchisees.

We base our approval of stadia on their location relative to other franchises and other teams in the League and other factors that we determine to be relevant in our reasonable discretion. If you lose the right to operate at the stadium, or if you have not entered into a stadium lease at least one hundred and twenty days prior to the commencement of any season, we may, in our sole discretion, terminate your franchise agreement. All payments you made to us prior to the date your franchise agreement is terminated for this reason are non-refundable, including your expansion fee.

The responsibility for ensuring your stadium conforms to local ordinances, building codes, and all required approvals is borne by you, subject to our approval and your agreements with any municipalities or other third-parties relevant to your stadium lease or construction plans. We do not assist with construction (including, generally, any provision of equipment, signs, fixtures, or supplies), remodeling, or decorating of any facilities or stadium, although we reserve the right to, nor do we assist with the hiring and training employees, except as described below.

- (2) <u>Training/Operational Assistance</u>. While we do not offer a formal pre-opening training program, we will provide periodic training, advice, and guidance (as we deem necessary) to you in the marketing, management and operation of your Club. If you request additional assistance from us (and we agree to provide such assistance), you will pay to us per diem charges and our out of pocket expenses in providing such additional assistance. (Franchise Agreement Section 3.5)
- (3) <u>Advertising and Promotions</u>. We have the right to review all advertising and promotional materials that you propose to use that include the League Marks (as defined in Item 13) and may, but are not required to, conduct advertising that benefits your Club or the League. (Franchise Agreement – Section 4.13, 4.14)

Post-Opening Assistance

During the operation of your Club:

- (1) <u>Operational Assistance</u>. We will provide periodic training, advice or offer guidance (as we deem necessary) to you in the marketing, management and operation of your Club. If you request additional assistance from us (and we agree to provide such assistance), you will pay to us per diem charges and our out of pocket expenses in providing such additional assistance. (Franchise Agreement Section 3.5)
- (2) <u>Advertising and Promotions</u>. There is no prescribed advertising program for the System, although we reserve the right to develop and implement such a program. We reserve the right to review all advertising and promotional materials that you propose to use that include the League Marks and may, but are not required to, conduct advertising that benefits your Club or your League. (Franchise Agreement Sections 4.14, 6.3)

Currently, there is no advertising council composed of franchisees that advises us on advertising policy, you are not required to participate in a local or regional advertising cooperative, and you are not required to participate in any advertising fund.

- (3) <u>Game and League-Related Assistance</u>. We will establish and coordinate regular season and post-season game schedules for the League, establish League Rules, and perform such other activities integral to the operation of a professional soccer league, as further provided in your Franchise Agreement and in the Manuals. (Franchise Agreement Section 3.2)
- (4) <u>Insurance</u>. We will use commercially reasonable efforts to arrange for commercial general liability insurance carriers to offer policies to all franchise owners in the League. In addition, we may, in the future, designate carriers that offer workers' compensation insurance. (Franchise Agreement – Section 4.16)
- (5) <u>Collective Bargaining Assistance</u>. We may require you to appoint us as your exclusive representative with respect to collective bargaining. If this occurs, we will promptly provide you with requested information about our collective bargaining activities. Collective bargaining costs will be allocated such that the amount payable by us is twice the amount payable by each team in the League. (Franchise Agreement Section 3.6)

We are not obligated to provide you with assistance relating to: products or services you might offer; hiring and training or employees; improvements or developments in the franchise business; pricing and bookkeeping; or accounting and inventory control procedures.

Advertising

You will be permitted to create and distribute as much advertising for your franchise as you would like within your territory. Your advertising, marketing and promotion shall conform to such standards and requirements as we may specify from time to time (which may require our prior written approval). It shall not be misleading, it shall be conducted in a dignified manner, and it shall not negatively portray or adversely affect the public perception of or goodwill associated with the League, the League Office, or other teams in the League. We may require you to participate in mandatory promotions as we may develop and implement from time to time. We do not maintain or administer an advertising fund for any advertising or promotional programs, but we may do so in the future. We are not advised by an advertising council composed of franchise owners, and you are not required to participate in any advertising cooperatives. You are not required to spend any amount on advertising in your territory. If you desire to use marketing and promotional plans and materials that use the League Marks (as defined in Item 13) and have not been

provided or previously approved by us, you must submit samples of all such marketing and promotional plans and materials to us for prior approval.

We are not obligated to conduct advertising on your behalf or for your benefit. However, we reserve the right to conduct advertising for the benefit of the System generally, for the League and for teams that participate in the League, including your Club, from time to time.

Computer Systems

We do not currently require you to buy or use computer systems provided by us or that meet any required specifications; however, we may require you to buy or use certain computer systems in the future provided that such systems are available on commercially reasonable terms.

Website

We may require you to operate a website for your franchise within a website system we develop and to discontinue all outside websites. Currently, we require you to maintain a website for your franchise which is interconnected with our temporary website (www.uslsoccer.com/usl-super-league) and the other clubs in the League. The full website will be available later in 2023 at www.uslsuperleague.com. You may not establish a separate website without our prior written approval. You are responsible for maintaining and paying for your website. All websites related to your Club must comply with all requirements and guidelines that we may establish from time to time and must prominently display the logo(s) of the League or its sponsors in such location(s) as we may prescribe, in our Manuals or otherwise. You may not promote your Club or use the League Marks in any fashion on any social and/or networking websites, including, but not limited to, Facebook, LinkedIn and Twitter, without our prior written consent and compliance with our social media policy, a copy of which is included in our Manuals.

Manuals

We will loan you a copy of the Manuals (currently, Operations, Broadcast, and Competition). The Manuals contain instructions and guidance for administering your franchise (franchise agreement – Section 6.5). The Manuals (including updates and amendments as we may publish from time to time) may be provided through either a hard (paper) copy or electronically (including .pdf). We may change the Manuals in writing, provided that revisions to rules shall apply only after the time that the revision is communicated to you. The Table of Contents for the respective Manuals are included as <u>Exhibit H</u> hereto.

Broadcasts

The League owns all broadcast rights of league matches, to include regional and digital broadcasts.

Training

We do not offer a formal training program for franchisees. However, mandatory continuing education sessions and workshops are provided to all franchisees at the Annual General Meeting and at other times throughout the year in our discretion. We may require all franchisees to attend and we may choose where these sessions are located. In years past, continuing education sessions and workshops have included the following topics: ticket sales, sponsorship sales, best practices, Club operations, game-day operations, player registration, venue management, insurance, mitigating risk to your organization, reporting, expanding in your community, fan experience management, venue enhancement ideas, video and game tape analysis, social media, and getting fans to the game. This is not an exhaustive list of the seminars provided, and we reserve the right to arrange for seminars that we reasonably believe are in the best interests

of the System or the League from time to time.

We will not provide training or offer guidance with respect to your compliance with any laws, ordinances or other legal matters. The cost of training (instruction and required materials) will generally be borne by us, except that if you request assistance from us in addition to what we offer (and we agree to provide such assistance), you will pay to us per diem charges and our out of pocket expenses in providing such additional assistance. All other expenses in connection with training (including, without limitation, the costs of: transportation; lodging; meals; wages; and worker's compensation insurance) shall be borne by you. All training programs shall be at such times as may be designated by us. These training programs shall be provided at our headquarters in Tampa, Florida or another location we may designate. Additionally, we reserve the right to conduct the training online or via pre-recorded webinar.

ITEM 12 – TERRITORY

Protected Territory

You will host your Club's home games from a specific stadium, which stadium must be approved by us and comply with the standards set forth in your agreement and the Manuals. The stadium will be known as your "Home Stadium" and will be designated in your franchise agreement.

In connection with your purchase of the Club, we will grant to you a "protected territory." Your protected territory will surround your home stadium. We will negotiate with you and agree upon your protected territory, and your protected territory will be depicted in the franchise agreement. You may not solicit customers outside of your protected territory, nor may you use channels of distribution (e.g., the internet, catalog, sales, telemarketing, or other direct marketing) to solicit and/or make sales outside of your protected territory and partner in your protected territory maintains an online presence, your Club may offer merchandise for sale on such partner's online website.

You will operate your Club only in your protected territory, except as necessary to participate in away matches and as we may otherwise permit. This includes all marketing, tryouts, training, soccer camps and clinics you conduct in connection with the operation of your Club. We will not modify your territory without your consent and we will not permit another team in the League or in any other women's professional soccer league operated by us, our Parent, or our affiliates to have its home stadium in your protected territory. Any disputes regarding territory between teams will be decided by us.

Rights We Reserve

Notwithstanding the foregoing, you may face competition from other franchisees, from teams we may own, or from other Leagues that we control. However, we will not establish or operate, or grant to any other person the right to establish or operate any women's professional soccer team with its home stadium within your protected territory without your consent.

All rights not expressly granted to you are expressly reserved to us, including:

- (a) that we may solicit or accept orders for products inside your territory without compensation to you;
- (b) that we may own, acquire, establish and/or operate and license and/or franchise the right to others to establish and operate teams at any location outside of your protected territory, except that in no event will we own, acquire, establish and/or operate or license and/or franchise the right to others to establish and operate a franchise in the same League with its home stadium in your protected territory if you own and operate a franchise in that territory;
- (c) that, so long as we abide by certain rights we grant to you that are described below, we may own, acquire, establish and/or operate and license and/or franchise the right to others to establish and operate teams at any location inside your protected territory that do not participate in the League in which your Club participates (for example, we could offer for sale a USL League Two franchise to be located in your protected territory).
- (d) that we may own, acquire, establish and/or operate and license and/or franchise the right to others to establish and operate, businesses under our trademarks that are not engaged in creating, administering and operating soccer leagues, at any location within or outside your protected

territory;

- (e) that we may sell and distribute, directly or indirectly, or license and/or franchise the right to others to sell and to distribute, directly or indirectly, any products through outlets that are primarily retail in nature, or through mail order, toll free numbers, or the Internet, including those products bearing our trademarks, at or from any location within or outside of your protected territory; and
- (f) that we own any and all broadcast rights related to League matches, including those involving your Club, as described in <u>Item 11</u> above.

Relocation

You may not relocate your Club from your home stadium without our consent. If you desire to relocate, you must submit a written request, together with a site approval package. We may approve or reject your request in our sole discretion and may require that you satisfy any of the following conditions for approval: (1) you not be in default under any provision of your franchise agreement or any other agreement with us or our affiliates; (2) the new location meets our then-current standards and is located within the Protected Territory; (3) the lease (if applicable) for the proposed substitute location complies with our then-current lease requirements and is approved by us; (4) you possess the financial resources to meet the costs associated with relocating; and (5) you enter into our then-current form of franchise agreement. If you are permitted to relocate outside of your protected territory, you must pay a relocation fee to us equal to ten percent (10%) of the published expansion fee in the League at that time.

You will not have any options, right of first refusal, or similar rights to acquire additional franchises in the System within the territory or contiguous territories.

ITEM 13 – TRADEMARKS

League Marks

Under your franchise agreement, we grant you the non-exclusive right to use certain trademarks (the "League Marks") in connection with your Club's administration, marketing, promotion and operations, in accordance with rules set forth in the Manuals. The League Marks are either owned by us or are licensed to us by our parent through a licensing agreement through 2049. The following charts list the principal League Marks that you may use with your Club.

An application for the registration of the League Marks in the following chart has been filed with the Principal Register of the United States Patent and Trademark Office.

DESIGN	MARK	FILED	APPL#	<u>CLASSES</u>
[Word Mark]	USL Super League	9/20/21	97035862	41
[Word Mark]	Step Up	9/20/21	97035899	25, 41
UGL S Super League	USL S Super League (and design)	9/20/21	97035924	9, 16, 25, 28, 41
	USL S (and design)	9/20/21	97035949	9, 16, 25, 28, 41

There are no currently effective determinations of the USPTO, Trademark Trial and Appeal Board, the Trademark Administrator of any state, or any court; nor is there any pending infringement, opposition or cancellation proceedings, or material litigation, involving the League Marks listed above. There are no currently effective agreements that significantly limit our right to use or license the League Marks listed above in a manner material to the franchise. We have filed all requisite affidavits. We have not renewed any principal trademark registration as no such registrations have lapsed.

You use of the League Marks shall be in accordance with guidelines we may issue from time to time in the Manuals or otherwise in writing. We may require you to submit all proposed uses of the League Marks to us in advance for approval, which may not be unreasonably withheld. You may not use the League Marks or your Club Marks in combination with the marks or logos of other teams in the System except for promotional purposes. You may not pledge or encumber your Club Marks or the right to use the League Marks. You may not use the League Marks as part of your corporate or other legal name, or as part of any e-mail address, domain name, or other identification of you in any electronic medium.

You must modify or discontinue the use of a League Mark and adopt or use additional or substituted League Marks (after a reasonable amount of time), if we instruct you to do so. If this happens, you are responsible for your tangible costs of compliance (i.e. changing signs) and we do not have to reimburse you for any loss of revenue due to any modified or discontinued League Mark, or for your expenses of promoting a

modified or substitute League Mark. We may also develop or acquire additional League Marks and make them available for your use.

You must promptly notify us of any actual or potential infringement, counterfeiting, or other unauthorized use of the League Marks. We have the sole right to direct and control any administrative proceeding or litigation involving the League Marks and the right, but not the obligation, to take action against uses by others that may constitute infringement of the League Marks.

We do not know of any superior rights or infringing uses that could materially affect your use of the League Marks.

Club Marks

You shall identify your Club by such name, logos, characters, trademarks, trade names and other identifying words or symbols as you select, subject to our approval, which we may withhold in our discretion ("Club Marks"). We have the right to approve any changes to your Club Marks. Your Club Marks must comply with any reasonable guidelines we establish from time to time. You shall apply for and diligently pursue the registration of all Club Marks in the country in which your protected territory is located at your sole cost and expense, and you may be required to hire a trademark attorney of our choosing to do so. If you fail to register your Club Marks, we may apply to register the Club Marks on your behalf and deduct any costs associated with such application from your Letter of Credit.

In your franchise agreement, you will grant to us and each of our sponsors a non-transferable, non-exclusive right and license to use all Club Marks in our System-related activities, advertising, broadcasts, merchandise and other materials without charge and the right to sub-license the foregoing rights and licenses. You will also grant to each other franchisee that owns a franchise that participates in the same League as your Club the right to use the Club Marks in connection with game tickets and programs produced by such franchisee with respect to matches with your Club as well as in connection with promoting your matches.

ITEM 14 – PATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION

You do not receive the right to use an item covered by a patent or registered copyright, but you can use the proprietary information contained in the League's Manuals. Although we have not filed an application for a copyright or registration for the Manuals, we claim a copyright and the information contained in the Manuals is confidential and proprietary to us.

You may use any proprietary information we communicate to you only in connection with the operation of your Club during the term of your franchise agreement, including any confidential information, knowledge or know-how concerning our franchise system, the League and the marketing, management and operation of your Club and the content of our Manuals, and agree not to communicate, divulge or use for the benefit of any other person or entity any such proprietary information. You may divulge our proprietary information only to your employees that must have access to it in order to operate your Club. You must maintain the confidentiality of our proprietary information throughout the term of your franchise agreement and for two (2) years after the expiration or termination of that agreement for any reason. If any of our proprietary information for so long as it remains a trade secret.

We claim the exclusive copyright to the broadcasts of all matches played by your Club.

ITEM 15 – OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

Absent our written consent, you must retain and exercise direct management control over the business and operations relating to your Club, and you may not enter into any management agreement or other similar arrangement. If we consent to a management agreement we may at our option revoke such consent upon delivery of written notice to you. A representative of your franchise must attend up to two mandatory owners' meetings each year at your cost and expense. Additionally, you must employ a minimum number and type of personnel, including a President, Head Coach, Assistant Coach, Director of Marketing, Director of Ticket Sales, Director of Corporate Sales, Director of Operations, Director of Communications/Media Relations, Director of Promotions/Community Relations, four (4) Ticket Sales Account Executives, a trainer, clerical staff, and such other staff as may be required by the Manuals, as amended from time to time. There are no limitations on whom you may hire as employees and there are no training programs other than referenced in Item 11.

All of your direct and indirect owners of greater than 5% are required to execute a guaranty, indemnification and acknowledgment of the franchisee's obligations under the franchise agreement in the form provided in Exhibit A-2. Such individuals are also subject to those restrictive covenants listed in Section 4.22 of your franchise agreement. An owner's spouse is not subject to the covenants or required to execute a guaranty (unless he/she is also an owner). You are also required to meet United States Soccer Federation standards, as they relate to franchises and their owners. USSF may change these standards without notice or approval from us, but the standards currently contain requirements relating, in part, to: ownership make-up; ownership net worth; stadium seating capacity requirements; financial reporting; field dimensions; field surfaces; and personnel.

ITEM 16 – RESTRICTIONS ON WHAT THE FRANCHISE MAY SELL

You must conduct your Club as required by the Manuals and your franchise agreement. We do not restrict the type of goods or services that you may offer, except as provided below. We reserve the right to restrict goods or services in the future.

<u>Advertising/Sponsorship Inventory.</u> For certain core sponsorship/advertising inventory (e.g., in-stadium signage, public address announcements, game walk-out), 90% of the inventory is reserved to you and 10% is reserved to us. With respect to inventory appearing or overlaid on the broadcast (including commercial spots), such inventory is shared between you and us and the terms of such arrangement are set forth more fully in the franchise agreement. From time to time, we may require that you exchange certain inventory with us, provided that the inventory exchanged is of a reasonably equivalent value. You may be required not to sell certain inventory that we've retained or to make your existing sponsorship or advertising agreements subject to our rights to sell certain inventory. You may also be required not to promote or advertise certain goods, products or services in connection with your team, if the League has entered into (or is contemplating entering into) an agreement for the promotion or advertising of a competing good, product or service in connection with the League.

<u>Consumer Products</u>. You may be required to purchase (including for sale to the general public) from certain sources designated by us from time to time, provided that such goods are available on commercially reasonable terms (as compared to other teams in the League for similar products). You may sell consumer products within your home stadium and any store primarily identified as relating to the team, although the right to sell consumer products through the internet is reserved exclusively to us. In the event products bearing your logo are sold by third parties, 80% of the net revenues therefrom will be distributed back to you.

ITEM 17 – RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this Franchise Disclosure Document.

	Provision	Section in Franchise Agreement	Summary
a.	Length of the franchise term	Section 2.1; Schedule 3.	10 seasons.
b.	Renewal or extension	Section 2.2; Schedule 3.	An additional 10 seasons, provided you meet certain requirements, then in accordance with our then-current form of franchise agreement.
с.	Requirements for franchisee to renew or extend	Section 2.2	You must give us written notice of your intent to renew prior to the first official game of the penultimate Season. Furthermore, you must: not be in default of or owe any money under your franchise agreement or any agreement with us, our affiliates, or approved suppliers; execute the then-current form of franchise agreement (which may have materially different terms and conditions than your current franchise agreement); execute a general release in our favor; evidence of right to return to your home stadium for the renewal term; satisfy our then-current standards for financial responsibility; renew your Letter of Credit for the renewal term; and pay a renewal fee.
d.	Termination by franchisee	Section 12.2	You may terminate your franchise agreement for cause upon a minimum of thirty (30) days notice. That period may be extended until the conclusion of the then-current season or, if we are diligently working to cure the default, by a reasonable time period.
e.	Termination by franchisor without cause	Section 12.3	Upon written notice, we may terminate your franchise and all franchises in the League if the League ceases to be recognized by USSF or if we incur substantial and continuing losses, as determined in our reasonable discretion;

	Provision	Section in Franchise Agreement	Summary
			or if there are not at least 6 teams contractually committed to participate in upcoming season, upon written notice to you.
f.	Termination by franchisor with cause	Sections 12.1	We may terminate your franchise for cause if certain events (described in (f) and (g) below) occur, or if you fail to meet certain minimum performance standards. In some cases, you will have the opportunity to cure the default.
g.	"Cause" defined – curable defaults	Section 12.1.1; Definitions	You fail to pay fees or other amounts to us or our affiliates when due; you fail to pay other amounts that are due to suppliers or otherwise due under your franchise agreement; you fail to comply with your franchise agreement or any other agreement with us or our affiliates; you fail to maintain any required insurance coverage; you transfer or attempt to transfer an equity interest in violation of your franchise agreement; you fail to pay taxes when due; you fail to meet minimum performance requirements in two sets of two consecutive fiscal years and you are unable to secure a purchaser for your franchise; or you consistently fail to comply with league rules.
h.	"Cause" defined – non-curable defaults	Sections 12.1.2; Definitions	You or any of your 5% Owners admit to inability to pay debts when due; make a general assignment for the benefit of creditors; authorize, commence, or consent to certain actions under the bankruptcy laws; have a similar action taken against you; or have an attachment remain on assets for at least thirty (30) days. You or any of your 5% Owners fail to take certain actions in regard to an adverse final judgment. You fail to secure adequate rights to play in a home stadium by certain deadlines. You knowingly submit false information to us or maintain false books and records. You fail to provide and maintain the Letter of Credit. You cease to operate the team or fail to present the team for a scheduled match. You do not intend to continue operating in the League. You

	Provision	Section in Franchise Agreement	Summary
			fail to comply with applicable laws. You or any of your 5% Owners have your assets blocked under laws relating to terrorist activities. You conduct a practice or game without workers compensation insurance. You intentionally divulge classified information. You commit three or more material defaults, regardless of whether such defaults have been cured. You make material misrepresentations in connection with obtaining the franchise.
i.	Franchisee's obligations on termination/non-renewal	Sections 4.22, 4.23, 11, 12.5, 12.6, 12.7, 12.8, 12.9, 12.10, 12.11, 13, 14.1, and 14.2	Upon termination or expiration, you have a number of obligations including without limitation that you must immediately cease to operate your Club; discontinue any use of League Marks; after termination; pay all sums owed to us or our affiliates; return all proprietary information, including the Manuals; cease use of your telephone number; assign your Club's trademarks to the League; and comply with the non- competition covenants listed in (q). Additionally, if your franchise agreement is terminated for cause, we may require you to assign your franchise's assets to us.
j.	Assignment of contract by franchisor	Section 10.3	We may freely assign the franchise agreement provided that we also assign our rights in all similar agreements with other teams in the League.
k.	"Transfer" by franchisee – defined	Definitions; Section 10	Any sale, issuance, transfer, assignment, pledge, or encumbrances of rights under your franchise agreement or rights in the Club.
1.	Franchisor approval of transfer by franchisee	Section 10.1 and 10.2	You may not transfer your rights under the franchise agreement or your rights in the Club without our approval. We will approve transfers to wholly owned affiliates, family members, and upon death in accordance with a will, provided that the transfer does not result in a change of control or violation of our rules or standards, and further provided that you give us notice and go through our standard transfer procedure.

	Provision	Section in Franchise Agreement	Summary
m.	Conditions for franchisor approval of transfer	Sections 10.1 and 10.2	The Transfer is of the type described in (k) or the following conditions are met: Transferee has necessary business experience, aptitude, and financial resources, and satisfies our then-current standards for new franchisees; all outstanding obligations have been paid/satisfied; you and your owners have executed a general release of us and our affiliates; transferee has executed our then-current form of franchise agreement; transferee has completed our application form and received our then- current disclosure document; payment of a transfer fee equal to 10% of the then- current expansion fee; transferee has received all financials related to the Club; and we have received copies of all transfer documents.
n.	Franchisor's right of first refusal to acquire franchisee's business	Not Applicable	Not Applicable
0.	Franchisor's option to purchase franchisee's business	Section 12.5, 12.6, 12.7, and 12.9	If you cease to operate the Club or if we terminate your agreement for cause during a season or after the date the schedule for the season has been made public, we may take any action necessary to continue operating the Club through the relevant season. In such event, upon our request, you shall assign your rights in the Club's assets to us at no cost. Additionally, in such event, you shall indemnify us for any losses incurred in our operation of the team.
p.	Death or disability of franchisee	Section 10.2	In the event of your death or that of a 5% Owner, we will consider the transfer in accordance with (k), provided that the interest is transfer to a family member and takes place within one year of the death in accordance with a valid will or the laws of intestacy.
q.	Non-competition covenants during the term of the franchise	Section 4.22	During the term and for a period of five (5) years thereafter, you or any of your 5% Owners may not: (1) solicit, divert, or attempt to divert the Club, any other team in the League, or any other team in another league operated by our affiliates to join another men's or women's

	Provision	Section in Franchise Agreement	Summary
			professional soccer league (a "Rival Soccer League"); (2) induce or attempt to induce any of our, our affiliates', or our teams' suppliers, vendors or landlords to cease doing business with us or perform any act injurious or prejudicial to us, our affiliates, or our teams; (3) employ or seek to employ any person employed by us or other teams in the League; (4) transfer your Club's intellectual property to a third party for use in a Rival Soccer League; (5) have any interest in a team in a Rival Soccer League (including a management interest), (5) transfer your Club's interest in the stadium where it plays home matches to a team in a Rival Soccer League.
r.	Non-competition covenants after the franchise is terminated or expires	Section 4.22 and 12.5	The covenants described in section (q) above shall continue for five (5) years following the expiration or termination of your franchise agreement or any transfer, except that the restriction on participating in a Rival Soccer League shall not apply if your franchise agreement is terminated by you as described in d. above or by us as described in e. above. This provision may not be enforceable under applicable state law.
s.	Modification of the agreement	Sections 3.3, 14.12	No modifications unless: (i) superseded by League Rules agreed upon by the Board of Governors or (ii) agreed to by both parties in writing. However, the Manuals are subject to change by us and we may develop additional rules or procedures for the operation of your Club and the League.
t.	Integration/merger clause	Section 14.12	Only the terms of the franchise agreement are binding (subject to state law). Any representations and promises outside of the franchise agreement and this disclosure document may not be enforceable.
u.	Dispute resolution by arbitration or mediation	Section 13	Except for trademark disputes and requests for specific performance, temporary restraining orders and

	Provision	Section in Franchise Agreement	Summary
			temporary or preliminary injunctive relief, all disputes must be arbitrated before the American Arbitration Association or such other arbitration tribunal as we may decide.
v.	Choice of forum	Section 13.2	All arbitration and litigation shall take place in Tampa, Florida. This may be subject to applicable state law.
w.	Choice of law	Section 13.1	Your franchise agreement is governed by Florida law. This may be subject to applicable state law.

If this Franchise Disclosure Document is issued in connection with the offer and sale of franchises in the State of California, please see Exhibit G-1, which contains additional disclosures for <u>Item 17</u>.

ITEM 18 – PUBLIC FIGURES

We do not use any public figure to promote the System. However, there is no prohibition on your use of public figures to endorse or recommend your Club or on our future use of public figures to endorse the System or the League.

ITEM 19 – FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in <u>Item 19</u> may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this <u>Item 19</u>, for example, by providing information about possible performance at a particular location or under particular circumstances.

We do not make any representations about a franchisee's future financial performance or the past financial performance of company-owned or franchised outlets. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the Franchisor's management by contacting Legal Department, 1715 N. Westshore Blvd., Suite 825, Tampa, Florida 33607 (legal@uslsoccer.com) (813) 963-3909, the FTC, and the appropriate state regulatory agencies.

ITEM 20 – OUTLETS AND FRANCHISEE INFORMATION

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the end of the Year	Net Change
	2020	0	0	0
Franchised	2021	0	0	0
	2022	0	0	0
	2020	0	0	0
Company-Owned	2021	0	0	0
	2022	0	0	0
	2020	0	0	0
Total Outlets	2021	0	0	0
	2022	0	0	0

Table No. 1League Outlet SummaryFor Years 2020 to 2022

Table No. 2Transfers of Outlets From Franchisees to New Owners (Other than the Franchisor)For Years 2020 to 2022

None / Not applicable.

Table No. 3Status of Franchised OutletsFor Years 2020 to 2022

None / Not applicable.

Table No. 4Status of Company-Owned OutletsFor Years 2020 to 2022

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired from Franchisees	Outlets Closed	Outlets Sold To Franchisees	Outlets at End of the Year
	2020	0	0	0	0	0	0
Total	2021	0	0	0	0	0	0
	2022	0	0	0	0	0	0

Table No. 5Projected Openings as of December 31, 2022

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets In the Next Fiscal Year	Projected New Company-Owned Outlets In the Current Fiscal Year
Alabama	0	0	0
Arizona	0	0	0
California	0	0	0
Colorado	0	0	0
Florida	0	0	0
Georgia	0	0	0
Idaho	0	0	0
Illinois	0	0	0
Indiana	0	0	0
Kentucky	1	0	0
Maryland	0	0	0
Michigan	0	0	0
Minnesota	0	0	0
Mississippi	0	0	0
Nebraska	0	0	0
New Jersey	0	0	0
New Mexico	0	0	0
New York	0	0	0
North Carolina	0	0	0
Ohio	0	0	0
Oklahoma	0	0	0
Oregon	0	0	0
South Carolina	0	0	0
Tennessee	0	0	0
Texas	1	0	0
Virginia	0	0	0
Washington, D.C.	1	0	0
Wisconsin	0	0	0
Total	3	0	0

Exhibit D lists the name, address, and telephone number of all current franchisees for each of their outlets as of December 31, 2022.

<u>Exhibit E</u> lists the name, city, state and current business telephone number (or if unknown, the last known home telephone number) of every franchisee who had an outlet terminated, canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business during the most recently completed fiscal year or who has not communicated with us within 10 weeks of December 31, 2022. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise System.

During the last three fiscal years, no current or former franchisees have signed confidentiality clauses that restrict them from discussing with you their experiences as a franchisee in our franchise System.

ITEM 21 – FINANCIAL STATEMENTS

Our Parent's audited financial statements for the periods (i) January 1, 2020 to December 31, 2020, (ii) January 1, 2021 to December 31, 2021; and (ii) January 1, 2022 to December 31, 2022 are attached as <u>Exhibit F-1</u> to this Franchise Disclosure Document. Our Parent has signed a guaranty in which it absolutely and unconditionally guarantees to assume our obligations to you under the Franchise Agreement. A copy of such guaranty is attached as <u>Exhibit F-2</u>.

ITEM 22 – CONTRACTS

A copy of the Franchise Agreement is attached to this Franchise Disclosure Document as <u>Exhibit A-1</u>. A copy of (1) the guaranty to be executed by your owners and (2) the lease addendum to be executed by you and your landlord are also attached as Exhibits A-2 and A-3, respectively. Please refer to state specific amendments to the Franchise Agreement in Schedule 9 of the Franchise Agreement for any additional agreements that apply in your state.

ITEM 23 – RECEIPTS

Two copies of an acknowledgment of your receipt of this Franchise Disclosure Document are located at the end of this Franchise Disclosure Document. Please return one signed copy to us and retain the other copy for your records.

EXHIBIT A-1 – FORM OF FRANCHISE AGREEMENT

[See attachment]

FRANCHISE AGREEMENT

This Franchise Agreement (this "<u>Agreement</u>") is made effective as of the date included in Schedule 3 (the "<u>Effective Date</u>"), by and between USL Super League, LLC a Florida limited liability company, having its principal place of business at 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607 (the "<u>League Office</u>") and the Person identified on Schedule 2 hereto (the "<u>Team Operator</u>"). The League Office and the Team Operator are each a "<u>Party</u>" and collectively the "<u>Parties</u>" herein.

I. GRANT OF RIGHTS

The League Office grants to the Team Operator the right, and the Team Operator hereby undertakes the obligation, upon the terms and conditions set forth in this Agreement, including the definitions set forth in Schedule 1 attached hereto (all of which are incorporated herein by reference), to operate the Team throughout the Term.

II. TERM; RENEWAL; EXPIRATION

2.1. <u>Term</u>. This Agreement shall be effective as of the Effective Date and, except as otherwise provided herein, this Agreement shall expire effective immediately following the conclusion of the Final Season, as set forth in Schedule 3 (the "Term").

2.2. <u>Renewal Term</u>. The Team Operator may continue its operation of the Team for the renewal term set forth in Schedule 3 (the "<u>Renewal Term</u>") solely by meeting the following requirements:

2.2.1. The Team Operator executes a New Agreement prior to the commencement of the third-to-last Season of the Term (the "<u>Renewal Date</u>");

2.2.2. As of the Renewal Date, the Team Operator is not in material default under any provision of this Agreement or any other agreement between the Team Operator and the League Office Entities or the approved Providers of the League and has complied in all material respects with the terms and conditions of such agreements throughout the terms thereof;

2.2.3. On or before the Renewal Date, the Team Operator executes a general release, in a form prescribed by the League Office, of any and all claims against the League Office Entities and their respective Personnel;

2.2.4. As of the Renewal Date, the Team Operator either: (a) remains in possession of or retains the right to use the Home Stadium and agrees (regardless of cost) to renovate, remodel, or expand the Home Stadium and otherwise modify the Home Stadium as required to comply with standards then applicable for new teams in the League; or (b) secures a substitute site that the League Office has approved and the Team Operator constructs or renovates a stadium at that site according to standards then applicable for new teams in the League;

2.2.5. On or before the Renewal Date, the Team Operator pays to the League Office the renewal fee set forth in Schedule 4 (the "<u>Renewal Fee</u>"); and

2.2.6. On or before the Renewal Date, the Team Operator and the Principal Owner present evidence reasonably satisfactory to the League Office that it satisfies the League Office's standards of financial responsibility and has sufficient financial resources and means to continue to operate the Team during the Renewal Term.

2.3. <u>Expiration</u>. Any Renewal Term shall be conditioned upon the satisfaction of the conditions set forth in Section 2.2 above. If the Team Operator fails to meet any of the conditions set forth therein, the rights granted to the Team Operator hereunder shall automatically expire at the end of the Term.

III. RIGHTS AND OBLIGATIONS OF THE LEAGUE OFFICE

3.1. <u>Reservation of Rights</u>. The League Office shall retain all rights not expressly granted to the Team Operator under this Agreement.

3.2. <u>Game Schedule</u>. The League Office will establish and coordinate regular season and post-season game schedules for the League. Neither the Team Operator nor the Principal Owner may promote or participate in any Match unless such Match is approved by the League Office. During the course of any season, the Team Operator shall ensure that there are no more than three weekends (Friday, Saturday and Sunday) on which Home Matches may not be scheduled (black-out dates) at the Home Stadium.

3.3. League Rules.

3.3.1. The Team Operator acknowledges the importance of operating the Team in accordance with League Rules and agrees to comply therewith. The Team Operator further acknowledges that such League Rules may relate to League Activities and Team Activities including, without limitation, to:

(a) the operation and coordination of the League competition, including matters relating to: player registration and eligibility; referees; game cancellations and makeups; playoff and championship structure and eligibility; and score keeping;

(b) the presentation of Home Matches, including the appearance and presentation of: (i) the Broadcasts and related programming in all forms of media; (ii) the Field of Play; and (c) any products, equipment, apparel, or other items used, displayed, advertised or promoted on (i) or (ii).

(c) the allocation of specific Core Commercial Inventory, provided such League Rules are consistent with Section 6.3.3.

(d) the dimensions, appearance, surface, and quality standards of the playing field and any equipment used thereon at the Home Stadium;

(e) the use of the Team Marks, League Marks and Competition Marks in all mediums, the management and protection thereof, and the activities related thereto;

(f) the (i) types, models, or brands and (ii) minimum and required standards and specifications for facilities, services, goods, supplies, and equipment that the Team Operator may use or sell;

(g) the obligations imposed on the Team Operator by any League Commercial Affiliations; Licensing Arrangements; Broadcast agreements; or arrangements with approved Providers of the League.

(h) the obligations imposed on the Team Operator with respect to the

hosting, promotion, and operation of the League's championship Match;

(i) the designation of League Matches as Featured Games, provided that the Team may only be included in three regular season Featured Games in any given Season;

(j) the manner in which the Team Operator may promote or otherwise affiliate with other soccer teams (whether youth, amateur, or professional);

(k) the development of players (including the league(s) in which an Academy Team may participate);

(l) the implementation of security protocols and procedures relating to the health, safety, or protection of (i) the Team and (ii) attendees of the Home Stadium for Team-related events;

(m) standards and specifications with respect to the creation, maintenance, design, functionality and content of the Team Site and Team Digital Platforms, including links to the websites of third parties, legal notices, policies, and terms of usage;

(n) the delivery of the Team Lists to the League Office, the use, management and protection thereof, and the activities related thereto;

and off-field);

(o) the conduct (or misconduct) of players and coaching staff (both on-field

the League;

(p) the solicitation of players and coaching staff employed by other teams in

(q) the signing and Transfer of players, specifically including the form of player contracts and player Transfer contracts that must be used by the Team Operator, as well as the recruitment of players and coaches from other teams in the League;

(r) workers compensation, general liability, and other insurance requirements related to Team Activities;

(s) the Persons who may become Equity Owners of the Team Operator;

(t) the minimum requirements for stadiums in the League, including minimum requirements for: seating capacity; lighting; locker rooms; press boxes; storage; internet; broadcast capabilities; field surface; field dimensions; and security;

(u) the reporting requirements relating to the Team and Team Operator; and

(v) the establishment, implementation and enforcement of such League

Rules.

3.3.2. The Team Operator acknowledges and agrees that the League Office may create, change or modify League Rules as the League Office reasonably deems appropriate and, upon reasonable advance notice, the Team Operator shall implement, accept, and otherwise comply with such revised League Rules at the Team Operator's sole cost and expense. Notwithstanding the foregoing, complete and detailed uniformity might not be possible or practical, and the League Office specifically reserves the

right and privilege, as the League Office deems best, to allow for variations or accommodations for any team or team operator based upon the peculiarities of any condition or factors that the League Office considers important to the successful operation of the League or that team or team operator. The Team Operator has no right to require the League Office to grant the Team Operator a similar variation or accommodation.

3.4. <u>Territory</u>. During the Term, the League Office Entities shall not establish, operate, or grant to any other team operator the right to establish or operate a team in a women's outdoor professional soccer league whose home stadium is in the geographic area included in Schedule 3 (the "Protected Territory"). Except as reasonably required to participate in away Matches, all Team Activities shall occur solely within the Protected Territory.

3.5. <u>Training</u>. The League Office shall offer training to the Team Operator for such time and on such topics as the League Office may deem reasonable or necessary in its sole discretion ("Team Operator Training"). The League Office may require the attendance of specific Personnel at Team Operator Training, and Team Operator Training may be offered at such locations as the League Office may designate. Notwithstanding the foregoing, the Team Operator acknowledges that the League Office shall not be responsible for training or offering guidance with respect to Team Operator's compliance with any Laws or other legal matters.

3.6. <u>Collective Bargaining</u>. The Team Operator hereby appoints the League Office as its exclusive representative with respect to Collective Bargaining Activities; provided that (i) the League Office shall act as the agent of the League in any negotiations relating to the foregoing and shall not represent or hold itself out as acting on its own behalf; and (ii) the League Office shall not enter into agreements related to the foregoing without the prior approval of the Board of Governors. The League Office shall promptly provide the Team Operator with such details as the Team Operator may from time to time reasonably request in connection with the League Office's activities hereunder. The costs incurred by League Office Entities related to such Collective Bargaining Activities shall be allocated such that the amount payable by the League Office is twice the average amount payable by an individual team operator.

3.7. Sanctioning.

3.7.1. The League Office shall use commercially reasonable efforts to ensure that the League is sanctioned as a professional league by USSF throughout the Term. Accordingly, one or more Governing Bodies may govern, sanction and/or oversee the League from time to time. The Team Operator expressly acknowledges that any such Governing Body may simultaneously sanction one or more leagues and/or Rival Leagues and/or change or revoke the status or designation conveyed by such Governing Body to or affiliation with the League, without the prior consent of the League Office.

3.7.2. The League Office does not warrant and makes no guaranty whatsoever that present or future designations conveyed upon the League by a Governing Body (including any divisional sanctioning by USSF) will remain the same throughout the Term, and the Team Operator expressly acknowledges that it is not purchasing this Team in reliance upon or anticipation of the sanctioning of the League by any Governing Body, any specific League designation associated therewith, or the maintenance of such sanctioning or designation throughout the Term. The Parties agree that the League Office shall have no liability to the Team Operator based upon the League's sanctioning or designation (or lack thereof) by any Governing Body.

3.8. <u>Gender Equity</u>. The Team Operator acknowledges the interest of the League in ensuring that its players are treated in a comparable manner to the male players participating in the highest level

league operated by League Office Entities (currently, the USL Championship). In connection with the foregoing, the Team Operator acknowledges and agrees that (unless there is a collective bargaining agreement governing such terms and conditions of employment of the players), the League Office may establish, implement, and modify standards, rules and regulations relating to the minimum compensation and other benefits to be provided to players by teams in the League; *provided that* such standards, rules and regulations are consistent with those found in (a) the collective bargaining agreements of men's professional players in the highest level league operated by League Office Entities (currently, the USL Championship) or (b) with respect to any gender-specific topics, the collective bargaining agreements of other comparable women's professional sports leagues.

IV. RIGHTS AND OBLIGATIONS OF TEAM OPERATOR

4.1. Home Stadium.

4.1.1. The Team Operator will construct the Home Stadium in full compliance with League Rules, and it (or one or more of its Affiliates) will remain the owner or primary tenant thereof. The primary purpose of the Home Stadium will be to serve as the home stadium of the Team. The League Office will be under no obligation to permit the Team to use any interim stadium prior to the construction of the Home Stadium.

4.1.2. The Team Operator will, as a material condition of this Agreement, strictly comply with the Stadium Development Schedule in Schedule 9. Failure to comply with the deadlines set forth in Schedule 9 shall be a material breach of the Franchise Agreement.

4.1.3. Subject to Section 4.2.4, all Home Matches must be played at the Home Stadium. Hosting Home Matches at any site other than the Home Stadium, whether within or outside of the Protected Territory, shall require the written advanced approval of the League Office, to be granted or withheld in the League Office's sole discretion.

4.1.4. The Team Operator hereby acknowledges and agrees that approval by the League Office of the Home Stadium or the Stadium Lease (or any subsequent home stadium or stadium lease) does not constitute an assurance, representation or warranty of any kind, express or implied, as to the suitability of the Home Stadium or the Stadium Lease for any purpose. With respect to any prospective relocation from the Home Stadium, the League Office shall provide the Team Operator with such assistance as the League Office determines in its sole discretion.

4.2. <u>No Relocation</u>. The Team Operator shall not relocate to a new or substitute home stadium without the prior written approval of the League Office. The Team Operator is not permitted to apply for relocation until the conclusion of the fifth Season in which the Team participates in the League. If the Team Operator desires to relocate the Team, the following terms and conditions shall apply:

4.2.1. Without the prior written consent of the League Office, the Team Operator shall not, and shall not authorize or permit any of its Affiliates or any of its or their Personnel to, directly or indirectly: (i) solicit, initiate, encourage, facilitate or continue inquiries regarding a Relocation Proposal to a new market; (ii) enter into discussions or negotiations with, or provide any information to, any Person concerning a possible Relocation Proposal to a new market; or (iii) enter into any agreements or other instruments (whether or not binding) regarding a Relocation Proposal, regardless of whether the relocation is outside the Protected Territory.

4.2.2. Upon receiving consent in accordance with Section 4.2.1 4.2.1 aboveabove, the Team Operator shall submit such materials and information as the League Office may reasonably request

for the evaluation of the Relocation Proposal. The League Office may, in its sole discretion, approve or reject the requested Relocation Proposal and may condition its approval on the satisfaction of certain conditions, including without limitation: (a) the Team Operator is not in material default under any provision of this Agreement, or any other agreement between the Team Operator and the League Office or their respective Affiliates; (b) the proposed substitute location meets the then-current standards for the League; (c) the Stadium Lease (if applicable) for the proposed substitute location complies with the League Office's then-current Stadium Lease requirements (which shall include the requirement that the Stadium League Office required as of the Effective Date with respect to the Home Stadium); (d) the Team Operator obtains the League Office's approval of the proposed Stadium Lease; and (e) the Team Operator possesses the financial resources to meet the costs associated with relocating, as determined in the League Office's office's discretion. For clarification, the above list of options is non-exhaustive and not intended to limit the League Office's discretion to approve or reject a Relocation Proposal.

4.2.3. In the event that the Relocation Proposal contemplates the relocation of the Team outside of the Protected Territory to a new market (the "<u>New Market</u>"), the League Office may reject a Relocation Proposal to the New Market consistent with its discretion, as described in Section 4.2.2, or it may condition any such approval upon the satisfaction of the following additional conditions: (a) the Team Operator enters into a New Agreement; (b) if the League Office has paid any consultants, vendors, or other third parties in connection with an evaluation of the New Market, the reimbursement of such costs; and (c) the Team Operator pays the relocation fee set forth in Schedule 4 (the "<u>Relocation Fee</u>").

4.2.4. If, through no fault of the Team Operator, the Home Stadium is damaged or destroyed by an event such that repairs or reconstruction cannot be completed within sixty (60) days thereafter, then the Team Operator shall have forty-five (45) days after such event in which to apply for the League Office's approval to relocate (within the Protected Territory) and/or reconstruct the Home Stadium, which approval shall not be unreasonably withheld, conditioned or delayed.

4.3. <u>Operation Within Protected Territory</u>. Except as otherwise expressly set forth herein and as necessary to participate in away Matches, the Team Operator shall conduct Team Activities only inside the Protected Territory. Without the prior written consent of the League Office, the Team Operator shall not enter into any agreement that is inconsistent with the purpose of this Section 4.3. Any territorial dispute between the Team Operator and any other team operator shall be resolved by the League Office in good faith.

4.4. <u>Adherence to Schedule</u>. The Team Operator shall ensure the Team's attendance at all scheduled Matches, including in any competition required by a Governing Body. The Team Operator shall comply with all policies regarding cancellation, rescheduling of Matches and compensation to teams in the League affected by cancellations or no-shows.

4.5. <u>Professional League Standards</u>. The Team Operator shall strictly comply with the Professional League Standards. The Team Operator acknowledges that: (i) the Professional League Standards currently include rules and standards related, in part, to: ownership composition; Principal Owner net worth; stadium capacities; field dimensions; field standards and turf requirements; required personnel; and reporting obligations, although the requirements and standards may relate to additional areas in the future; and (ii) it has been provided with a copy of USSF's Professional League Standards in force as of the Effective Date.

4.6. <u>Team Management</u>. Unless the League Office consents in writing, the Team Operator and the Principal Owner must at all times retain and exercise direct management control over the business and operations relating to the Team. Without limitation of its rights in law or equity, the League Office

has the full and absolute right to rely on any and all statements or communications by or from the Principal Owner, and the Team Operator shall hold harmless the League Office from any reliance thereon. The Team Operator may not enter into any management agreement or other similar arrangement with any Person for all or a part of the operation or business of the Team (a "Management Arrangement") without the prior written consent of the League Office. Even after the League Office's approval of a Management Arrangement, the League Office may at its option revoke that approval, and upon delivery of written notice to the Team Operator require the Team Operator to terminate the Management Arrangement.

4.7. Minimum Performance Standard. The Team Operator shall ensure that it meets the Minimum Performance Standard for each Fiscal Year. The Team Operator will be considered not to have met the Minimum Performance Standard if its (a) Local Revenues or (b) turnstile attendance at League Matches do not exceed half of the median Local Revenues or median turnstile attendance of the other team operators in the League (the "Minimum Performance Standard"). If the Minimum Performance Standard is not met for a period of two consecutive Fiscal Years, the Team Operator shall, by the end of the following Fiscal Year, create and present to the League Office a plan to meet the Minimum Performance Standard. Throughout the remainder of the Term, if the Team Operator does not meet the Minimum Performance Standard in any two consecutive Fiscal Years, the League Office may require the Team Operator to Transfer the rights granted under this Agreement to a third party acceptable to the League Office. The League Office shall send a default notice providing that the Team Operator shall have a period of ninety (90) days to secure a purchaser for the rights granted under this Agreement and a subsequent period of ninety (90) days to complete the sale. In the event the Team Operator does not secure a purchaser or complete the sale on such timeline, the League Office may terminate this Agreement pursuant to Section 12.1.2(q).

4.8. <u>Letter of Credit</u>.

4.8.1. From July 15 of the year preceding the Initial Season through the remainder of the Term, the Team Operator shall, at all times, have established a Letter of Credit in the amount of one hundred fifty thousand dollars (\$150,000) or such greater amount as may be required by the League Office to comply with Professional League Standards, that secures the Team Operator's obligations hereunder and in the performance of Team Activities. To the extent the League Office draws upon the Letter of Credit as permitted hereunder, the Team Operator shall replenish the amount of such Letter of Credit to the requisite amount within thirty (30) days of notice of such draw.

4.8.2. In the event that (a) the Team Operator has three (3) or more payments to the League Office that are overdue in any twelve (12) month period; (b) the Team Operator has overdue payments that, at any given point during the Term, exceed ten thousand dollars (\$10,000); or (c) the Team Operator otherwise materially breaches this Agreement, the League Office may increase the required amount of the Letter of Credit in its reasonable discretion.

4.8.3. To the extent the Team Operator fails to make any payment as and when due in accordance with Article V or any other provision of this Agreement or any other agreement between a League Office Entity and the Team Operator, the League Office may draw such amounts from the Letter of Credit, together with any late payment or other fees payable in connection with such delinquent amounts, <u>provided that</u> the Team Operator has not provided the League Office with reasonably sufficient evidence that it has paid such delinquent amounts in full within five (5) days of notice thereof.

4.9. <u>Field Signage System</u>. The Team Operator shall ensure that it has a system capable of displaying signage (e.g., soccer field boards) sufficient to surround, at minimum, the side of the field visible on camera during the Broadcasts.

4.10. Broadcast Production.

4.10.1. The Team Operator must produce a live Broadcast of each of its Home Matches, which must be made available to the League Office in such manner and format as the League Office reasonably requires for dissemination by the League Office.

4.10.2. As of the Effective Date, the League Office will enter into an agreement for the establishment of a dedicated network connecting the venues of each of the teams in the League and the archival of each of the League Matches. Each team operator in the League (including the Team Operator) will be required to contribute a proportionate amount towards the establishment and use of the network and archival services. The annual contribution amount has yet to be determined but is expected to be no less than twenty-five thousand dollars (\$25,000) for each team operator in the League.

4.11. <u>Sale of Tickets</u>. The Team Operator shall provide for the sale of Home Tickets to each of its Home Matches. The Team Operator shall comply with any reasonable ticket policies set forth from time to time by the League Office, including with respect to the Team Operator's participation in any ticket distribution programs. At the League Office's request, the Team Operator shall provide the League Office a reasonable number of desirable Home Tickets without charge. The League Office shall request all Home Tickets at least 48 hours in advance of such game. Home Tickets provided to the League Office shall be used by League Office Entities for promotional purposes, for employees or guests, or for charitable purposes and shall not be sold to third parties.

4.12. <u>Team Lists</u>. The Team Operator hereby grants to the League Office an exclusive, perpetual, royalty free license to use the Team Lists solely in connection with League Activities and the Team Operator shall deliver or otherwise make available the Team Lists to the League Office as required by League Rules; provided that the League Office shall not license or otherwise Transfer the Team Lists unless within parameters established by the Board of Governors in accordance with Section 9.7 (by Majority Vote).

4.13. Team Site.

4.13.1. The League Office grants to the Team Operator the right to, and the Team Operator shall, operate and manage the Team Site in accordance with League Rules. The Team Operator shall ensure the continued registration of the Domain Name. Other than the Team Site, the Team Operator shall not authorize the creation or operation of any website relating to the Team.

4.13.2. The Team Operator shall pay all costs associated with the initial design and ongoing maintenance of the Team Site. If a re-design of the Team Site is required by the League Office, the League Office shall bear such costs, provided that if the Team chooses (and the League Office permits the Team) to deviate from any templates afforded by the League Office, the cost of such deviations shall be borne by the Team Operator. In the event that the Team Operator desires to redesign the Team Site, such costs shall be borne by the Team Operator.

4.14. Local Advertising. The Team Operator may conduct Local Advertising in accordance with League Rules, which may require prior approval of all advertising copy, and this Agreement. All Local Advertising shall not be misleading, shall be conducted in a dignified manner, and no Local Advertising shall negatively portray or adversely affect the public perception of or goodwill associated with the League, the League Office, or other teams in the League. Except as otherwise permitted by League Rules, the Team Operator shall not use the League Marks in its Local Advertising without the written consent of the League Office. Further, the Team Operator shall ensure that any photos, images, videos, or other forms of art or media used in its Local Advertising, on the Team Site, or in the

Broadcasts are lawfully and properly used.

4.15. <u>Players</u>. The Team Operator's contracts with players shall be in the standard contract form for the League, which may be modified from time to time at the discretion of the League Office, and the status, rights and privileges of the players shall at all times be subject to League Rules.

4.16. <u>Insurance and Worker's Compensation</u>. The Team Operator shall obtain and pay for insurance through any group plans identified by the League Office, if any. The Team Operator shall maintain general liability insurance policies with the carrier(s) selected by the League Office, naming the League Office Entities as additional insureds on the face of each policy, and acknowledges that such insurance policies shall have no less than \$1,000,000 per occurrence limits for bodily injury and property damage and \$2,000,000 aggregate limits for each team. The Team Operator shall also maintain workers' compensation insurance in such amounts as the League Office may reasonably require or in such higher amounts as may be required by applicable Law. All insurance policies shall be timely renewed, and policies and certificates together with evidence of payment of premiums shall be delivered to the League Office at least thirty (30) days prior to the expiration of such policies by certified mail, hand delivery with written receipt, or overnight courier service with verification of receipt.

4.17. <u>Inspections</u>. Upon reasonable advance notice, the Team Operator shall permit the League Office and its Personnel to enter the Home Stadium and the Team Operator's principal place of business during normal business to inspect the Team Operator's operations and facilities. The Team Operator shall cooperate with representatives in such inspections by rendering such assistance as they may reasonably request and, upon notice from the League Office or its Personnel and without limiting other rights of the League Office under this Agreement, shall take such steps as may be necessary to promptly correct any deficiencies detected during any such inspection that constitute defaults hereunder or otherwise violate League Rules. The League Office agrees that no inspection hereunder shall unreasonably interfere with the business of the Team Operator or its Affiliates.

4.18. <u>Personnel</u>. Throughout the Term, the Team Operator shall employ or engage qualified individuals to perform the Team Operator's obligations under this Agreement, provided, however, that the Team Operator must have engaged individuals to handle the responsibilities of the following positions: general manager; director of marketing/sales; director of communications/media relations; director of promotions/community relations; director of game operations; head coach; assistant coach; trainer; ticketing manager; finance director; and at least four (4) ticket sales account executives. The League Office reserves the right to increase or lower these requirements in its reasonable discretion, provided it delivers to the Team Operator reasonable advanced notice of such change.

4.19. <u>Permits; Licenses</u>. The Team Operator shall comply in all material respects with applicable Laws and shall timely secure and maintain in force in its name all required licenses, permits, and certificates required for the lawful operation of the Team and otherwise relating to the conduct of its business pursuant to this Agreement, including licenses to do business, fictitious name registrations and sales tax permits. The Team Operator shall certify in writing to the League Office that all such permits and certifications have been obtained and provide copies to the League Office as requested and at its sole costs and expense.

4.20. <u>Budget Review</u>. The Team Operator agrees to prepare an annual budget and business and marketing plan and to submit an informational copy of such budget and plan to the League Office not less than 60 days before the beginning of each Fiscal Year, except as may be provided by the League Rules.

4.21. Books and Records.

4.21.1. The Team Operator shall (i) maintain full and accurate books of account and records on an accrual basis, using the categories and chart of accounts required by the League Office, to reflect the operations of the Team and the business of the Team Operator (collectively, "Books and Records"); (ii) adopt the Fiscal Year as its fiscal year, unless required to do otherwise under the Internal Revenue Code of 1986, as amended; (iii) no more than 20 days after the end of every month, provide to the League Office unaudited financial statements for the previous month; (iv) provide other periodic reports as requested by the League Office, all in accordance with the League Rules; and (v) not later than ninety (90) days after the end of the Fiscal Year, provide the League Office with one or more of the following as the League Office may request, certified by the Team Operator's chief financial or principal accounting officer to be true and correct: complete financial statements for the Fiscal Year (including a balance sheet, statement of operations and statement of cash flow) prepared in accordance with generally accepted accounting principles consistently applied; income tax returns for the Fiscal Year; and statements reflecting all sources and amounts of revenue generated during the Fiscal Year. Any knowingly or materially false certification shall be a material breach of this Agreement. The League Office may require the Team Operator to have audited financial statements prepared annually during the Term.

4.21.2. The Team Operator shall maintain throughout the Term and for not less than three (3) years following the termination or expiration of this Agreement, expiration or non-renewal of this Agreement, full, complete, and accurate books, records and accounts in accordance with GAAP and in the form and manner prescribed by the League Office from time to time in the Manuals or otherwise in writing.

4.22. <u>Restrictive Covenants</u>.

4.22.1. *Restrictive Covenants*. The Team Operator acknowledges that, pursuant to this Agreement, the Team Operator will receive valuable, specialized training and Confidential Information, including information regarding the operational, sales, promotional and marketing methods and techniques of the League Office. Subject to Section 14.1, the Team Operator and each of its Equity Owners of five percent (5%) or greater covenant that, during the Term and for a continuous uninterrupted period of five (5) years thereafter or, with respect to such an Equity Owner, five (5) years after such Equity Owner's relinquishment of any rights in the Team Operator, they shall not, either directly or indirectly, on behalf of, or in conjunction with any Person, except as set forth below or otherwise approved in writing by the League Office:

(a) Solicit, divert or attempt to divert any team in the League or any other women's professional soccer league operated by the League Office Entities to join a Rival League or in concert or association with any such team or on its own accord organize, develop, plan for, conduct research, or engage consultants or other third parties, or otherwise take any steps whatsoever (including meeting with proposed participants) to start, develop, create, form, merge with, join or acquire (through any means) a Rival League;

(b) Induce or attempt to induce any Provider or landlord of the League Office Entities or teams in the League to cease doing business therewith; or do or perform, directly or indirectly, any other act injurious or prejudicial to the goodwill associated with the League or the League Marks;

(c) Solicit for employment any individual who is at that time employed by the League Office Entities;

(d) License, assign, or otherwise Transfer the intellectual property of the Team (specifically including the Team Marks) to a third party for use in such third party's operation of a

Rival League or a team participating in a Rival League;

(e) Own, maintain, operate, or have any interest in (as owner or otherwise) any Rival League or in any team that participates (or has agreed to participate) in a Rival League; <u>provided, however</u> that this Section 4.22.1(e) does not apply to the ownership or operation of any team participating in a league operated by League Office Entities or any youth soccer team;

(f) Assign, license, sublease, or otherwise Transfer the Home Stadium or the right to use the Home Stadium to a third party whose purpose is to operate a team that participates in a Rival League;

(g) Publicize the intent or desire to become, develop, create, form, merge with, join or acquire (through any means) a team in a Rival League.

4.22.2. *Modification*. If any court of competent jurisdiction determines that any of the covenants set forth in this Section 4.22 is excessive in duration or scope or is unreasonable or unenforceable, it is the intention of the parties that such restriction may be modified or amended by the court to render it enforceable to the maximum extent permitted by law.

4.22.3. Other Professional Team. In addition to the restrictive covenants set forth in 4.22.1, the Team Operator and each of its Equity Owners of five percent (5%) or greater further covenant that, during the Term and for a continuous uninterrupted period of five (5) years thereafter or, with respect to such an Equity Owner, five (5) years after such Equity Owner's relinquishment of any rights in the Team Operator, they shall not, either directly or indirectly, on behalf of, or in conjunction with any Person, own, maintain, operate, or have any interest in (as owner or otherwise) (or apply to own, operate, or have any interest in) any professional team (regardless of sport) that will use the same stadium as the Team, except (a) as otherwise approved in writing by the League Office or (b) if the Team Operator has recorded operating profits for at least five (5) consecutive years.

4.22.4. *Equitable Remedies*. The Team Operator and each of its Equity Owners of five percent (5%) or greater acknowledges and agrees that (a) a breach or threatened breach by such party of any of its obligations under Section 4.22 would give rise to irreparable harm to the League Office for which monetary damages would not be an adequate remedy and (b) if a breach or a threatened breach by such party of any such obligations occurs, the League Office will, in addition to any and all other rights and remedies that may be available to it at law, at equity, or otherwise in respect of such breach, be entitled to equitable relief, including a temporary restraining order, an injunction, specific performance and any other relief that may be available from a court of competent jurisdiction, without any requirement to (i) post a bond or other security, or (ii) prove actual damages or that monetary damages will not afford an adequate remedy. The Team Operator and each of its Equity Owners of five percent (5%) or greater agrees that they shall not oppose or otherwise challenge the appropriateness of equitable relief or the entry by a court of competent jurisdiction of an order granting equitable relief, in either case, consistent with the terms of this Section 4.22.

4.23. <u>Operating Covenant</u>. The Team Operator shall continuously operate the Team throughout the Term, including without limitation by causing the Team to participate in any and all League Matches. Failure of the Team Operator to continuously operate the Team in accordance with this section 4.23 shall be a material breach of this Agreement.

4.24. Academy Participation.

4.24.1. Within three (3) years of the Initial Season, the Team Operator shall ensure that

both a boys and a girls Academy Team participates in either the USL Academy Cup or USL Academy League (or both). If the Academy Team initially participates in the USL Academy Cup, but not the USL Academy League, the Team Operator shall ensure that the Academy Team commences participation in the USL Academy League the following year. Thereafter, the Team Operator shall ensure that an Academy Team participates in the USL Academy League throughout the remainder of the Term. As used herein, "<u>Academy Team</u>" means a boys or girls youth team that (i) is owned and operated by the Team Operator or (ii) is affiliated in some manner with the Team, or some combination thereof.

4.24.2. While the USL Academy Cup and USL Academy League are currently only operating in select age groups, the Team Operator acknowledges that one or both of such competitions may become available to additional age groups in the future, and that the obligations set forth in Section 4.24.1 apply to each of the age groups available at such time.

4.24.3. The Team Operator further acknowledges and agrees that under no circumstances may the Academy Team participate in any other league or competition that restricts the ability of the Academy Team (or its players) to participate in either the USL Academy Cup or USL Academy League.

V. FEES; COSTS; AND EXPENSES

5.1. <u>Expansion Fee</u>. In consideration of the rights granted to the Team Operator hereunder, the Team Operator shall pay a fee to the League Office in the amount set forth in Schedule 4 (the "<u>Expansion Fee</u>"). Unless otherwise set forth in Schedule 4, the Team Operator shall pay the Expansion Fee contemporaneously with the execution of this Agreement, by certified cashier's check or by wire transfer of immediately available funds. The Expansion Fee is deemed to be fully earned by the League Office upon the execution of this Agreement and is non-refundable in consideration of administrative and other expenses incurred by the League Office in granting the rights hereunder and for the League Office's lost or deferred opportunity to grant the rights to other Persons.

Participation Fee. Prior to the commencement of each Season during the Term, the Team 5.2. Operator shall pay a participation fee to the League Office (the "Participation Fee"). The Participation Fee for the Initial Season is as set forth in Schedule 4. The League Office may increase the Participation Fee each Season; provided, however, without the approval of the Board of Governors, such increase shall not exceed the greater of (a) fifteen percent (15%) of the previous Season's Participation Fee and (b) a percentage of the previous Season's Participation Fee equal to twelve percent (12%) plus the percentage rate of increase of the Consumer Price Index for the immediately preceding 12-month period (the "Maximum Participation Fee Increase"). The Participation Fee for each Season shall be determined by the League Office and communicated to the Team Operator at the earliest practicable date, and the League Office shall use good faith efforts to ensure that such is communicated no later than September 1st of the preceding year. The League Office may reasonably change the due dates and frequency of payments for the Participation Fee. All changes in the Participation Fee by the League Office shall apply to future Participation Fees only. The League Office may waive, suspend or defer any fees required to be paid by team operators or a class of team operators, in whole or in part, in its discretion at any time and in any instance or under any circumstances. Except as set forth in Section 12.3, the Participation Fee is nonrefundable.

5.3. <u>Ramp-Up Fee</u>. Prior to the commencement of the Initial Season during the Term, the Team Operator shall pay an annual fee to the League Office (the "<u>Ramp-Up Fee</u>"). The Ramp-Up Fee is as set forth in Schedule 4.

5.4. <u>Team Operator Anticipated Fees, Costs and Expenses.</u>

5.4.1. In addition to the Participation Fee the Team Operator shall also pay:

(a) all costs and expenses relating to the administration, marketing, promotion, and operation of the Team that the League Office has not undertaken pursuant to the terms of this Agreement including, without limitation, the Team Operator's office and general and administrative costs;

(b) all costs and expenses associated with the Team's travel to and attendance at Matches.

(c) all costs and expenses associated with traveling to or attending any Team Operator Training or other League meetings;

(d) all applicable fees, fines and other amounts owed to the League Office, whether under this Agreement or any related agreement;

(e) all costs and expenses relating to referees (including, but not limited to, game day officials fees, referee travel, room, meals and training);

compensation;

(f)

(g) all costs and expenses relating to players, except as set forth in Section

all costs and expenses of Team-related insurance and worker's

3.6;

(h) its share of costs related to Collective Bargaining Activities, in accordance with Section 3.6;

(i) all costs and expenses relating to the Field of Play, specifically including any equipment used therein to display temporary signage (e.g., soccer field boards);

(j) in the event that the Team Operator hosts the League's championship Match, the costs and expenses relating to the hosting, promotion, and operation of such Match;

(k) all costs and expenses associated with the Domain Name, the maintenance of the Team Site, and any re-design thereof <u>unless</u> such re-design is required by the League Office;

(1) in the event that the League Marks are added to, modified, or discontinued, all costs and expenses of transitioning from the old League Marks to the new League Marks, <u>provided that</u> the Team Operator has been notified of the change in advance and is granted a reasonably sufficient period of time in which to comply;

(m) all costs and expenses of policing and enforcing the Team Marks against unauthorized use by third parties, including any costs and expenses incurred by the League Office in furtherance thereof, subject to Section 7.5;

(n) all costs and expenses of complying with Professional League Standards and of proving compliance therewith to the satisfaction of the League Office and any applicable Governing Body; (o) all audit and reporting costs and expenses, including the cost of any equipment, systems, or services used to report information to the League Office;

(p) all costs and expenses relating to the production of a live Broadcast of each of the Home Matches and the transmission thereof to the League Office;

(q) all costs and expenses of fulfilling its obligations pursuant to any League Commercial Affiliations that are not undertaken by the League Office;

(r) any costs that, pursuant to a Board of Governors vote, are designated to be shared amongst team operators rather than borne by a single team operator (or a group of team operators) (e.g., costs of post-season travel) ("<u>Pool Payments</u>");

(s) any fines payable to the League Office or other team operators for noncompliance with League Rules;

(t) any affiliation, player registration, or other fees or fines imposed on the League by a Governing Body and passed through to the team operators; and

(u) all fees and other amounts that League Office Entities then have paid or have agreed to pay on behalf of the Team Operator to Providers.

5.5. <u>Payments to Providers</u>. The Team Operator acknowledges that the League Office makes payments to Providers on behalf of team operators in the League and charges the Team Operator fees in connection with such payments. <u>Schedule 7</u> is an anticipated list of fees that would be owed by the team operators in the League in connection with the League Office's payments to Providers in 2023. The Team Operator acknowledges that such fees have yet to be finalized and, once finalized, are subject to change, consistent with this Agreement.

5.5.1. The League Office may purchase goods or services to be used: (a) in connection with the Broadcasts on a leaguewide basis (e.g., music, graphics, archival, live stats); or (b) throughout both the League Site and the Team Site (e.g., ad server, content delivery network), the cost of which shall be allocated such that the amount payable by the League Office is twice the average amount payable by an individual team operator, unless a different allocation is determined by the Board of Governors.

5.5.2. To the extent that any of the costs set forth in Section 5.4.1 or Section 5.5.1 are to be shared between and amongst team operators, or between and amongst the League Office and team operators, or between and amongst different leagues operated by League Office Entities, the Team Operator agrees that the League Office Entities' allocation of such costs is final.

5.6. <u>Championship Match</u>. In the event that the Team is scheduled to participate in the League's championship Match, the League Office may require the Team Operator to host such Match. Any costs or expenses associated with hosting, promoting and operating the championship Match in the manner required by League Rules shall be borne exclusively by the Team Operator.

5.7. <u>No Set Off.</u> Except as explicitly permitted by this Agreement, the Team Operator may not set off or withhold payment of any amounts due to League Office Entities for any reason, including the nonperformance or alleged nonperformance by the League Office of any of its obligations hereunder.

5.8. <u>Payments</u>.

5.8.1. The Team Operator shall deliver to the League Office those documents the League Office requires from time to time to authorize the League Office to automatically debit the Team Operator's checking account for those payments due on an established schedule to League Office Entities. The current form of automatic withdrawal authorization shall be executed contemporaneously herewith. All payments made to the League Office shall be made in U.S. Dollars. All payments by the Team Operator to League Entities shall be applied in such order and for such obligations as the League Office may designate from time to time.

5.8.2. Any payment not actually received by the League Office on or before the due date thereof shall be overdue. If any payment is overdue, the Team Operator shall pay the League Office immediately upon demand, in addition to the overdue amount: (a) a late payment fee in an amount equal to five percent (5%) of the overdue amount; and (b) interest on the overdue amount from the date it was due until paid, at the rate of one and one-half percent (1.5%) per month, or the maximum rate permitted by law, whichever is less. Entitlement to such interest shall be in addition to any other remedies the League Office may have.

5.9. <u>Taxes</u>. The Team Operator shall promptly pay all Taxes incurred by Team Operator Entities pursuant to this Agreement or otherwise relating to Team Activities. In the event that any Taxes are imposed on the League Office Entities on amounts payable by the Team Operator under this Agreement, the Team Operator shall promptly reimburse the League Office Entities for all Tax payments made such that the amount of the Team Operator's payments retained by the League Office Entities, after paying the applicable Taxes, equals the full amount of the payments the Team Operator was required to make under this Agreement had the Taxes not been imposed on the League Office Entities.

VI. COMMERCIAL RIGHTS

6.1. <u>League Assets</u>. The Parties acknowledge and agree that League Office Entities are the exclusive owners of the Broadcast Rights, the League Commercial Affiliations, the League Marks, the Competition Marks, the Historic Marks, the League IP, the Manuals, and the League Site (collectively, the "League Assets"). The Team Operator also acknowledges and agrees that the League Marks, the Competition Marks, the Historic Marks, and all goodwill now or in the future pertaining to the League Marks and Competition Marks are the sole and exclusive property of the League Office Entities and that the Team Operator shall not raise or cause to be raised any questions concerning, or objections to, the validity or ownership of such League Assets on any grounds whatsoever. The Team Operator shall at any time and from time to time at its expense, make, execute, acknowledge and deliver such agreements, documents and instruments and do such other further acts (including without limitation, executing an assignment of any right, title, or interest in the League Assets) as the League Office may request or as may be necessary or appropriate in order to implement and effect fully the intentions, purposes and provisions of this Section 6.1

6.2. <u>Team Assets</u>. The Parties acknowledge and agree that the Team Operator is the exclusive owner of the Home Tickets, the Team Commercial Affiliations, the Team Marks, the Team Site, and the Team Lists (collectively, the "Team Assets"). Notwithstanding the foregoing, the Team Operator agrees that the League Office may exercise control over and exploit, subject to the terms and conditions of this Agreement including but not limited to Section 6.3, the Team Assets in connection with League Activities. The Team Operator shall at any time and from time to time at its expense, make, execute, acknowledge and deliver such agreements, documents and instruments and do such other further acts (including without limitation, executing a license agreement containing such terms and conditions as are customary in such licenses) as the League Office may request or as may be necessary or appropriate in order to implement and effect fully the intentions, purposes and provisions of this Section 6.2.

6.3. <u>Commercial Affiliations</u>.

6.3.1. League Commercial Affiliations.

(a) The League Office has the right to exploit, free of charge, the following sponsorship and advertising rights and inventory relating to the Team for the purpose of fulfilling League Commercial Affiliations or for the League's own advertising: (i) ten percent (10%) of the Core Commercial Inventory; (ii) Negotiation Rights; (iii) Team Lists; (iv) Collective Use Rights; and (v) a reasonable number of Home Tickets, <u>provided that</u> (A) the rights and obligations applicable to the Team Operator are generally applicable to team operators in the League (or an identifiable subset of team operators, if applicable); and (B) in those instances set forth in Section 9.7, the terms of the League Commercial Affiliation are approved by the Board of Governors.

(b) Notwithstanding any other provision in this Agreement, the Parties understand and agree that the League Office owns and may exploit, free of charge, all sponsorship and advertising rights and inventory relating to the Championship Match for whatever purpose, including for the purpose of fulfilling League Commercial Affiliations or for the promotion of the League.

6.3.2. Team Commercial Affiliations.

(a) The Team Operator may enter into Team Commercial Affiliations for (i) the front of the jersey; (ii) the Core Commercial Inventory and Broadcast Inventory allocated to the Team in Section 6.3.3; and (iii) Ancillary Commercial Inventory; <u>provided that</u>, in each instance, the Team Operator shall only execute agreements in accordance with League Rules (which may include specifications of form as well as requirements for the prior approval of the League Office). The Team Operator agrees to promptly provide the League Office with such details as the League Office may from time to time reasonably request in connection with the Team Operator's activities hereunder.

(b) From time to time, the League Office may reserve or otherwise restrict, on an exclusive or non-exclusive basis, certain Product Categories for sale to League Commercial Affiliates. The Team Operator shall comply with such reservations or restrictions, <u>except that</u> if the League Office adds or amends any reserved or restricted Product Categories such that the Team Operator's compliance therewith would cause it to breach a pre-existing Team Commercial Affiliation in the same or a substantially similar Product Category, the Team Operator may nonetheless fulfill its obligations under such Team Commercial Affiliation for the duration of the remaining term of such Team Commercial Affiliation; however, the Team Operator may not exercise any optional extensions or renewals of such Team Commercial Affiliation. As of the Effective Date, restricted or reserved Product Categories are included in Schedule 10 hereto.

6.3.3. Inventory Allocation.

(a) The League Office will designate and categorize the Core Commercial Inventory in writing. The League Office will allocate at least ninety percent (90%) of each category of Core Commercial Inventory (e.g., in-stadium print collateral; space on the Team Site) to the Team Operator, with the remainder being reserved to the League Office.

(b) In event the League Office wishes to exploit a portion of the Teamallocated Core Commercial Inventory or the Ancillary Commercial Inventory, the League Office shall submit to the Board of Governors a proposal to team operators in the League, which shall be accepted provided that the team operators (as a whole) are compensated with cash or inventory of reasonably equivalent value (as a whole). (c) The League Office will allocate to the Team Operator at least seventyfive percent (75%) of the Broadcast Inventory on regular season Matches Exhibited: (i) via Local Broadcast, provided that the Local Broadcast does not conflict with any national Broadcast; and (ii) via Digital Distribution, provided that (x) the Exhibition of such Match does not conflict with any national Broadcast and (y) the Match has not been designated as a Featured Match.

(d) With respect to any playoff Match or Championship, the Team Operator's allocation of Broadcast Inventory is the same as any regular season Match, <u>except that</u> the League Office may elect, in its discretion, to pay for the production of the Broadcast and to retain all of the Broadcast Rights (including the Broadcast Inventory) with respect thereto.

(e) If unsold, the Core Commercial Inventory and Broadcast Inventory allocated to the League Office (or any portions thereof) may be released back to the Team Operator on an annual basis (or such other period of time as the League Office may designate) at the earliest practicable date prior to the commencement of the applicable Season.

6.4. <u>Broadcasts</u>. The League Office hereby appoints the Team Operator as its exclusive agent with authority to license the Local Broadcast Rights; provided that (i) the Team Operator shall act as the agent of the League Office in any negotiations relating to any of the foregoing and shall not represent or hold itself out as acting on its own behalf; (ii) the Local Broadcast Rights are made subservient to any national television Broadcast Rights; and (iii) the Team Operator shall only execute agreements in accordance with League Rules (which may include specifications of form as well as requirements for prior approval from the League Office). The Team Operator agrees to promptly provide the League Office with such details as the League Office may from time to time reasonably request in connection with the Team Operator's activities hereunder. For the avoidance of confusion, the Local Broadcast Rights do not include the right to Exhibit any Match via Digital Distribution.

6.5. <u>Ticket Agents</u>. The Team Operator hereby appoints the League Office as its exclusive agent (to the exclusion of the Team Operator) with authority to enter into arrangements with Ticket Agents relating to the sale of Home Tickets; provided that the League Office shall act as the agent of the Team Operator in any negotiations relating to the foregoing and shall not represent or hold itself out as acting on its own behalf. The League Office may require that all revenues from the sale of Home Tickets shall be paid from the Ticket Agent directly to the League Office, and the League Office shall distribute such funds to the Team Operator in accordance with an established set of settlement procedures.

6.6. <u>Licensed Products</u>.

6.6.1. The Team Operator hereby grants to the League Office, during the Term of this Agreement, the exclusive right (to the exclusion of others as well as the Team Operator representing itself) to enter into Licensing Arrangements. Subject to the foregoing sentence, the Team Operator may sell Licensed Products in its Stadium and in its Team Store, in each instance subject to League Rules.

6.6.2. The Team Operator further empowers the League Office to negotiate the terms of such Licensing Arrangements within the parameters agreed upon between the League Office and the Board of Governors from time to time. In the event that the Team Operator is approached directly by a third party about the right to manufacture, sell or distribute Licensed Products, the Team Operator shall refer such third party to the League Office. The Team Operator agrees that during the Term of this Agreement, it will not negotiate with any other Person to represent it in any capacity in connection with the manufacture, sale or distribution of Licensed Products.

6.6.3. In consideration of the League Office's performance of its duties under this

Section 6.6, the League Office shall retain a commission in the amount of twenty percent (20%) of the Net Licensing Revenues.

6.7. Approved Providers; Equipment.

6.7.1. The League Office may require the Team Operator to purchase or lease certain types, models, or brands of facilities, services, goods, supplies, and equipment (including for sale to the general public) designated in the Manuals or approved in writing by the League Office (i) from Providers designated or approved by the League Office in writing or (ii) in accordance with specifications provided by the League Office in writing; <u>provided that</u> such services, goods, supplies, and equipment are (a) available on commercially reasonable terms and (b) are reasonably required for Team Activities. Those services, goods, supplies and equipment with restricted (or exclusive) Providers as of the Effective Date are included in Schedule 8, although the Team Operator further acknowledges that such are subject to change, consistent with this Agreement.

6.7.2. Subject to Section 6.7.1, the Team Operator acknowledges that the Provider of such services, goods, supplies, and equipment may be a League Office Entity, and that League Office Entities may receive payments or other compensation from Providers on account of such Providers' dealings with the Team Operator or other team operators. The League Office Entities may use all amounts so received for any purpose they deem appropriate.

6.7.3. Notwithstanding Sections 6.7.1 or 6.7.2 above, throughout the Term, the Team Operator shall also agree to use the following, the agreements for which shall be in the same form as those required of other teams in the League:

(a) vendors designated by the League Office to provide the following categories of software products: (i) team mobile application and/or data warehouse (initially, anticipated to be FanThreeSixty) and (ii) software to create short form video content or clips (initially, anticipated to be WSC);

(b) a ticketing platform or instance designated by the League Office (initially, the Team Operator would be required to use a version of a SeatGeek ticketing system);

(c) the provider designated by the League Office to produce a broadcast of each home match in accordance with League Rules (initially, anticipated to be VISTA Worldlink);

(d) for any Team identified online store, the Team Operator would use the online retail platform recommended by the League Office; and

(e) the Team Operator would use any stadium point-of-sale system recommended by the League Office.

With respect to paragraphs 6.7.3(b), 6.7.3(d), and 6.7.3(e) above, to the extent that such ticketing, online retail, or stadium point-of-sale systems integrate with a payment processor recommended by the League Office, the Team Operator would also ensure that such system used the recommended payment processor.

6.8. <u>Stadium Facilitation Reimbursement</u>. In the event that (a) the League Office helps facilitate the construction, renovation, or development of a stadium and (b) the Team Operator ultimately uses such stadium as its Home Stadium, the Team Operator agrees to reimburse the League Office for any documented, out-of-pocket costs or expenses incurred in connection therewith, plus twenty percent (20%). Such costs or expenses may include, for example, the cost of economic impact studies, stadium

consultants, and any associated travel and per diems.

VII. LEAGUE AND TEAM MARKS

7.1. <u>Selection of League Marks</u>. The League Marks are subject to change from time to time in the League Office's sole and absolute discretion. For the avoidance of the doubt, the League Mark may also, from time to time, be a Composite Mark.

7.2. <u>License and Use of League Marks</u>.

7.2.1. The League Office hereby grants to the Team Operator a non-exclusive, royalty free license to use, display, and reproduce the League Marks and Competition Marks in connection with Team Activities, and the limited right to sublicense such rights in accordance therewith, <u>provided that</u> the Team Operator only uses the League Marks and Competition Marks designated by the League Office and as permitted by League Rules.

7.2.2. The Team Operator acknowledges and agrees that League Rules may require the Team Operator to display the League Marks and Competition Marks on promotional materials and merchandise, including, without limitation, team uniforms, game programs, and signage. The Team Operator shall comply with all such requirements at its sole cost and expense.

7.2.3. The League Office may, from time to time and in its sole discretion, add to, modify or discontinue use of any or all of the current League Marks or Competition Marks. The League Office shall inform the Team Operator of such changes in writing and, after a reasonable transition period, the Team Operator agrees to operate under the revised League Marks or Competition Marks, as applicable.

7.3. <u>Selection of Team Marks</u>.

7.3.1. The Team Marks shall be selected by the Team Operator and acceptable to the League Office in its sole discretion. The Team Operator's current Team Marks, if any, are included in Schedule 5. The Team Operator shall submit all proposed revisions or additions to the Team Marks to the League Office for approval, which the League Office may withhold in its sole discretion. No revisions to the Team Marks are permitted without the written consent of the League Office. In the event that the Team Operator changes its official Team name or any Team Marks related to that name (the "<u>Historic Marks</u>"), the Team Operator shall, upon request from the League Office, assign such Historic Marks and their goodwill to the League Office without charge.

7.3.2. The Team Operator shall be solely responsible for complying with all trademark Laws in its selection and use of Team Marks. The League Office assumes no responsibility and shall not be held liable for the Team Operator's selection of Team Marks or for the effectiveness or desirability thereof.

7.4. License and Use of Team Marks.

7.4.1. The Team Operator hereby grants to the League Office a non-exclusive, royalty free license to use, display, and reproduce the Team Marks and Player Likenesses in connection with League Activities, and the limited right to sublicense such rights in accordance therewith.

7.4.2. The Team Operator hereby grants to each other team in the League the right to use the Team Marks and Player Likenesses in connection with the promotion of Matches including,

without limitation, the right to display and reproduce the Team Marks and Player Likenesses on game programs and tickets, in accordance with League Rules.

7.4.3. Other than to promote Matches, the Team Operator shall not use the Team Marks in combination with the marks or logos of any other soccer teams (including teams in the League).

7.4.4. The Team Operator shall not pledge or encumber any interest in the Team Marks without the prior written consent of the League Office, nor shall the Team Operator pledge or encumber its right to use the League Marks.

7.5. Infringement of Marks.

7.5.1. Each party shall promptly notify the other party of any actual or potential infringement, counterfeiting, or other unauthorized use of the Team Marks or League Marks by any other Person (an "<u>Infringement</u>") of which it becomes aware.

7.5.2. The Team Operator shall have the right, in its discretion, to enforce its rights in any of the Team Marks, including to bring action with respect to any Infringement. Notwithstanding the foregoing, if within ten (10) days following either party's receipt of a notice provided under Section 7.5.1, the Team Operator does not initiate legal action with respect to any Infringement, or if the Team Operator subsequently decides not to proceed with any such action, the League Office shall have the right, but no obligation, to bring or take any such action as it determines is necessary in its reasonable business judgment to halt any such Infringement and to control the conduct of such enforcement action, including settlement.

7.5.3. Notwithstanding Section 7.5.2, the League Office shall have the first right, but not the obligation, to enforce rights in the Team Marks if: (a) the Infringement of the Team Marks is also (or is related to) the infringement of other team marks in the League and (b) the League Office has a good faith belief that such Infringement has impaired or will impair the value of the applicable marks or otherwise adversely affect the League Office's rights under this Agreement (or similar agreements with other team operators). In such case, subject to the approval of the Board of Governors in accordance with Section 9.7.2(c) (by Supermajority Vote), the League Office may immediately bring or take any such action as it determines is necessary in its reasonable business judgment to halt any such Infringement and to control the conduct of such enforcement action, including settlement.

7.5.4. Both Parties shall provide such assistance as may be reasonably requested by the party taking action against any alleged Infringement in accordance with this Section 7.5 (the "<u>Enforcing</u> <u>Party</u>"), in connection with any such enforcement action (including being joined as a party to such action as necessary to establish standing). Any monetary recovery resulting from such enforcement action shall first be used to pay the legal expenses of the Enforcing Party and then to reimburse any legal expenses incurred by the other party in cooperating in such action as requested by the Enforcing Party, with the balance, if any, of such monetary recovery to be delivered to the Enforcing Party.

VIII. OWNERSHIP

8.1. <u>Corporate Structure</u>.

8.1.1. The Team Operator shall maintain its corporate structure such that it has a Principal Owner at all time. The initial Principal Owner is included on Schedule 2. A current copy of the Team Operator's governing documents and state registration documents must be submitted and kept on file with the League Office, and any change to the governing documents of the Team Operator must be

approved by the League Office in writing prior to taking effect.

8.1.2. The Team Operator shall, directly or indirectly, maintain sole ownership of any assets, rights, and agreements that are necessary or incidental to the operation of the Team (including (i) the Team Assets; (ii) the Stadium Lease; (iii) the Domain Name; (iv) the player contracts; (v) the coaching staff contracts; and (vi) the information or other intellectual property developed in connection with the Team) and shall not Transfer (including by pledging or encumbering) any of such assets, rights, or agreements without the express written consent of the League Office. Notwithstanding the foregoing, the League Office shall not unreasonably withhold its consent to the pledge or hypothecation of such assets, rights, or agreements in connection with any financing arrangements obtained by the Team Operator (provided such financing arrangements comply with League Rules). However, for the avoidance of confusion, neither this Agreement nor any Equity Interest may be Transferred other than in accordance with Article X.

8.1.3. From time to time as requested by the League Office, the Team Operator shall provide the League Office with the following, certified by the president or Principal Owner to be true and correct: (a) details on its ownership structure, including ownership and capitalization tables; (b) basic details (e.g., contact information) on its Equity Owners; (c) confirmation that the Team Operator continues to comply with Section 8.1.2 above; (d) details regarding debt or financing arrangements of the Team Operator Entities; and (d) such other information as the League Office may reasonably request, in the format so requested, with respect to the ownership and structure of the Team Operator.

8.2. <u>Equity Owners</u>. The Team Operator hereby represents to the League Office, and the League Office is entering into this Agreement in reliance upon such representation, that the Equity Owners identified on Schedule 6 are all of the holders of Equity Interests, in such percentages as set forth thereon.

8.3. <u>Guaranty</u>. All Equity Owners of five percent (5%) or more of the outstanding Equity Interests shall execute a guaranty, indemnification and acknowledgment of the Team Operator's obligations under this Agreement in the form required by the League Office (the "Guaranty"). If an Equity Owner of five percent (5%) or more of the outstanding Equity Interests is an entity, each of such Equity Owner's legal or beneficial owners must execute the Guaranty as well.

IX. BOARD OF GOVERNORS

9.1. <u>General</u>. The Board of Governors (the "Board of Governors") is made up of (i) one (1) representative of the League Office and (ii) the principal owner of each team operator unconditionally obligated to field a team in an upcoming season (each, a "Governor"). The Principal Owner shall be the Governor representing Team Operator. The Board of Governors may consider items delegated thereto by the terms of this Agreement or by the League Office. Subject to Section 9.7, the League Office may choose which matters to bring (or not bring) to the Board of Governors for a vote in its sole discretion.

9.2. <u>Voting</u>. Each Governor has one vote on the Board of Governors. In the event that a team (including, as applicable, the Team) is Controlled by a team in another league, the League Office may, in its discretion, exclude that team's Governor from certain votes taken by the Board of Governors, except that (a) the League Office may not exclude such team's Governor from any vote related to the basic structure of the competition and (b) to the extent a team's Governor is excluded from a vote, such team would not be bound by the result.

9.3. <u>Meetings</u>. The Board of Governors shall meet at such time and at such place as the League Office may designate. Meetings of the Board of Governors may be held either in person or by

means of telephone or video conference or other communications device that permits all Governors participating in the meeting to hear each other, at the offices of the League Office, or such other place as may be determined from time to time by the League Office. The League office shall provide prior notice of all meetings of the Board of Governors, with each such notice setting forth to the extent reasonably possible the purpose for which the meeting is called. A Governor may be required to attend up to two inperson meetings each year, and the Principal Owner may be required to attend one of such in-person meetings.

9.4. <u>Quorum</u>. Governors from a majority of eligible teams constitute a quorum for the transaction of any business by the Board of Governors, unless otherwise provided in rules adopted by the Board of Governors.

9.5. <u>Action By Written Consent</u>. Notwithstanding anything herein to the contrary, any action of the Board of Governors may be taken without a meeting if a consent in writing, setting forth the action to be taken, is signed by Governors from two third of the team operators and the League Office (if a Majority Vote is required) or unanimously (if a Super Majority Vote is required). Such consent shall have the same force and effect as a vote at a meeting where a quorum was present.

9.6. <u>Incorporation Into Agreement</u>. The Team Operator hereby agrees to be bound by the acts, decisions, and directives of the Board of Governors, and the acts or decisions of the Board of Governors shall be considered to have been incorporated into this Agreement as if fully set forth herein (and, in the case of conflict, shall supersede this Agreement). To the extent reasonably required, the Team Operator shall enter into such agreements and cause such actions to be performed as may be reasonably required to comply with such acts, decisions, and directives. The League Office may enforce the decisions of the Board of Governors through those means already made available to it (including, without limitation, those enforcement methods described in the Manuals or herein) or otherwise approved by the Board of Governors.

9.7. <u>Limitation of Authority</u>. The League Office shall not establish League Rules or otherwise enter into agreements with third parties relating to the following matters without the approval of the Board of Governors, which shall be achieved through either a Majority Vote or Supermajority Vote, as set forth below.

9.7.1. The following matters require a Majority Vote:

(a) The basic structure of the competition (e.g., number of games and season length) <u>except that:</u> (i) with respect to the first Season of the League, the League Office shall determine the structure of the competition (which it currently anticipates will commence in September of 2023, break in late December 2023 and January 2024, and conclude in late May/early June of 2024)¹ and (ii) with respect to any Season thereafter, the League Office shall use good faith efforts to structure the competition in a substantially similar manner as the previous season;

(b) The amount of any Pool Payments and the parameters within which League Office Entities may hold and distribute such funds;

(c) The parameters within which the League Office may license or Transfer

¹ For reference, the "international calendar" that the League Office presently intends to adopt is different than the calendar currently in place with the USL Championship and USL League One (each of which generally runs from March through October).

the Team Lists;

(d) The increase in the Participation Fee, from one Fiscal Year to the next, by more than the Maximum Participation Fee Increase;

(e) Such other matters as the League Office may present to the Board of Governors from time to time that do not require a Supermajority Vote.

9.7.2. The following matters require a Supermajority Vote:

(a) The establishment or amendment of any revenue or profit-sharing program or programs between or amongst team operator entities and League Office Entities;

(b) Any League Commercial Affiliation for an exclusive Product Category other than soccer balls either (i) on a leaguewide basis or (ii) which would grant to such Commercial Affiliate the right to exploit the Team Marks in isolation from other teams in the League;

(c) Any Legal Proceeding filed on behalf of teams to protect against the wrongful use of Team Assets by third parties;

Operator Entities; and

Any financing requirements or debt limitations relating to the Team

Such other matters as the Board of Governors may determine require a

Supermajority Vote.

9.8. <u>Committees</u>. The League Office may, from time to time, establish committees comprised of Governors. Such committees may provide non-binding advice to the League Office and the Board of Governors, and in no instance may the authority of the League Office or the Board of Governors be

delegated to any such committee.

X. TRANSFERS

10.1. No Transfers Without Approval.

(d)

(e)

10.1.1. The Team Operator understands and acknowledges that the rights and duties in this Agreement are personal to the Principal Owner and the other Equity Owners of the Team Operator, and that the League Office is entering into this Agreement in reliance on the business skill, financial capacity, and the personal character of such Equity Owners. Accordingly, the Team Operator shall not Transfer any Equity Interest, or this Agreement, or any of its rights or obligations under this Agreement, nor shall any Equity Owner Transfer any Equity Interest without, in each instance, seeking and receiving the approval of the League Office.

10.1.2. The League Office will permit the Team Operator or any Equity Owner named in Schedule 6 as of the Effective Date (or any transferee Equity Owner subsequently approved by the League Office) to engage in the Transfer of Equity Interests, provided that: (a) such Transfer would not result in a Change of Control of the Team Operator; (b) the proposed transferee would not be restricted by League Rules from becoming an Equity Owner; (c) the Team Operator gives the League Office at least thirty (30) days' advance written notice of the proposed Transfer (including the identity and contact information for any proposed transferee and any other information the League Office may require in order to review the proposed Transfer); (d) the Team Operator follows the then-current procedure for

processing Transfers; and (e) the Team Operator executes any documents required by the League Office for processing Transfers.

10.1.3. <u>Change of Control Transfers</u>. If the proposed Transfer is of this Agreement or of an Equity Interest that would result in a Change of Control, the League Office will process the application for a Change of Control according to its then-current procedures, including review of criteria and requirements regarding upgrading the Home Stadium, credit, background investigations, operations ability and capacity, prior business dealings, relation to community, guarantees, and other factors concerning the proposed transferee (and, if applicable, its owners) the League Office deems relevant (including its belief that the Team will be operated by the proposed transferee at its then-current, or higher, level). The League Office may review all information regarding the Team given to the proposed transferee copies of any reports that the Team Operator has given the League Office or the League Office has made regarding the Team. The League Office may approve or reject the Transfer in its sole discretion and may require conditions to be met before or concurrently with the effective date of the Transfer, including without limitation:

(a) the Transferee has the necessary business experience, aptitude, and financial resources to operate the Team and meets the then-applicable standards set forth in the League Rules.

(b) the Team Operator has paid all fees and other amounts owed pursuant to Section 4.5; has submitted all required reports and statements; and has not violated any material provision of this Agreement or any other agreement with League Office Entities during both the sixty (60) day period before the Team Operator requested the League Office's consent to the Transfer and the period between the Team Operator's request and the effective date of the Transfer;

(c) the Transferee and its owners shall (if the Transfer is of this Agreement), or the Team Operator and its Equity Owners shall (if the Transfer would result in a Change of Control), sign the League Office's then current form of franchise agreement and related documents (including guarantees and assumptions of obligations), any and all of the provisions of which may differ materially from any and all of those contained in this Agreement, including the Participation Fee, and the term of which franchise agreement will be equal to the remaining unexpired portion of the Term, except that, for the avoidance of confusion, the League Office would not charge any additional expansion fee;

(d) the Team Operator and its transferring Equity Owners sign the thencurrent form of termination agreement and a general release, in a form satisfactory to the League Office, of any and all claims against any League Office Entities and their respective Personnel;

(e) the Team Operator shall pay to the League Office the Change of Control Fee included on Schedule 4.

10.2. <u>Transfer Upon Death</u>.

10.2.1. Upon the death of an Equity Owner, his or her Equity Interests may be Transferred in accordance with such individual's will or, if such individual dies intestate, in accordance with the Laws of intestacy governing the distribution of such individual's estate, <u>provided that</u> the Transfer of the deceased's Equity Interest is (i) to an approved third party or a Family Member and (ii) such Transfer would not cause the Team Operator to be in violation of League Rules. However, in the event of the death of the Principal Owner and the transfer of Control to a Family Member who is not, in the reasonable discretion of the League Office, sufficiently knowledgeable or experienced to act as a Principal Owner, the League Office may appoint a trustee, chosen with the consent of such Family Member (not to be unreasonably withheld), to act as the Principal Owner (for the benefit of the Family Member). The trustee (and any agreed upon successor trustees) shall act in this capacity until such time as either (a) the League Office approves the Family Member to act as the Principal Owner or (b) Control is Transferred to a new Principal Owner, in either instance not to exceed two (2) years without the consent of the League Office.

10.2.2. Upon the death of any Equity Owner of more than with an interest ten percent (10%) of the Equity Interests, any Transfer of such Equity Interests shall be subject to the same terms and conditions as any *inter vivos* Transfer.

10.3. League Office Transfers. The League Office may Transfer this Agreement and all or any part of its rights or obligations herein to any Person, so long as the League Office Transfers all or any part of its rights or obligations in all similar agreements with team operators to such Person (the "Other Agreements"). Upon such Transfer, any designated assignee of the League Office shall become solely responsible for all obligations of the League Office under this Agreement and the Other Agreements from the date of assignment. In addition, and without limitation to the foregoing, the Team Operator expressly affirms and agrees that the League Office may sell its assets (including the League Assets); may sell its securities in a public offering or in a private placement; may merge, acquire other corporations, or be acquired by another entity; and may undertake a refinancing, recapitalization, leveraged buy-out or other economic or financial restructuring, in each case without any prior consent or approval by the Team Operator.

XI. INDEMNIFICATION

11.1. <u>Indemnification</u>. The Team Operator shall indemnify, defend, and hold harmless the League Office Entities and their respective Personnel (the "League Office Indemnitees") against any and all Losses that are incurred by the League Office Indemnitees in connection with the operation of the Team (including the operation of the Team by the League Office Entities pursuant to Section 12.6, and notwithstanding any claims that the League Office Indemnitees were negligent) or arising out of or related to any third-party claim alleging:

11.1.1. breach or non-fulfillment of any provision of this Agreement by the Team Operator or the Team Operator's Personnel;

11.1.2. any negligent or more culpable act or omission of the Team Operator or its Personnel (including any reckless or willful misconduct) in connection with the operation of the Team (including the operation of the Team by the League Office Entities pursuant to Section 12.6) or the performance of its obligations under this Agreement;

11.1.3. any bodily injury, death of any person or damage to real or tangible personal property caused by the negligent or more culpable acts or omissions of the Team Operator or its Personnel (including any reckless or willful misconduct);

11.1.4. any failure by the Team Operator or its Personnel to comply with any applicable Laws in the performance of its obligations under this Agreement; or

11.1.5. that the use of the Team Marks infringes upon the rights of a third party.

11.2. <u>Notice of Third-party Claims</u>. The League Office shall give notice to the Team Operator (a "Claim Notice") within fourteen (14) days after obtaining knowledge of any Losses or discovery of

facts on which the League Office Indemnitees intend to base a request for indemnification under Section 11.1. The League Office's failure to provide a Claim Notice to the Team Operator under this Section 11.2 does not relieve the Team Operator of any liability that the Team Operator may have to the League Office Indemnitees, but in no event shall the Team Operator be liable for any Losses that result directly from a delay in providing a Claim Notice, which delay materially prejudices the defense of the related third-party claim. The Team Operator's duty to defend applies immediately, regardless of whether the League Office Indemnitees have paid any sums or incurred any detriment arising out of or relating, directly or indirectly, to any third-party claim.

11.3. <u>The League Office's Control of Defense</u>. Notwithstanding anything to the contrary in this Article XI, the League Office may employ, at any time, separate counsel to represent it; provided, that (i) the League Office shall be solely responsible for the costs and expenses of any such separate counsel and (ii) the Team Operator shall remain responsible to the League Office for any Losses indemnified under this Article XI.

11.4. <u>Settlement of Indemnified Claims by the Team Operator</u>. The Team Operator shall give prompt written notice to the League Office of any proposed settlement of a claim that is indemnifiable under Section 11.1. The Team Operator may not, without the League Office's prior written consent, settle or compromise any claim or consent to the entry of any judgment regarding which indemnification is being sought hereunder.

XII. DEFAULTS AND TERMINATION

12.1. Termination Rights of the League Office.

12.1.1. <u>Subject to Cure</u>. The League Office may terminate this Agreement by written notice to the Team Operator, effective on the date stated in such notice or the earliest date permitted by applicable Law, if:

(a) The Team Operator fails to pay the League Office or any of its Affiliates any fees or other amounts due under this Agreement or any other agreement between the Team Operator and the League Office or its Affiliates and does not cure that default within ten (10) days after delivery of the written notice of default;

(b) The Team Operator fails to pay when due any financial obligation to a Provider and does not cure that default within thirty (30) days after delivery of the written notice of default;

(c) The Team Operator fails to comply with any other provision of this Agreement or League Rules and does not cure that default within thirty (30) days after delivery of the written notice of default;

(d) The Team Operator fails to comply with any other agreement with League Office Entities relating to Team Activities and does not cure that default within thirty (30) days (or such shorter time period that the other agreement specifies for curing that default) after delivery of the written notice of default;

(e) The Team Operator does not buy, maintain, or send evidence to the League Office of any required insurance or workers compensation coverage and does not cure that default within ten (10) days after delivery of the written notice.

(f) The Team Operator or any Equity Owner Transfers or attempts to Transfer an Equity Interest in violation of Article X and does not cure that default within thirty (30) days after delivery of the written notice of default;

(g) The Team Operator fails to timely pay any Taxes to a Governmental Authority, unless the Team Operator is in good faith contesting its liability for those Taxes or have received an extension from the applicable Governmental Authority of the time within which to make such payments, and does not cure that default within thirty (30) days after delivery of the written notice of default;

12.1.2. <u>Upon Notice</u>. The League Office may terminate this Agreement immediately, without giving the Team Operator an opportunity to cure the default, effective upon delivery of written notice to the Team Operator (or such later date as required by applicable Law) if:

(a) The Team Operator or any Guarantor admits its inability to pay its debts as they become due or makes a general assignment for the benefit of creditors;

(b) The Team Operator or any Guarantor commences or consents to any case, proceeding, or action seeking: (i) reorganization, arrangement, adjustment, liquidation, dissolution, or composition of debts under any Law relating to bankruptcy, insolvency, reorganization, or relief of debtors; or (ii) appointment of a receiver, trustee, custodian, or other official for any portion of its property;

(c) The Team Operator or any Guarantor takes any corporate or other action to authorize any of the actions set forth above in Section (a) or (b);

(d) Any Legal Proceeding against the Team Operator or any Guarantor is commenced seeking an order for relief against it as debtor, or seeking reorganization, arrangement, adjustment, liquidation, dissolution, or composition of it or its debts under any Law relating to bankruptcy, insolvency, reorganization, or relief of debtors, or seeking appointment of a receiver, trustee, custodian, or other official for it or any portion of its property, and such case, proceeding, or other action: (i) results in an order for relief against it that is not fully stayed within seven (7) business days after being entered; or (ii) remains un-dismissed for forty-five (45) days;

(e) An attachment remains on assets of the Team Operator or the Guarantor for at least thirty (30) days;

(f) The Team Operator or the Principal Owner fails, within sixty (60) days after the entry of a final judgment against it in any amount exceeding Fifty Thousand Dollars (\$50,000), to discharge, vacate, or reverse the judgment, to stay its execution, or, if appealed, to discharge the judgment within thirty (30) days after a final adverse decision in the appeal;

(g) The Team Operator loses the right to operate at the Home Stadium, except in accordance with this Agreement, or has not entered into an approved Stadium Lease for the upcoming Season by August 1 of the preceding calendar year;

(h) The Team Operator knowingly maintains false books or records, or knowingly submits any false reports or information to the League Office;

(i) The Team Operator fails to provide and maintain the Letter of Credit;

(j) The Team Operator ceases to operate the Team or it is reasonable, under the facts and circumstances, to conclude that the Team Operator does not intend to continue to operate the Team in the League;

(k) The Team Operator fails to present the Team at the time and place it is scheduled to play in a Match or fails to play and complete any such Match, unless such failure arises out of acts or events beyond its reasonable control, including acts of God, labor disputes, unavoidable accidents, fire, riot or civil commotion, or government actions or decrees;

(1) The Team Operator fails to comply in any material respect with all applicable Laws, including Anti-Terrorism Laws, (including if such failure by the Team Operator or any of its Equity Owners results in the Team Operator's or the Principal Owner's assets, property or interests being blocked under any Laws relating to terrorist activities).

(m) The Team conducts a practice or game without approved workers compensation insurance;

(n) Contrary to the terms of this Agreement, the Team Operator intentionally and knowingly discloses or divulges the contents of League Rules or other Confidential Information provided to the Team Operator by the League Office;

(o) The Team Operator commits three (3) or more material defaults under this Agreement in any twelve (12) month period, whether or not each such default has been cured after notice, or the Team Operator materially violates a material League Rule on three (3) or more separate occasions in any twelve (12) month period; or

(p) The League Office discovers that the Team Operator or the Principal Owner has made material misrepresentations to League Office Entities in connection with obtaining the right to enter into this Agreement or in conducting Team Activities.

(q) The Team Operator fails to meet the Minimum Performance Standard in four out of any six consecutive Fiscal Years.

12.1.3. <u>General</u>. In the event that the League Office issues a termination notice to the Team Operator, the Team Operator expressly acknowledges that:

(a) In any Legal Proceeding in which the validity of the League Office's termination of this Agreement is contested, it may cite and rely upon all of the Team Operator's defaults or violations of this Agreement, not only the defaults or violations referenced in any written default notice sent to it.

(b) No such termination notice will relieve the Team Operator of its obligations that survive termination of this Agreement, including its de-identification and indemnification obligations.

12.2. <u>Termination Rights of the Team Operator</u>.

12.2.1. Upon any material default hereunder by the League Office, the Team Operator may terminate this Agreement by giving written notice of termination stating the nature of such material default to the League Office at least thirty (30) days prior to the effective date of termination; <u>provided</u>, <u>however</u>, that the League Office may avoid termination by initiating a remedy to cure such material

default, curing it to the reasonable satisfaction of the Team Operator, and by promptly providing proof thereof to the Team Operator within the applicable time period set forth above; <u>provided further</u>, <u>however</u>, that if the League Office cannot reasonably correct the default within this thirty (30) day period but, prior to the expiration of such period, it provides the Team Operator with reasonable evidence of its effort to correct the default within a reasonable time period, then the cure period will run through the end of such reasonable time period.

12.2.2. If any such default is not cured within the specified time period, or such longer time period as the Team Operator may allow or applicable law may require, this Agreement shall terminate without further notice to the League Office, effective immediately upon the later of (i) the conclusion of the then-current Season and (ii) the expiration of the applicable time period.

12.3. Additional Termination Rights of the League Office. The League Office may disband the League and terminate this Agreement at any time after the completion of the Initial Season, in its sole discretion, upon sixty (60) days written notice to the Team Operator if either (a) the Governing Body which currently sanctions the League (or which sanctioned the League at any time during the prior twenty-four (24) months), if any, ceases or refuses to sanction the League; or (b) the League Office or League Office Entities incur substantial and continuing losses as determined by the League Office in its reasonable discretion, which renders the economic viability of the League or League Office Entities unsustainable. Upon any termination by the League Office pursuant to this Section 12.3, the League Office shall refund the Team Operator the full amount of any Participation Fees previously paid with respect to the upcoming Season, if any, and return the full Letter of Credit Security (or whatever portion of the Letter of Credit is then remaining, if applicable). The refund of such Participation Fees and return of the League Office terminates this Agreement pursuant to this Section 12.3.

12.4. 2024 Sanctioning.

12.4.1. The Parties understand and acknowledge that the 2024-2025 Season will be the first Season of the League, and that the League Office is in the process of obtaining commitments from other teams to participate in the League and, in connection therewith, developing an application to receive sanctioning (or provisional sanctioning) by USSF as either a Division 1 or Division 2 women's outdoor professional league ("<u>Sanctioning</u>"). If, for whatever reason, the League does not receive Sanctioning for the 2024-2025 Season, the League Office may delay the start of the League and submit an application for Sanctioning for the 2025-2026 Season. If, for whatever reason, by December 1, 2025, the League still has not been sanctioned as a professional league by USSF, either Party may terminate this Agreement by providing written notice of such termination to the other Party and, upon such occurrence, the League Office shall promptly return to the Team Operator the entirety of any Participation Fees already delivered to the League Office.

12.4.2. With respect to any failure to receive Sanctioning or any termination of this Agreement resulting therefrom, (a) the remedies set forth in Section 12.4.1 above represent the Team Operator's sole and exclusive remedy and (b) the League Office shall otherwise have no liability to the Team Operator arising therefrom.

12.5. Effect of Termination.

12.5.1. Upon termination or expiration of this Agreement, all rights granted hereunder to the Team Operator shall terminate and:

(a) The Team Operator shall immediately cease to operate the Team. The

Team Operator and its Equity Owners shall not thereafter, directly or indirectly, represent to the public or hold themselves out as owners of the Team;

(b) The Team Operator shall immediately cease any further use the League Marks in any capacity;

Office Entities;

(c) The Team Operator shall promptly pay all sums owing to the League

(d) The Team Operator shall immediately turn over to the League Office any copies (including digital copies) of the Manuals, as well as all videos, records, files, instructions, and other materials in the Team Operator's possession relating to the operation of the Team (all of which are acknowledged to be the property of the League Office Entities), and shall retain no copy or record of any of the foregoing except as may be reasonably necessary to comply with applicable Law; and

(e) At the request of the League Office, the Team Operator shall immediately notify and make arrangements with its telephone service provider that the telephone number then currently assigned to the Team Operator reverts to and becomes the property of the League Office;

(f) The Team Operator may be required to assign to the League Office the Team Marks and Historic Marks, in accordance with Section 7.3.1.

12.5.2. If this Agreement is terminated by the League Office For Cause, in addition to the obligations set forth in Section 12.5.1, the League Office may also require the Team Operator to assign to the League Office (at no cost to the League Office) all of the Team Operator's right, title and interest in, to or under any of the Team Assets, as the League Office requests in its sole discretion, and the Team Operator hereby agrees to fully and completely cooperate with the League Office to give effect to this Section.

12.6. <u>Right of the League Office to Operate the Team</u>. If the Team Operator ceases to operate the Team or the League Office terminates this Agreement For Cause either (a) during a Season or (b) after the earlier of (i) November 1 and (ii) the date the schedule for the upcoming Season has been made public, the League Office may (but shall not be obligated to) take any action reasonably necessary to continue performing Team Activities for the applicable Season (but, in no event, beyond the applicable Season). In connection with the foregoing:

12.6.1. The Team Operator shall assign to the League Office at no cost such of the Team Operator's right, title and interest in and to the Team Assets as the League Office may request. Until the applicable assignments have been completed, the Team Operator hereby grants to the League Office a non-exclusive, royalty-free right and license to use, only in connection with Team Activities, the Team Marks, the Team Lists, and such other information and intellectual property as may be reasonably necessary to perform Team Activities.

12.6.2. As further set forth in Article XI (and subject to the provisions thereof), the Team Operator hereby agrees to indemnify and hold harmless League Office Entities against any and all Losses incurred by League Office Entities, including reasonable attorney's fees and costs (but not the League Office's overhead), arising out of or related to the League Office's operation of the Team pursuant to this Section 12.6.

12.6.3. The rights granted to the League Office in this Section 12.6 shall be without prejudice to and exclusive of any other rights or remedies available to the League Office pursuant to this

Agreement, at Law or in equity.

12.7. <u>Continued Use of Team Marks</u>. Unless this Agreement is terminated by the League Office For Cause and regardless of whether the League Office exercises the purchase right set forth in Section 12.8.1, the League Office may sell off any existing products bearing the Team Marks for a period of ninety (90) days following the conclusion of the final Match played by the Team, but the League Office's right to use the Team Marks and to grant or license such right to others shall otherwise immediately cease, except with regard to the publication of statistical or historical information (which right shall remain in perpetuity). The League Office to use the Team Marks cease any further use at the earliest practicable date, but in no instance longer than one (1) year following such termination or expiration.

12.8. Purchase Rights.

12.8.1. <u>Team Marks</u>. Upon termination of this Agreement by the League Office For Cause, the Team Operator shall, upon request from the League Office, assign the Team Marks, any Historic Marks, and their goodwill to the League Office without charge. Upon the expiration or termination of this Agreement other than For Cause, the League Office has the option, exercisable by written notice within sixty (60) days from such expiration or termination, to purchase the Team Marks and Historic Marks for an amount equal to the Net Licensing Revenue directly attributable to such Team Marks and Historic Marks during the last full Fiscal Year.

12.8.2. Equipment. Upon the expiration of this Agreement or if this Agreement is terminated by the League Office For Cause, the League Office shall have the option, to be exercised within thirty (30) days after termination, to purchase any or all of the furnishings, equipment, signs, fixtures, supplies, or inventory of the Team Operator or its Affiliates related to Team Activities, at the lesser of the Team Operator's cost or Fair Market Value. The cost for such items shall be determined based upon a five (5) year straight-line depreciation of original costs. For equipment that is five (5) or more years old, the parties agree that Fair Market Value shall be deemed to be ten percent (10%) of the equipment's original cost. If the League Office elects to exercise any option to purchase herein provided, it shall have the right to set off against the purchase price for such assets all amounts due from the Team Operator.

12.9. <u>Reinstatement and Extension</u>. If the provisions of this Agreement provide for periods of notice less than those required by applicable Law or provide for termination, cancellation or expiration of this Agreement other than in accordance with applicable Law, the League Office may reinstate or extend the Term for the purpose of complying with applicable Law by submitting a written notice to the Team Operator without waiving any of the League Office's rights under this Agreement.

12.10. <u>Discontinuing Services to the Team Operator</u>. Notwithstanding any other provision in this Agreement, for so long as the Team Operator is in default under any of the terms and conditions of this Agreement, League Office Entities shall not be obligated to provide any of the services required to be provided to the Team Operator hereunder, and the failure by any League Office Entities to provide such services shall not be deemed to be a breach or an event of default hereunder.

XIII. DISPUTE RESOLUTION

13.1. <u>Choice of Law</u>. This Agreement, the documents executed in connection herewith, the relationship of the parties hereunder and all disputes between the parties with respect to any of the foregoing shall be governed by and construed in accordance with the laws of the State of Florida without

giving effect to any conflict of law provision (whether of the State of Florida or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Florida.

13.2. <u>Consent to Jurisdiction</u>. THE TEAM OPERATOR AND THE LEAGUE OFFICE CONSENT AND IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION AND VENUE OF ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION LOCATED IN HILLSBOROUGH COUNTY, FLORIDA, AND WAIVE ANY OBJECTION TO THE JURISDICTION AND VENUE OF SUCH COURTS. ANY LAWSUIT ARISING OUT OF OR RELATING TO THIS AGREEMENT MUST BE BROUGHT IN A COURT OF COMPETENT JURISDICTION IN HILLSBOROUGH COUNTY, FLORIDA. HOWEVER, THE EXCLUSIVE CHOICE OF VENUE AND JURISDICTION DO NOT PRECLUDE THE BRINGING OF ANY ACTION BY THE PARTIES OR THE ENFORCEMENT BY THE PARTIES OF ANY JUDGMENT OBTAINED IN ANY SUCH JURISDICTION, IN ANY OTHER APPROPRIATE JURISDICTION OR THE RIGHT OF THE PARTIES TO CONFIRM OR ENFORCE ANY ARBITRATION AWARD IN ANY APPROPRIATE JURISDICTION.

13.3. <u>Limitation of Claims</u>. Any and all claims, actions, or Legal Proceedings arising out of or relating to this Agreement, the relationship of the Team Operator and the League Office, or the Team Operator's performance (or non-performance) of Team Activities, brought by either the League Office or the Team Operator against the other, shall be commenced within two (2) years from the occurrence of the facts giving rise to such claim or action, or such claim or action shall be barred.

13.4. <u>Limitation of Damages</u>. To the fullest extent permitted by law, no party shall be liable to any other party for any damages or other amounts which represent lost profits or indirect, incidental, special, consequential or punitive damages arising from the performance or nonperformance of this Agreement or any acts or omissions related thereto.

13.5. <u>Attorneys' Fees</u>. In the event that a League Office Entity institutes any Legal Proceeding against a Team Operator Entity, or vice versa, arising out of or relating to this Agreement, the prevailing party in the suit, action, or proceeding shall be entitled to receive, in addition to all other damages to which it may be entitled, the costs incurred by such party in conducting the suit, action, or proceeding, including reasonable attorneys' fees and expenses and court costs.

13.6. Arbitration.

13.6.1. The League Office and the Team Operator agree that, except for controversies, disputes, or claims related to or based on improper use of the League Marks or the Team Marks (or other disputes which may fairly be categorized as "trademark disputes"), all controversies, disputes or claims between the League Entities and their respective Personnel, on the one hand, and the Team Operator, its Equity Owners, Affiliates, or Personnel, on the other hand, arising out of or, in any way related to (a) this Agreement or any other agreement between the Team Operator and the League Office (including its formation, execution, interpretation, performance and any alleged breach), (b) the League Office's relationship with the Team Operator or (c) the validity of this Agreement or any other agreement between the Team Operator and the League Office shall be resolved through binding arbitration, on demand of either the League Office or the Team Operator, such arbitration to be administered by the AAA under its AAA Commercial Rules. All matters relating to arbitration will be governed by the United States Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). Despite the League Office's and the Team Operator's agreement to arbitrate, the League Office and the Team Operator each have the right, in a proper case, to seek specific performance, temporary restraining orders and temporary or preliminary injunctive relief from a court of competent jurisdiction; provided, however, that seeking such relief in a court of competent jurisdiction shall not relieve the League Office or the Team Operator from submitting the remaining

portions of the dispute to arbitration on the merits as provided in this Section.

13.6.2. The award rendered by the arbitrators shall be final and binding on the parties and may be entered and enforced in any court having jurisdiction, and any court where a party or its assets is located (to whose jurisdiction the parties consent for the purposes of enforcing the award). Judgment on the award shall be final and non-appealable. Subject to the limitation of liability set forth in Section 13.4, the arbitrator has the right to award or include in his or her award any relief which he or she deems proper, including, without limitation, money damages (with interest on unpaid amounts from the date due), specific performance, injunctive relief, and attorney fees and costs. The League Office and the Team Operator agree to be bound by the provisions of any limitation on the period of time in which claims must be brought under applicable law or this Agreement, whichever expires earlier. The League Office and the Team Operator further agree that, in any arbitration proceeding, each must submit or file any claim which would constitute a compulsory counterclaim (as defined by Rule 13 of the United States Federal Rules of Civil Procedure) within the same proceeding as the claim to which it relates. Any claim which is not submitted or filed as required is forever barred. The arbitrator may not consider any settlement discussions or offers that might have been made by either the Team Operator or the League Office.

13.6.3. There shall be three arbitrators, regardless of the amount in controversy, unless otherwise agreed by the parties.

13.6.4. All proceedings will be conducted in Tampa, Florida, unless otherwise agreed upon by the parties. Subject to the provisions of Section 11.2, judgment upon the arbitration award may be entered in any court of competent jurisdiction.

13.6.5. Except as may be required by law, neither a party nor the arbitrators may disclose the existence, content or results of any arbitration without the prior written consent of both parties, unless to protect or pursue a legal right.

13.6.6. The League Office and the Team Operator agree that arbitration will be conducted on an individual, not a class-wide, basis and that an arbitration proceeding between League Office Entities and their respective Equity Owners and Personnel, and the Team Operator (and/or the Team Operator's Equity Owners, Affiliates and/or Personnel) may not be consolidated with any other arbitration proceeding between the League Office and any other Person.

13.7. <u>Waiver of Jury Trial</u>. THE LEAGUE OFFICE AND THE TEAM OPERATOR EACH IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY DISPUTE ARISING OUT OF OR RELATED TO THIS AGREEMENT, THE DOCUMENTS EXECUTED IN CONNECTION HEREWITH, AND THE RELATIONSHIP OF THE PARTIES HEREUNDER.

XIV. MISCELLANEOUS

14.1. <u>Confidentiality</u>. From time to time during the Term of this Agreement, one party (as the "Disclosing Party") may disclose or make available to the other party (as the "Receiving Party") information about its business affairs, confidential intellectual property, trade secrets, third-party confidential information, and other sensitive or proprietary information (collectively, "Confidential Information"). The Receiving Party shall not, during the Term and, subject to the last sentence of this subsection, for a period of two (2) years following the expiration or earlier termination of this Agreement, communicate, divulge or use for the benefit of any other Person any Confidential Information, unless otherwise required to comply with applicable Law. The Receiving Party shall divulge

such Confidential Information only to such of its Personnel as must reasonably have access to it. Confidential Information shall not include information, knowledge or know-how that (i) the Receiving Party can demonstrate came to its attention prior to disclosure thereof by the Disclosing Party; (ii) becomes a part of the public domain other than as a result of a disclosure by the Receiving Party in violation of this Agreement; (iii) was independently developed by the Receiving Party without use of, or reference to, any Confidential Information; (iv) is generally considered to be common knowledge within the soccer industry or community; and/or (v) was obtained on a non-confidential basis from a third party who was not legally or contractually restricted from disclosing such information. To the extent any Confidential Information constitutes a trade secret under applicable law, the restrictions on the Receiving Party described herein shall remain in effect for so long as such Confidential Information remains a trade secret under applicable law. Notwithstanding the foregoing, the League Office may disclose Confidential Information (but not trade secrets) to USSF and other team operators in the League provided that, in each instance, the League Office reasonably determines that such disclosure is in the best interest of the League and the receiving party (a) is informed of the confidential nature of the Confidential Information, (b) is subject to confidentiality duties or obligations to the League Office that are no less restrictive than the terms and conditions of this Agreement, and (c) is provided only the minimum amount of Confidential Information necessary to achieve the permissible purpose of such disclosure.

14.2. <u>Non-Disparagement</u>. Each Party acknowledges and agrees that it shall not, during the Term and, subject to the last sentence of this subsection, for a period of two (2) years following the expiration or earlier termination of this Agreement, make any public statement (whether oral, written, or otherwise), or knowingly engage in any conduct, that is disparaging or derogatory or detrimental to the good name or business reputation of the other party, any of the other party's officers, directors, shareholders, or employees including, but not limited to, any statement that disparages the products, services, finances, financial condition, capabilities or other aspects of the business or operations of the other party. Notwithstanding anything to the contrary herein, neither party shall be in breach of this section for the making of any truthful statements under oath.

14.3. <u>League Work Stoppage</u>. In the event of a strike by the players in the League or a lockout of the players by the League (in either case a "<u>League Work Stoppage</u>") that precludes the satisfaction of either Party's obligations under this Agreement or applicable League Rules, all obligations of the Parties hereunder shall continue except that neither Party shall be in breach for failing to satisfy those obligations precluded by the League Work Stoppage.

14.4. <u>No Waiver</u>. Failure by the League Office or the Team Operator to enforce any rights under this Agreement shall not be construed as a waiver of such rights. Any waiver, including waiver of default, in any one instance shall not constitute a continuing waiver or a waiver in any other instance.

14.5. <u>No Joint Venture</u>. The League Office and the Team Operator are completely separate entities and are not fiduciaries, partners, or joint venturers and, except as set forth in this Agreement, neither Party is an agent of the other and neither Party shall have the power to bind the other. No act or assistance given by either Party or its Affiliates to the other Party or its Affiliates pursuant to this Agreement shall be construed to alter the relationship. The Team Operator shall be solely responsible for compliance with all applicable Law and for its policies, practices, and decisions relating to the operation of the Team. During the Term, the Team Operator shall hold itself out to the public as an independent party operating the Team pursuant to a franchise agreement with the League Office. Nothing in this Agreement authorizes the Team Operator to make any contract, agreement, warranty, or representation on behalf of the League Office, or to incur any debt or other obligation in the name of the League Office; and the League Office shall in no event assume liability for, or be deemed liable hereunder as a result of, any such action. 14.6. <u>Injunctive Relief</u>. Nothing herein contained shall bar the right of the League Office or the Team Operator to obtain specific performance or injunctive relief against threatened conduct that will cause loss or damages to such party, including, without limitation, violations of the terms of Section 4.22, under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary injunctions.

14.7. <u>Notice</u>. All notices, requests, demands, payments, consents and other communications hereunder shall be transmitted in writing by United States mail, postage prepaid, by hand delivery with written receipt, by overnight courier with verification of delivery addressed, if to the Team Operator at the contact information set forth in Schedule 2 and if to the League Office as follows:

USL SUPER LEAGUE, LLC 1715 N. Westshore Blvd. Suite 825 Tampa, Florida 33607 Attn: Legal

The League Office and the Team Operator may change the mailing address for notice purposes by giving written notice of such change of address to the other party. Mailed notices shall be deemed communicated five (5) days from the time of mailing. All other notices shall be deemed delivered as of actual delivery.

14.8. <u>Cost of Enforcement</u>. Each of the League Office and the Team Operator shall be entitled to recover from the other all fees, costs and expenses incurred by such party in enforcing the covenants, terms and conditions of this Agreement against the other party, including, without limitation, (a) the collection of any amounts required to be paid under this Agreement, (b) the enforcement of covenants and (c) the protection of the League Marks and Team Marks.

14.9. <u>Remedies Cumulative</u>. Except as otherwise set forth in Section 12.3 ("Additional Termination Rights of the League Office"), all rights and remedies provided in this Agreement are cumulative and not exclusive, and the exercise by either party of any right or remedy does not preclude the exercise of any other rights or remedies that may now or subsequently be available at Law, in any other agreement between the Parties or otherwise.

14.10. <u>Approvals</u>. Whenever this Agreement requires the prior authorization, approval or consent of a party hereto, the party seeking authorization, approval or consent shall make a timely written request to the other party therefor, and such authorization, approval or consent must be obtained in writing.

14.11. Interpretation of Rights. Whenever this Agreement provides that the League Office has a certain right, that right is absolute and, except as expressly limited or conditioned herein, the parties intend that the League Office's exercise of that right will not be subject to any limitation or review. The League Office has the right to operate, administrate, develop, and change the System in any manner that is not specifically precluded by the provisions of this Agreement, and all rights and licenses not specifically and expressly granted to and conferred on the Team Operator by this Agreement are, for all purposes, reserved to the League Office. Whenever the League Office reserves discretion in a particular area or where the League Office agrees to exercise its rights reasonably or in good faith, the League Office will satisfy its obligations whenever the League Office's decisions or actions will be deemed to be the result of Reasonable Business Judgment, even if other reasonable or arguably preferable alternatives are available, if the League Office's decisions or actions are intended, in whole or significant part, to promote or benefit the League generally, even if the decision or action also promotes the League Office's financial or other individual interest. Examples of decisions or actions designed to benefit or

promote the League generally, include, without limitation, decisions and actions intended to enhance the value of the League Marks, manage League Activities, improve uniformity, enhance or encourage modernization or improve the competitive position of the League.

14.12. <u>Entire Agreement; Amendment</u>. This Agreement, the exhibits hereto, and the documents referred to herein (including the Guaranty) constitute the entire agreement between the League Office and the Team Operator concerning the subject matter hereof, and supersede any prior agreements, no other representations having induced the Team Operator to execute this Agreement. No amendment, change, or variance from this Agreement shall be binding on either party unless mutually agreed to by the Parties and executed by their authorized officers or agents in writing.

14.13. <u>Severability</u>. The provisions of this Agreement are severable and any invalidity of any portion of this Agreement shall not affect the validity of the remaining portion.

14.14. Time Is of the Essence. Time is of the essence of this Agreement and each and every part hereof.

14.15. <u>Further Assurance</u>. The parties agree to execute such other documents and perform such additional acts as may be necessary or desirable to carry out the purposes of this Agreement.

14.16. <u>No Third-Party Beneficiaries</u>. Except as expressly provided to the contrary herein, nothing in this Agreement is intended, nor shall be deemed, to confer upon any Person other than the Team Operator, the League Office and their permitted successors and assigns, any rights or remedies under or by reason of this Agreement.

14.17. <u>Headings and Table of Contents</u>. The headings and table of contents used herein are for purposes of convenience only and shall not be used in construing the provisions hereof. As used herein, the male gender shall include the female and neuter genders, the singular shall include the plural, and the plural, the singular.

14.18. <u>Survival</u>. All representations, warranties, covenants and agreements of parties hereto that by their sense and context are intended to survive the expiration or termination of this Agreement, including, without limitation, each of Sections 4.22 (Restrictive Covenants and Upward Movement); 14.1 (Confidentiality); 14.2 (Non-Disparagement); 12.4 (Effect of Termination); 12.6 (Right of the League Office to Operate the Team); 12.7 (Continued Use of Marks); 12.8 (Purchase Rights); 12.9 (Reinstatement and Extension); and 12.10 (Discontinuing Services to Team Operator), as well as Articles XI (Indemnification) and XIII (Dispute Resolution) shall survive the expiration or termination of this Agreement regardless of the reason for termination.

14.19. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which when executed and delivered shall be deemed to be an original and all of which together shall constitute one and the same instrument. This Agreement may be executed and delivered by facsimile or other electronic imaging means (including .PDF), which signature and copies shall deemed to be an original.

14.20. <u>Disclosure Exemption</u>. The franchise sale is for more than \$1,233,000 - excluding the cost of unimproved land and any financing received from the franchisor or an affiliate - and thus is exempted from the Federal Trade Commission's Franchise Rule disclosure requirements, pursuant to 16 CFR 436.8(a)(5)(i).

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the Effective Date.

USL SUPER LEAGUE, LLC a Florida limited liability company

By:

Name: Title:

[Company] a [State of Org./Inc.] [Entity Type]

By:

Name: Title:

Definitions

"AAA" means the American Arbitration Association.

"AAA Commercial Rules" means the Commercial Arbitration Rules and Mediation Procedures of the AAA including, if appropriate, the AAA's Procedures for Large, Complex Commercial Disputes.

"Academy Team" has the meaning set forth in Section 4.24.1

"Affiliate" means, with respect to any Person, any Person directly or indirectly Controlling, Controlled by or under common Control with such Person.

"Agreement" has the meaning set forth in the introduction.

"Ancillary Commercial Inventory" means the sponsorship, promotional, and advertising inventory related to the Team and controlled by the Team Operator that is not considered Core Commercial Inventory.

"Anti-Terrorism Laws" mean Executive Order 13224 issued by the President of the United States, the USA PATRIOT Act, and all other present and future Laws addressing or in any way relating to terrorist acts and acts of war.

"Board of Governors" has the meaning set forth in Section 9.1.

"Books and Records" has the meaning set forth in Section 4.21.1.

"**Broadcast**" means the display or delivery of accounts of Matches, by way of radio, internet or television or any other medium now existing or hereafter created, whether live or recorded.

"**Broadcast Inventory**" means any advertising, sponsorship, or promotional inventory integrated into, overlaid on, or otherwise for sale as part of a Broadcast.

"**Broadcast Rights**" means all rights related to the Broadcasts throughout the Universe through any and all forms, means or modalities of electronic or other tangible or non-tangible Exhibition of audio, visual or audiovisual programming (whether now existing or developed in the future).

"Cable Television" means any system that is used to deliver multiple linear programming services in a closed and encrypted manner primarily for in the premises reception and viewing on a television set;

"Championship" means the championship game for the League held at the conclusion of the regular season and playoffs.

"**Change of Control**" means, with respect to the Team Operator, after the date hereof, the occurrence of any transaction or series of transactions which results in the change in identity of the Principal Owner.

"Change of Control Fee" means the fee set forth in Schedule 4.

"Collective Bargaining Activities" means activities related to the unionization (or potential unionization) or collective bargaining of players.

"Collective Use Rights" means the right to use and exploit the Team Marks, provided that they are equally emphasized with the team marks of all other teams in the League or an identifiable subset of teams therein (for example, a conference or division).

"Commercial Affiliations" mean advertising, sponsorship, and promotional agreements.

"Competition Marks" means the identifiers of League-related competitions (e.g., the playoffs), League events (e.g., all-star game), as well as any specific Matches (e.g., a rivalry match) regardless of whether the opponent is also a participant in the League, in each instance including but not limited to the trade names, logos, symbols, and mascots (whether registered or not), if any, of such events, competitions, and Matches.

"Composite Mark" means a League Mark that incorporates the name and/or logo of another brand (e.g., the Barclay's Premier League).

"Consumer Price Index" means the Consumer Price Index, All Urban Consumers, United States, All Items (1982 - 1984 = 100), as published by the Bureau of Labor Statistics of the United States Department of Labor or, if such index is not available, such other index as the parties may agree most closely resembles such index.

"Control" (including the terms "Controlled by" and "under common Control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, as trustee or executor, by contract or otherwise, including, without limitation, the ownership, directly or indirectly, of securities having the power to elect a majority of the board of directors or similar body governing the affairs of such Person.

"Core Commercial Inventory" means sponsorship and advertising inventory relating to the Team including without limitation: (i) advertising, product or service placement, sponsorships and promotions relating to commercial identification in the Field of Play; (ii) promotional activities in and around the Home Stadium; (iii) in-stadium print collateral; and (iv) advertising and promotions on the Team Site and Team Digital Platforms, as such inventory may be designated in League Rules.

"**Digital Distribution**" means the Exhibition via (i) the Internet (including Internet applications and services) or (ii) wireless technology used to Exhibit audio, visual or audiovisual programming via mobile devices such as smartphones, tablets and PDAs, including Wireless Application Protocol and Internet Protocol Television and other devices whether now existing or developed in the future, but excluding radio distribution.

"Disclosing Party" has the meaning set forth in Section 14.1.

"**Domain Name**" means the web address of the Team Site, as included on Schedule 5 or at such alternative location as may be approved by the League Office.

"Effective Date" has the meaning set forth in the introduction.

"**Equity Interest**" means any direct or indirect legal or beneficial interest in the Team Operator (including without limitation the right or obligation to receive all or a portion of the profits or incur all or a portion of the losses of the Team Operator).

"Equity Owner" means the direct or indirect owner of an Equity Interest.

"Exhibit" means to transmit, broadcast, distribute and/or exhibit.

"Fair Market Value" means the cash price that an unaffiliated third party would pay to acquire such property (real, tangible, intangible, or otherwise) in an arm's-length transaction, assuming with respect to the Fair Market Value of such property, that it was being sold in a manner reasonably designed to solicit all possible participants and permit all interested Persons an opportunity to participate and to achieve the best value reasonably available to the seller at the time.

"Family Member" means, as to a natural person, his or her spouse, parent, sibling, descendant (including through adoption), or the spouse of each such natural person.

"Featured Game" means a Match designated and featured by the League Office (*e.g.*, Games of the Week, rivalry games, kickoff matches, or other nationally branded games).

"**Field of Play**" means the sidelines, the playing field, and those areas otherwise subject to exposure during a Broadcast, including without limitation: any retaining walls or other backdrops to the game presentation as well as any field-level signage, benches and other forms of Team seating or covering.

"FIFA" means Federation Internationale de Football Association, the governing body for professional soccer internationally, or its successors.

"Final Season" means the Season set forth in Schedule 3.

"Fiscal Year" means a calendar year ending December 31, unless another ending date is required by the League Office.

"For Cause" means for those causes set forth in Sections 12.1.

"GAAP" means U.S. generally accepted accounting principles.

"Governing Body" means an organization that governs, sanctions and/or oversees the League, which may include, without limitation, FIFA and USSF.

"Governmental Authority" means the government of any nation or any political subdivision thereof, whether at the national, state, territorial, provincial, municipal or any other level, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of, or pertaining to, government.

"Guaranty" has the meaning set forth in Section 8.3.

"Guarantor" means a Person having executed the Guaranty.

"Home Match" or "Home League Match" means any Match or League Match, as applicable, involving the Team occurring within the Protected Territory or in which the Team is otherwise designated as the "home" team.

"**Home Stadium**" means the stadium identified on Schedule 3 as the Home Stadium or, if applicable, the stadium identified as the Interim Stadium on Schedule 3, but only upon the terms and conditions set forth therein.

"Home Tickets" mean admission tickets to Home Matches.

"Initial Season" means the Season set forth in Schedule 3.

"Law" means, as to any Person, any law (including common law), statute, ordinance, treaty, rule, regulation, order, decree, judgment, writ, injunction, settlement agreement, requirement or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"League" means the league identified on Schedule 3, operated by the League Office.

"League Activities" shall mean, with respect to the League, those activities that are integral to operating a professional soccer league, including without limitation, the marketing and promotion of such League, individual teams in such League, maintenance of such League and Team historical and statistical information, operation and display of the game competitions in such League, and commercial exploitation of such League and individual teams therein which includes, without limitation, securing League Commercial Affiliations, Broadcast agreements (television, radio, internet, mobile technology, etc.), Licensing Arrangements, and the creation of publications subject to the terms of this Agreement.

"League Allocation" means, with respect to Core Commercial Inventory and Broadcast Inventory, the allocation to the League Office set forth in Section 6.3.3.

"League Assets" has the meaning set forth in Section 6.1

"League Commercial Affiliations" means Commercial Affiliations relating to the League, including to either (i) every team in the League or (ii) identifiable subsets of the teams in the League (including, for example and without limitation, all of the teams of a particular conference, division, or geographic region).

"League Events" means events relating to the League competition (e.g., the championship or all-star game).

"League IP" means information and intellectual property obtained or developed in connection with the League or the playing of Matches.

"League Marks" mean the identifiers of the League, including but not limited to its trade names, logos, symbols, and mascots (whether registered or not), if any.

"League Match" means any game played (or scheduled to be played) by the Team as part of the regular season or playoffs of the League (including the League Championship), or any game in which the Team is eligible to play due, in whole or in part, to its participation in the League (specifically including any U.S. Open Cup or CONCACAF Champions League matches)

"League Office" has the meaning set forth in the introduction.

"League Office Entities" means the League Office and its Affiliates.

"League Rules" means all standards (including the Minimum Standards), specifications, requirements, criteria, and policies of general application that have been and are in the future developed and compiled by the League Office for use by the Team Operator in connection with the operation of the Team, Team Activities, or League Activities and all resolutions of the Board of Governors, in each instance whether contained in the Manuals or set out in this Agreement or other written communication.

"Legal Proceeding" means any suit, action, or proceeding, including arbitration.

"Letter of Credit" means (i) an irrevocable, standby letter of credit issued by a bank with offices in the United States and acceptable to the League Office; (ii) cash to be held by the League Office; or (iii) such other instrument as is acceptable to the League Office.

"Licensed Products" means the products for manufacture, sale, or distribution bearing the Team Marks.

"Licensee" means a Person granted or licensed rights by the League Office to use the Team Marks.

"Licensing Arrangement" means agreements or arrangements for the manufacture, sale or distribution of Licensed Products.

"Local Advertising" means advertising and promotion related to the Team, whether through print, media, or other.

"Local Broadcast" means the Exhibition of Matches via Standard Television or Standard Radio solely within the Local Broadcast Territory.

"Local Broadcast Rights" mean the rights relating to the Local Broadcasts.

"Local Broadcast Territory" means the greater of (a) the designated market area in which the Home Stadium is located and (b) the Protected Territory.

"Local Revenues" means the gross receipts accruing from Team Activities.

"Losses" means, collectively, any losses, damages, liabilities, deficiencies, claims, actions, judgments, settlements, interest, awards, penalties, fines, costs, or expenses of whatever kind, including reasonable attorneys' fees.

"**Majority Vote**" means (a) more than half of the team votes cast voting in the affirmative <u>and</u> (b) the affirmative vote of the League Office.

"Management Arrangement" has the meaning set forth in Section 4.6.

"Match" means any game played (or scheduled to be played) by the Team (including any pre-season or exhibition matches).

"Maximum Participation Fee Increase" has the meaning set forth in Section 5.2.

"Minimum Performance Standard" has the meaning set forth in Section 4.7.

"Negotiation Rights" mean rights of first refusal, rights of first negotiation, and similar such rights.

"Net Licensing Revenues" shall include all income received by the League Office (prior to the deduction of the League Office's commission) from licensees pursuant to any Licensing Arrangements including, but not limited to, advances, royalties, guarantees, fees and payments (whether in cash, barter or other form of consideration) less any payments made or expenses incurred by the League Office for or on behalf of the Team Operator with the prior approval of the Team Operator or the Board of Governors.

"New Agreement" means a franchise agreement in substantially the form then used by the League Office for team operators in the League, which shall supersede this Agreement in all respects and the terms of

which may differ from the terms of this Agreement, including requirements to pay additional and/or higher fees, except that the New Agreement shall not include (a) any requirement to pay an additional expansion fee or (b) any change that (i) materially and adversely affects the Team Operator's basic rights to operate the Team in the League in the Protected Territory; or (ii) materially increases the Team Operator's net obligations; or (iii) materially reduces the League Office's obligations.

"New Market" has the meaning set forth in Section 4.2.3.

"Other Agreements" has the meaning set forth in Section 10.3.

"Participation Fee" has the meaning set forth in Section 5.2.

"Party" and "Parties" have the meaning set forth in the introduction.

"**Person**" means an individual or a corporation, partnership, sole proprietorship, company, firm, limited liability company, joint venture, trust, business association, organization, joint stock company, unincorporated organization, group acting in concert, Governmental Authority or other entity.

"**Personnel**" of a Party means any officers, directors, agents, employees or other Persons engaged or appointed by the Party which, in the case of the Team Operator, includes players on the Team.

"**Platform**" means any technology used for delivery of information (including, but not limited to, video images) to end users or consumers, including as of the date hereof, without limitation, such technologies as Cable Television, direct broadcast satellite, open video systems, computer on-line services, the Internet, technologies designed to reach mobile phones and other mobile devices, multichannel multipoint video distribution systems, closed circuit television, and satellite master antenna television systems, and, during the term hereof, any successor, supplementary, replacement, or new technology performing substantially similar functions as (or enhanced or additional functions with respect to) Platforms currently in existence;

"**Player Likenesses**" shall mean, collectively, the names, nicknames, numbers, pictures, or images (whether still, motion, video, digital, or television), voices, signatures, facsimile signatures, caricatures, reputation, goodwill, persona, any aspect of the right of privacy, personality or publicity or other likenesses of players on the Team.

"Pool Payments" has the meaning set forth in Section 5.4.1(r).

"**Principal Owner**" means a natural person who: (i) Controls the Team Operator; (ii) is the beneficial owner of at least thirty-five percent (35%) of the Equity Interests (or such other amount as may be required by the Pro League Standards); (iii) has authority to bind the Team Operator; and (iv) meets the thresholds, including net worth, proscribed by the Pro League Standards from time to time.

"Product Categories" means categories of products, services, or retail operations.

"**Professional League Standards**" means the rules and standards instituted or enforced by USSF and applicable to the League and its teams, as such rules and standards may be amended from time to time in USSF's sole discretion.

"Protected Territory" has the meaning set forth in Section 3.4.

"Provider" means a provider of products or services to or for the Team.

"Reasonable Business Judgment" has the meaning set forth in Section 14.11.

"Receiving Party" has the meaning set forth in Section 14.1

"**Relocation Proposal**" means any inquiry, proposal, solicitation or offer from or to any Person that contemplates a relocation of the Team from the Home Stadium.

"Renewal Date" is December 31 of the third to last year of the Term.

"Renewal Term" has the meaning set forth in Section 2.2.

"**Rival League**" means any professional soccer league (or soccer league that permits its teams to roster professional soccer players) with teams in the United States.

"Season" means the time period from the first official League Match of a competition through the conclusion of the championship game.

"Settlement Procedures" means the League Rules comprising the account settlement procedures, as the same may be amended from time to time.

"Stadium Lease" means any lease, license, use or occupancy agreement for the Home Stadium.

"Standard Radio" means free (non-subscription) over-the-air audio-only signals intended for reception by the general public.

"**Standard Television**" means free (non-pay-per-view and non-subscription) over-the-air broadcast television signals intended for reception by the general public transmitted by full power terrestrial broadcast television stations on spectrum assigned by the FCC. For avoidance of doubt, "Standard Television" shall not include, without limitation, (i) any on-demand (VOD, etc), pay-per-view, subscription, time-shifted transmission or other service in which any portion of a Standard Television program or signal is provided to viewers for payment or in a manner that is not simultaneous with its free, over-the-air linear terrestrial broadcast; (ii) any wireless distribution to mobile phones or other mobile devices; (iii) any interactive or internet/broadband distribution or distribution through any other systems network; (iv) any other non-television protocol or Platform; or (v) distribution via any linear Cable Television network or programming service.

"**Supermajority Vote**" means (a) more than two thirds (2/3) of the team votes cast voting in the affirmative <u>and</u> (b) the affirmative vote of the League Office.

"Taxes" means any and all sales, use, gross receipts, environmental, ad valorem, or excise tax or any other similar taxes, fees, duties, or charges of any kind imposed by any Governmental Authority; <u>exclusive, however</u>, of any taxes, assessments, or other levies imposed on income or capital (including leased or purchased property, equipment, or software), any franchise taxes, any taxes in lieu of net income taxes and any other direct taxes.

"Team" means the soccer team to be operated by the Team Operator pursuant to this Agreement.

"Team Assets" has the meaning set forth in Section 6.2.

"Team Activities" means those activities integral to the administration, marketing, promotion and operation of the Team and the commercial exploitation thereof, including the sale and fulfillment of Team

Commercial Affiliations, Local Broadcast Rights and Home Tickets.

"Team Commercial Affiliations" means Commercial Affiliations related to the Team.

"**Team Digital Platforms**" means any and all digital platforms including, but not limited to official websites, social media platforms (e.g., Twitter, Instagram, Facebook) as well as pages or microsites on platforms such as YouTube.

"Team Lists" means all customer lists and records relating to the Team.

"**Team Marks**" means the identifiers of the Team, including but not limited to team name, logos, characters, mascots, trademarks, trade names and other identifying words or symbols of the Team.

"Team Operator" means the Person identified in Schedule 2.

"Team Operator Entities" means the Team Operator and any entities Controlled by the Team Operator.

"Team Operator Training" has the meaning set forth in Section 3.5.

"Team Site" means the website dedicated to the Team located at the Domain Name.

"**Team Store**" means a Team-identified store or any store owned by the Team Operator or under common Control with the Team Operator that sells predominately Team-branded merchandise. For the avoidance of confusion, "Team Store" does <u>not</u> include any e-commerce store.

"Term" has the meaning set forth in Section 2.1.

"**Ticket Agent**" means a Person that processes the sale of a ticket on either the primary market (e.g., Ticketmaster) or secondary, resale market (e.g., StubHub).

"**Transfer**" means any sale, issuance, transfer, assignment, pledge, or other encumbrance of an interest or the rights thereof. The terms "Transferee," "Transferred," "Transferor" and other forms of the word "Transfer" shall have correlative meanings.

"USSF" means the United States Soccer Federation (also known as U.S. Soccer), the governing body for professional soccer in the United States, or such entity or organization as may assume responsibilities similar to that of the USSF in the future.

Team Operator and Principal Owner

	Team Operator:	[Company]
--	----------------	-----------

Address:

[Address] [City], [State], [Zip Code]

Principal Owner:

Address:

Basic Terms

League:	USL Super League
Effective Date:	
Initial Season:	2023
Final Season:	[10 th Season]
Interim Stadium (if applicable):	
Home Stadium:	
Protected Territory:	
Renewal Term:	10 years, then in accordance with the then-current form of franchise agreement.

Payment Terms

Expansion Fee:	\$10,000,000	
Initial Season – Participation Fee Amount:	The Participation Fee chargeable in the Initial Season shall be consistent with the participation fee charged to other team operators (which, for the 2024-2025 Season, is \$200,000, with half (\$100,000) payable upon execution and the other half (\$100,000) payable in advance of the Initial Season, at such time as the League Office may reasonably determine. Thereafter, the Participation Fee may increase year-over-year in accordance with the terms and conditions of this Agreement.	
Ramp-Up Fee:	The initial Ramp-Up Fee is \$100,000, payable on or before August 15, 2023. For any subsequent season (prior to its commencement) in which the Team does not participate, the Ramp-Up Fee shall be equal to 50% of the Participation Fee that otherwise would have been charged for that Season.	
Change of Control Fee:	10% of the then-current, published expansion fee	
Relocation Fee:	10% of the then-current, published expansion fee	
Renewal Fee:	\$10,000, as adjusted annually (using 2021 as the base year) by the percentage rate of increase plus three percent for the immediately preceding 12-month period in the Consumer Price Index.	

Team Marks and Indicia

Domain Name:

Team Name:

Primary Logo:

Equity Owners

As of the Effective Date, the full name and mailing address of each Equity Owner is included below, along with a description of the nature of their Equity Interest. For the avoidance of confusion, the direct and indirect owners of any entity that is an Equity Owner must also be included.

Equity Owner Name	If Equity Owner is an Entity, the direct and indirect owners of the entity	Description and number of Equity Interests

The Equity Owners of greater than five percent (5%) of the Equity Interests, as identified above, hereby acknowledge that they have reviewed, or have had ample opportunity to review, the Franchise Agreement and agree to be bound by the restrictive covenants contained therein, including, without limitation, those contained in Section 4.22 ("Restrictive Covenants").

[Name]

[Name]

[Name]

[Name]

[Name]

[Name]

Item	Amount	Notes
Teamworks (reporting)	\$1,335	Pass-Through
Insurance (General Liability)	\$6,468	Pass-Through
USSF/CSA Affiliation Fees	\$35,000 ²	Pass-Through
USSF Player Registration Fee	\$2,000	Pass-Through
Referees	\$70,000	Pass-Through (estimated)
Playoff Travel Pool	\$7,500	Pool Payment
Regular Season Travel	Variable	Pass-Through (payments estimated based on team travel budgets, with end of year reconciliations)
Team Site design (if applicable)	\$5,995	Pass-Through
Team Site maintenance and hosting	\$1,575	Pass-Through
VISTA (site survey, as applicable)	\$500	Pass-Through
VISTA (shipping, as applicable)	at cost	Pass-Through
VISTA (transmission)	\$25,000	Pass-Through (estimated)
FanThreeSixty	\$5,000	Pass-Through
WSC	\$7,000	Pass-Through
Total	\$167,373 (plus regular season travel)	

Anticipated Fees and Pass-Throughs (Excluding Participation Fee)

 $^{^2}$ The total USSF league fee is currently \$350,000 for up to 10 teams, which is then split amongst all team operators. The fee of \$35,000 assumes there are 10 teams in the League. USSF charges a fee of \$30,000 for each team over 10.

Antici	pated	Exclusive	or	Restricted	Providers

Company	Туре	Goods / Services	Requirements on Team Operator
Tickets.com [™]	Required Provider (unless the Team Operator elects to use SeatGeek, per below)	Primary Ticketing	Team Operator must use the Tickets.com platform made available by and through the League Office as its exclusive ticket sales platform for all Home Matches.
SeatGeek™	Required Provider (unless Club elects to use Tickets.com, per above)	Ticketing	Club SeatGeek Agreement provides that SeatGeek will be the exclusive ticket seller of tickets for (a) every Home Match at the Home Stadium and (b) every other event at the Home Stadium for which the Team Operator or its Affiliate controls the ticketing.
Sports Engine	Exclusive Provider	Team Website Hosting and Maintenance	Team Operator must use Sports Engine to host and maintain its Team Site.
Select TM	Exclusive Provider	Soccer Balls	Team Operator must use Select Balls in all Matches
UniSport™	Exclusive Provider	League Patches; Official names and numbers	Team Operator must use UniSport for League patches and the names and numbers appearing on the Team's jerseys and jersey replicas.
VISTA Worldlink™	Exclusive Provider	Broadcast Production and Transmission	Team Operator must use VISTA for the broadcast production and transmission of its Home Matches.
FanThreeSixty	Required Provider	Mobile Application and Data Warehouse	Team Operator must use FanThreeSixty as the provider of any mobile application and/or data warehouse used by the Team.
WSC	Required Provider	Clip Creation and AI Analysis of Video Feed	Team Operator must use WSC as the provider of any AI analysis/clip creation of the video feed of its matches.

Stadium Development Schedule

Reserved and Restricted Product Categories

1. Restricted Categories.

Commercial Affiliations or Local Advertising in connection with the following product/service categories may be permitted <u>only after prior written approval</u> of USL. In each instance, a complete copy of the agreement to be approved (including all attachments) should be submitted in final draft form to <u>legal@uslsoccer.com</u>. Following written approval of the agreement, a fully executed copy of the agreement must also be provided within 10 days of execution.

- (1) Media/streaming companies;
- (2) Non-fungible tokens (or "NFTs") or other cryptographic assets on a blockchain;
- (3) Controversial topics (e.g., section 501(c)(4) organizations, social cause/issue advocacy);
- (4) Movies, video games and other media that contain or promote objectionable material or subject matter (e.g., overtly sexual or excessively violent material), as determined by the League;
- (5) Gambling or companies associated with gambling (including daily fantasy sports) will be handled on a case-by-case basis;
- (6) CBD and hemp oil (or companies associated with CBD or hemp oil) will be handled on a caseby-case basis; and
- (7) Any Local Advertising that may reasonably be perceived as reflecting negatively upon the League or any team in the League.

Continued on the following page

2. League-Reserved Categories.

The following product/service categories have been reserved for exclusive sale by the League for the time periods associated therewith. Clubs may not license, sell, or otherwise permit the use of their Marks in connection with any of the following products/service categories. In the event that the League is unable or elects not to enter into League Commercial Affiliation in the categories below, it will release such category back to the Clubs at the earliest date practicable.

Category	Time Period
Soccer balls	Perpetuity
Soccer video games	Perpetuity
Jersey suppliers	Reserved ³

³ For reference, any decision to adopt a league-wide supplier would come only after substantial input from the Clubs and would ultimately require the Board of Governors to approve any such arrangement.

SCHEDULE 10

ADDENDUM TO THE FRANCHISE AGREEMENT FOR THE STATE OF ILLINOIS

In recognition of the requirements of the Illinois Franchise Disclosure Act of 1987, 815 Illinois Compiled Statutes §§ 705/1 - 705/44, the parties to the attached Franchise Agreement (the "Agreement") agree as follows:

1. Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement.

2. Article II of the Agreement, under the heading "Term; Renewal; Expiration", shall be amended by the addition of the following new subsection 2.4, which shall be considered an integral part of the Agreement:

"2.4. If any of the provisions of this Article II are inconsistent with Section 705/20 of the Illinois Franchise Disclosure Act of 1987, the provisions of the Illinois Franchise Disclosure Act of 1987 shall apply. If the League Office refuses to renew this Agreement, the League Office shall compensate the Team Operator if (and to the extent) such compensation is required under Section 705/20 of the Illinois Franchise Disclosure Act of 1987."

3. Section 12.1 of the Agreement, under the heading "Termination by the League Office," shall be amended by the addition of the following new subsection 12.1.4, which shall be considered an integral part of the Agreement:

"12.1.4 <u>Compliance with the Illinois Franchise Disclosure Act of 1987</u>. If any of the provisions of this Section 11.2.5. are inconsistent with Section 705/19 of the Illinois Franchise Disclosure Act of 1987, the provisions of the Illinois Franchise Disclosure Act of 1987 shall apply."

4. Section 13.1 of the Agreement, under the heading "Choice of Law," is deleted in its entirety, and shall have no further force or effect, and the following shall be substituted in lieu thereof:

"13.1 This Agreement, the documents executed in connection herewith, the relationship of the parties hereunder and all disputes between the parties with respect to any of the foregoing shall be governed by and construed in accordance with the laws of the State of Illinois without giving effect to any conflict of law provision (whether of the State of Illinois or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Illinois."

5. Section 13.2 of the Agreement, under the heading "Consent to Jurisdiction," is deleted in its entirety and replaced with the following:

"13.2 THE TEAM OPERATOR AND THE LEAGUE OFFICE CONSENT AND IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION AND VENUE OF ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION LOCATED IN HILLSBOROUGH COUNTY, FLORIDA, AND WAIVE ANY OBJECTION TO THE JURISDICTION AND VENUE OF SUCH COURTS. ANY LAWSUIT ARISING OUT OF OR RELATING TO THIS AGREEMENT, BUT EXPRESSLY EXCLUDING CLAIMS ARISING UNDER THE ILLINOIS FRANCHISE DISCLOSURE ACT OF 1987, MUST BE BROUGHT IN A COURT OF COMPETENT JURISDICTION IN HILLSBOROUGH COUNTY, FLORIDA. HOWEVER, THE EXCLUSIVE CHOICE OF VENUE AND JURISDICTION DO NOT PRECLUDE THE BRINGING OF ANY ACTION BY THE PARTIES OR THE ENFORCEMENT BY THE PARTIES OF ANY JUDGMENT OBTAINED IN ANY SUCH JURISDICTION, IN ANY OTHER APPROPRIATE JURISDICTION OR THE RIGHT OF THE PARTIES TO CONFIRM OR ENFORCE ANY ARBITRATION AWARD IN ANY APPROPRIATE JURISDICTION.

6. Article XIII of the Agreement, under the heading "Dispute Resolution," shall be amended by the addition of the following new Section 13.8, which shall be considered an integral part of the Agreement:

"(c) Nothing contained in this Article XIII shall constitute a condition, stipulation, or provision purporting to bind any person to waive compliance with any provision of the Illinois Franchise Disclosure Act of 1987 or any other Illinois law (as long as the jurisdictional requirements of the Illinois Franchise Disclosure Act of 1987 are met)."

7. Each provision of this Addendum to the Franchise Agreement for the State of Illinois shall be effective only to the extent, with respect to such provision, the jurisdictional requirements of the Illinois Franchise Disclosure Act of 1987 are met independently, without reference to this Addendum to the Franchise Agreement for the State of Illinois, and only to the extent such provision is a then valid requirement of the statute.

IN WITNESS WHEREOF, the parties hereto have caused this Addendum to the Franchise Agreement for the State of Illinois to be duly executed as of the Effective Date of the Agreement.

LEAGUE OFFICE: USL Super League, LLC By United Soccer Leagues, LLC, its sole member

By:___

Signature

Name

Title

TEAM OPERATOR:

(Name of Entity)

By:____

Signature

Name

Title

SCHEDULE 10

ADDENDUM TO THE FRANCHISE AGREEMENT FOR THE STATE OF MINNESOTA

In recognition of the requirements of the Minnesota Franchise Act, Minnesota Statutes §§ 80C.01 - 80C.30, and of the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce, Minnesota Rules §§ 2860.0100 - 2860.9930, the parties to the attached Franchise Agreement (the "Agreement") agree as follows:

1. Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement.

2. Section 2.2.3 of the Agreement, under the heading "Renewal Term," is deleted in its entirety, and shall have no further force or effect, and the following paragraph shall be substituted in lieu thereof:

"2.2.3. On or before the Renewal Date, the Team Operator executes a general release, in a form prescribed by the League Office, of any and all claims against the League Office Entities and their respective Personnel, excluding only such claims as Franchisee may have under the Minnesota Franchise Act and the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce."

3. Section 2.3 of the Agreement, under the heading "Expiration", shall be amended by adding the following sentence, which shall be considered an integral part of the Agreement:

"Minnesota law provides franchisees with certain non-renewal rights. In sum, Section 80C.14, Subsection 4, of the Minnesota Franchise Act currently requires, except in certain specified cases, that Franchisee be given 180 days notice of non-renewal of this Agreement."

4. Section 8.2 of the Agreement, under the heading "License and Limitation on Use of USL Marks," shall be amended by the addition of the following new subsection 8.2.6., which shall be considered an integral part of the Agreement:

"8.2.6. Pursuant to Section 800.12, Subsection 1(g), of the Minnesota Statutes, Franchisor is required to protect any rights Franchisee may have to the USL Marks."

5. Section 10.1 of the Agreement, under the heading "No Transfer without Franchisor Approval," shall be amended by the addition of the following new paragraph, which shall be considered an integral part of the Agreement:

"Minnesota law provides franchisees with certain transfer rights. In sum, Section 80C.14, Subsection 5, of the Minnesota Franchise Act currently requires that consent to the transfer of the franchise may not be unreasonably withheld."

6. Section 10.2.3 of the Agreement, under the heading "Conditions to Approval," is deleted in its entirety, and shall have no further force or effect, and the following paragraph shall be substituted in lieu thereof:

"10.2.3. Franchisee and/or its Owners, as applicable, shall have executed a general release under seal, in a form satisfactory to Franchisor, of any and all claims against Franchisor and its officers, directors, owners and employees, in their corporate and individual

Addendum – Minnesota

capacities, excluding only such claims as Franchisee or its Owners may have under the Minnesota Franchise Act and the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce."

7. Section 11.2 of the Agreement, under the heading "Termination by Franchisor," shall be amended by the addition of the following new subsection 11.2.5, which shall be considered an integral part of the Agreement:

"11.2.5.<u>Compliance with the Minnesota Franchise Act</u>. Minnesota law provides franchisees with certain termination rights. In sum, Section 80C.14, Subsection 3, of the Minnesota Franchise Act currently requires, except in certain specified cases, that Franchisee be given 90 days notice of termination (with 60 days to cure) of this Agreement."

8. Section 2860.4400J of the Franchise Rules and Regulations promulgated by the Minnesota Commissioner of Commerce prohibits a franchisor from requiring a franchisee to consent to liquidated damages. Section 12.10 of the Agreement, under the heading "Liquidated Damages," shall be deleted in its entirety.

9. Section 15.2 of the Agreement is deleted in its entirety, and shall have no further force or effect, and the following Section shall be substituted in lieu thereof:

"15.2. <u>Injunctive Relief</u>. Nothing herein contained shall bar the right of Franchisor to seek injunctive relief against threatened conduct that will cause it loss or damages, including, without limitation, violations of the terms of <u>Sections 7.4, 7.6 or 8.2</u>, under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary injunctions."

10. Article XV shall be amended by the addition of the following new Section 15.15, which shall be considered an integral part of the Agreement:

"15.15. <u>Effect of Minnesota Law</u>. Notwithstanding anything to the contrary set forth herein, pursuant to Minnesota Rules § 2860.4400J, nothing herein shall abrogate or reduce, or require Franchisee to waive, any of Franchisee's rights as provided for in the Minnesota Statutes, Chapter 80C, or its rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction."

11. Section 16.2 of the Agreement, under the heading "Consent to Jurisdiction," shall be amended by the addition of the following new subsection (c), which shall be considered an integral part of the Agreement:

"(c) Section 80C.21 of the Minnesota Franchise Act and Section 2860.4400J of the Franchise Rules and Regulations promulgated by the Minnesota Commissioner of Commerce prohibit Franchisor from requiring litigation to be conducted outside of the State of Minnesota. In addition, nothing in the Franchise Disclosure Document or this Agreement can abrogate or reduce any of Franchisor's rights as provided for in the Minnesota Franchise Act, or Franchisor's rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction."

12. Section 16.3 of the Agreement is deleted in its entirety, and shall have no further force or effect, and the following Section shall be substituted in lieu thereof:

"16.3. <u>Limitation of Claims</u>. Any and all claims and actions arising out of or relating to this Agreement, the relationship of Franchisee and Franchisor, or Franchisee's operation of the Franchised Team, brought by either party hereto against the other, whether in mediation, or a legal action, shall be

commenced within three (3) years after the cause of action accrues, or such claim or action shall be barred."

13. Each provision of this Addendum to the Franchise Agreement for the State of Minnesota shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Minnesota Franchise Act or the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce are met independently, without reference to this Addendum to the Franchise Agreement for the State of Minnesota, and only to the extent such provision is a then valid requirement of the statute.

IN WITNESS WHEREOF, the parties hereto have caused this Addendum to the Franchise Agreement for the State of Minnesota to be duly executed as of the Effective Date of the Agreement.

LEAGUE OFFICE: USL Super League, LLC By United Soccer Leagues, LLC, its sole member

By:___

Signature

Name

Title

TEAM OPERATOR:

(Name of Entity)

By:___

Signature

Name

Title

ADDENDUM TO THE FRANCHISE AGREEMENT FOR THE STATE OF NEW YORK

In recognition of the requirements of the Article 33 of the General Business Law of the State of New York, §§ 680 - 695, the parties to the attached Franchise Agreement (the "Agreement") agree as follows:

1. Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement.

2. Section 2.2.3 of the Agreement, under the heading "Renewal Term" is deleted in its entirety, and shall have no further force or effect, and the following paragraph shall be substituted in lieu thereof:

"2.2.3. On or before the Renewal Date, the Team Operator executes a general release, in a form prescribed by the League Office, of any and all claims against the League Office Entities and their respective Personnel; <u>provided</u>, <u>however</u>, that all rights enjoyed by the Team Operator and any causes of action arising in its favor from the provisions of Article 33 of the General Business Law of the State of New York, and the rules and regulations issued thereunder, shall remain in force, it being the intent of this provision that the non-waiver provisions of Sections 687.4 and 687.5 of Article 33 of the General Business Law of the State of New York be satisfied."

3. Section 10.1.3(d) of the Agreement, under the heading "Change of Control Transfers," is deleted in its entirety, and shall have no further force or effect, and the following paragraph shall be substituted in lieu thereof:

"(d) the Team Operator and its transferring Equity Owners sign the then-current form of termination agreement and a general release, in a form satisfactory to the League Office, of any and all claims against any League Office Entities and their respective Personnel; <u>provided</u>, <u>however</u>, that all rights enjoyed by the Team Operator and its Equity Owners and any causes of action arising in their favor from the provisions of Article 33 of the General Business Law of the State of New York, and the rules and regulations issued thereunder, shall remain in force, it being the intent of this provision that the non-waiver provisions of Sections 687.4 and 687.5 of Article 33 of the General Business Law of the State of New York be satisfied."

4. Section 10.3 of the Agreement, under the heading "League Office Transfers," shall be amended by the addition of the following new sentence, which shall be considered an integral part of the Agreement:

"However, no assignment shall be made except to an assignee who, in the good faith judgment of the League Office, is willing and able to assume the League Office's obligations under this Agreement."

5. Section 13.1 of the Agreement, under the heading "Choice of Law," shall be amended by the addition of the following new sentence, which shall be considered an integral part of the Agreement:

"The foregoing choice of law should not be considered a waiver of any right conferred upon the League Office or the Team Operator by Article 33 of the General Business Law of the State of New York."

6. Each provision of this Addendum to the Franchise Agreement for the State of New York shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of Article 33 of the General Business Law of the State of New York are met independently, without reference to this

Addendum to the Franchise Agreement for the State of New York, and only to the extent such provision is a then valid requirement of the statute.

IN WITNESS WHEREOF, the parties hereto have caused this Addendum to the Franchise Agreement for the State of New York to be duly executed as of the Effective Date of the Agreement.

LEAGUE OFFICE: USL Super League, LLC By United Soccer Leagues, LLC, its sole member

By:Si	gnature	
Name		
Title		
TEAM O	PERATOR:	
(Name of	Entity)	
By:	gnature	
Si	gnature	
Name		

SCHEDULE 10

ADDENDUM TO THE FRANCHISE AGREEMENT FOR THE STATE OF RHODE ISLAND

In recognition of the requirements of the Rhode Island Franchise Investment Act, Rhode Island Code §§ 19-28.1-1 - 19-28.1-34, the parties to the attached Franchise Agreement (the "Agreement") agree as follows:

1. Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement.

2. Section 13.1 of the Agreement, under the heading "Choice of Law," shall be amended by adding the following language, which shall be considered an integral part of the Agreement, to the conclusion of such Section:

"Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that 'A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

3. Each provision of this Addendum to the Franchise Agreement for the State of Rhode Island shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Rhode Island Franchise Investment Act are met independently, without reference to this Addendum to the Franchise Agreement for the State of Rhode Island, and only to the extent such provision is a then valid requirement of the statute.

IN WITNESS WHEREOF, the parties hereto have caused this Addendum to the Franchise Agreement for the State of Rhode Island to be duly executed as of the Effective Date of the Agreement.

[signature page follows]

LEAGUE OFFICE: USL Super League, LLC By United Soccer Leagues, LLC, its sole member

By:____

Signature

Name

Title

TEAM OPERATOR:

(Name of Entity)

By:___

Signature

Name

Title

SCHEDULE 10

ADDENDUM TO THE FRANCHISE AGREEMENT FOR THE STATE OF WASHINGTON

In recognition of the requirements of the Washington Franchise Investment Protection Act, Revised Code of Washington \$ 19.100.010 – 19.100.940, the parties to the attached Franchise Agreement (the "Agreement") agree as follows:

1. Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement.

2. Section 19.100.180 of the Washington Franchise Investment Protection Act may supersede the Agreement in the Team Operator's relationship with the League Office including the areas of termination and renewal of the franchise. There may also be court decisions which may supersede the Agreement in the Team Operator's relationship with the League Office including the areas of termination and renewal of the franchise.

3. If any provisions of the Agreement are inconsistent with the relationship provisions of Section 19.100.180 or other requirements of the Washington Franchise Investment Protection Act, the provisions of the Washington Franchise Investment Protection Act will prevail over the inconsistent provisions of the Agreement with regard to any franchise sold in Washington.

4. In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act shall prevail.

5. In any arbitration involving a franchise purchased in Washington, the arbitration site shall be either in the State of Washington, or in a place mutually agreed upon at the time of the arbitration, or as determined by the arbitrator.

6. Any release or waiver of rights executed by the Team Operator will not include rights under the Washington Franchise Investment Protection Act except if the Team Operator executes the release or waiver as part of a negotiated settlement after the Agreement is in effect and where the parties are both represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act and/or rights or remedies under the Washington Franchise Investment Protection Act and/or negotiate settlement and the Washington Franchise Investment Protection Act and/or rights or remedies under the Washington Franchise Investment Protection Act, such as a right to a jury trial, may not be enforceable.

7. The League Office may collect the transfer fees if they reflect the League Office's reasonable estimated or actual costs in effecting the Team Operator's transfer.

8. Each provision of this Addendum to the Franchise Agreement for the State of Washington shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Washington Franchise Investment Protection Act are met independently, without reference to this Addendum to the Franchise Agreement for the State of Washington, and only to the extent such provision is a then valid requirement of the statute.

[Signatures begin on the following page.]

IN WITNESS WHEREOF, the parties hereto have caused this Addendum to the Franchise Agreement for the State of Washington to be duly executed as of the Effective Date of the Agreement.

LEAGUE OFFICE: USL Super League, LLC By United Soccer Leagues, LLC, its sole member

By:___

Signature

Name

Title

TEAM OPERATOR:

(Name of Entity)

By:____

Signature

Name

Title

EXHIBIT A-2 – FORM OF OWNER GUARANTY

[See attachment]

Form of Guaranty

This GUARANTY (this "Guaranty"), dated as of ______ is made by ______ is made by ______ ("Guarantor"), in favor and for the benefit of USL Super League, LLC, a Florida limited liability company with its principal place of business at 1715 N. Westshore Blvd., Suite 825, Tampa, FL 33607 ("Beneficiary").

Reference is made to the Franchise Agreement, dated as of _______, 20_____ (as it may be amended, amended and restated, supplemented, or otherwise modified from time to time in accordance with its provisions (the "**Underlying Agreement**"), by and between [Team Operator League Entity], a [State of Organization/Incorporation] [Entity Type] with its principal place of business at [Address] ("**Obligor**"), and Beneficiary. In consideration of the substantial direct and indirect benefits derived by each Guarantor from the transactions in the Underlying Agreement, and in order to induce Beneficiary to enter into the Underlying Agreement, each Guarantor, jointly and severally with each other Guarantor, hereby agrees as follows:

1. <u>Guaranty</u>. Each Guarantor jointly and severally absolutely, unconditionally and irrevocably guarantees to Beneficiary the full and punctual payment and performance of all present and future obligations, liabilities, covenants and agreements required to be observed and performed or paid or reimbursed by Obligor under or relating to the Underlying Agreement, plus all costs, expenses and fees (including the reasonable fees and expenses of Beneficiary's counsel) in any way relating to the enforcement or protection of Beneficiary's rights hereunder (collectively, the "**Obligations**").

2. <u>Guaranty Absolute and Unconditional</u>. Each Guarantor agrees that its Obligations under this Guaranty are irrevocable, continuing, absolute and unconditional and shall not be discharged or impaired or otherwise affected by, and each Guarantor hereby irrevocably waives any defenses to enforcement it may have (now or in the future) by reason of:

(a) Any illegality, invalidity or unenforceability of any Obligation or the Underlying Agreement or any related agreement or instrument, or any law, regulation, decree or order of any jurisdiction or any other event affecting any term of the Obligations.

(b) Any change in the time, place or manner of payment or performance of, or in any other term of the Obligations, or any rescission, waiver, release, assignment, amendment or other modification of the Underlying Agreement.

(c) Any taking, exchange, substitution, release, impairment, amendment, waiver, modification or non-perfection of any collateral or any other guaranty for the Obligations, or any manner of sale, disposition or application of proceeds of any collateral or other assets to all or part of the Obligations.

(d) Any default, failure or delay, willful or otherwise, in the performance of the Obligations.

(e) Any change, restructuring or termination of the corporate structure, ownership or existence of Guarantor or any insolvency, bankruptcy, reorganization or other similar proceeding affecting Obligor or its assets or any resulting restructuring, release or discharge of any Obligations.

(f) Any failure of Beneficiary to disclose to Guarantor any information relating to the business, condition (financial or otherwise), operations, performance, properties or prospects of Obligor now or hereafter known to Beneficiary, Guarantor waiving any duty of Beneficiary to disclose such information.

(g) The failure of any other guarantor or third party to execute or deliver this Guaranty or any other guaranty or agreement, or the release or reduction of liability of Guarantor or any other guarantor or surety with respect to the Obligations.

(h) Any delay or failure of or forbearance by Beneficiary in asserting any claim or demand or in exercising or enforcing any right or remedy, whether by action, inaction or omission, under the Underlying Agreement or otherwise.

(i) The existence of any claim, set-off, counterclaim, recoupment or other rights that Guarantor or Obligor may have against Beneficiary (other than a defense of payment or performance) whether in connection with the Obligations, the Underlying Agreement or any other transaction.

(j) Any other circumstance (including, without limitation, any statute of limitations), act, omission or manner of administering the Underlying Agreement or any existence of or reliance on any representation by Beneficiary that might vary the risk of Guarantor or otherwise operate as a defense available to, or a legal or equitable discharge of, Guarantor.

3. <u>Certain Waivers; Acknowledgments</u>. Guarantor further acknowledges and agrees as follows:

(a) Guarantor hereby unconditionally and irrevocably waives any right to revoke this Guaranty and acknowledges that this Guaranty is continuing in nature and applies to all presently existing and future Obligations, until the complete, irrevocable and indefeasible payment and satisfaction in full of the Obligations.

(b) This Guaranty is a guarantee of payment and performance and not of collection. Beneficiary shall not be obligated to enforce or exhaust its remedies against Obligor or under the Underlying Agreement before proceeding to enforce this Guaranty.

(c) This Guaranty is a direct guaranty and independent of the obligations of Obligor under the Underlying Agreement. Beneficiary may resort to any Guarantor or all Guarantors for payment and performance of the Obligations whether or not Beneficiary shall have resorted to any collateral therefor or shall have proceeded against Obligor or any other guarantors with respect to the Obligations. Beneficiary may, at Beneficiary's option, proceed against Guarantor(s) and Obligor, jointly and severally, or against Guarantor(s) only without having obtained a judgment against Obligor.

(d) Each Guarantor hereby unconditionally and irrevocably waives promptness, diligence, notice of acceptance, presentment, demand for performance, notice of non-performance, default, acceleration, protest or dishonor and any other notice with respect to any of the Obligations and this Guaranty and any requirement that Beneficiary protect, secure, perfect or insure any lien or any property subject thereto.

(e) Each Guarantor agrees that its guaranty hereunder shall continue to be effective or be reinstated, as the case may be, if at any time all or part of any payment of any Obligation is voided, rescinded or recovered or must otherwise be returned by Beneficiary upon the insolvency, bankruptcy or reorganization of Obligor.

(f) Each Guarantor agrees to be individually bound by all of the covenants contained in Section 4.22 of the Underlying Agreement. Each Guarantor agrees that such covenants are fair and reasonable and will not deprive such Guarantor of his or her livelihood.

4. <u>Subrogation</u>. Each Guarantor waives and shall not exercise any rights that it may acquire by way of subrogation, contribution, reimbursement or indemnification for payments made under this Guaranty until all Obligations shall have been paid and discharged in full. Subject to the foregoing, upon payment and performance of all such Obligations, each Guarantor shall be subrogated to the rights of Beneficiary against Obligor, and Beneficiary agrees to take such steps as Guarantors may reasonably request, at Guarantors' expense, to implement such subrogation.

5. <u>Representations and Warranties</u>. To induce Beneficiary to enter into the Underlying Agreement, Each Guarantor represents and warrants that: [(a) Guarantor is a duly organized and validly existing [ENTITY TYPE] in good standing under the laws of the jurisdiction of its organization; (b) this Guaranty constitutes Guarantor's valid and legally binding agreement in accordance with its terms; (c) the execution, delivery and performance of this Guaranty have been duly authorized by all necessary action and will not violate any order, judgment or decree to which Guarantor or any of its assets may be subject; and (d) Guarantor is currently solvent and will not be rendered insolvent by providing this Guaranty.]

6. <u>Notices</u>. All notices, requests, consents, demands and other communications hereunder (each, a "**Notice**") shall be in writing and delivered to the parties at the addresses set forth herein or to such other address that may be designated by the receiving party in writing. All Notices shall be delivered by personal delivery, nationally recognized overnight courier, e-mail or certified or registered mail (return receipt requested, postage prepaid). Except as otherwise provided in this Guaranty, a Notice is effective only (a) with written confirmation of delivery or transmission; (b) upon receipt of the receiving party, and (c) if the party giving the Notice has complied with the requirements of this section.

7. <u>Assignment</u>. This Guaranty shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; provided, however, that a Guarantor may not, without the prior written consent of Beneficiary, assign any of its rights, powers or obligations hereunder. Any attempted assignment in violation of this section shall be null and void.

8. <u>Governing Law; Service of Process</u>. THIS GUARANTY SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF FLORIDA, WITHOUT REFERENCE TO ANY CHOICE OF LAW DOCTRINE. EACH PARTY IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN SECTION 6 HEREOF AND AGREES THAT NOTHING HEREIN SHALL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY MANNER PERMITTED BY APPLICABLE LAW.

9. <u>Submission to Jurisdiction</u>. Each Guarantor hereby irrevocably and unconditionally agrees that any legal action, suit or proceeding arising out of or relating to this Guaranty or the Underlying Agreement may be brought in the courts of the State of Florida or of the United States of America for the Middle District of Florida. By the execution and delivery of this Guaranty, each Guarantor hereby irrevocably and unconditionally submits to the exclusive jurisdiction of any such court in any such action, suit or proceeding. Final judgment against a Guarantor in any action, suit or proceeding shall be conclusive and may be enforced in any other jurisdiction, including the country in which such Guarantor is domiciled, by suit on the judgment.

10. <u>Venue</u>. Each Guarantor irrevocably waives, to the fullest extent permitted by applicable law, any objection which it may now or hereafter have to the laying of venue of any action, suit or proceeding arising out of or relating to this Guaranty or the Underlying Agreement, brought in the courts of the State of Florida or in the United States District Court for the Middle District of Florida, and any claim that any such action, suit or proceeding brought in any such court has been brought in an inconvenient forum.

11. <u>Process Agent [ONLY INCLUDED WHEN OWNER IS NOT DOMICILED IN THE UNITED</u> <u>STATES]</u>.

(a) [NAMED GUARANTOR] hereby designates, appoints and empowers [NAME OF AGENT], (the "**Process Agent**") with offices as of the date of this Guaranty at [MUST BE US ADDRESS], as its authorized agent to receive for and on its behalf service of summons or other legal process in any such action, suit or proceeding in the State of Florida. Such service may be made by mailing or delivering a copy of such process to such Guarantor in case of the Process Agent at the Process Agent's above address, and such Guarantor hereby irrevocably authorizes and directs the Process Agent to accept such service on its behalf.

(b) Such Guarantor covenants and agrees that, for so long as it shall be bound under this Guaranty, it shall maintain a duly appointed agent for the service of summons and other legal process in the State of Florida, United States of America, for the purposes of any legal action, suit or proceeding brought by any party in respect of this Guaranty and shall keep Beneficiary advised of the identity and location of such agent. If for any reason there is no authorized agent for service of process in Florida, such Guarantor irrevocably consents, to the service of process out of the said courts by mailing copies thereof by registered United States air mail postage prepaid to it at its address specified in the opening paragraph of this Guaranty.

12. <u>Waiver of Jury Trial</u>. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHTS TO TRIAL BY JURY WITH RESPECT TO ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY OR ANY OF THE OBLIGATIONS HEREUNDER.

13. <u>Cumulative Rights</u>. Each right, remedy and power hereby granted to Beneficiary or allowed it by applicable law or other agreement shall be cumulative and not exclusive of any other, and may be exercised by Beneficiary at any time or from time to time.

14. <u>Severability</u>. If any provision of this Guaranty is to any extent determined by final decision of a court of competent jurisdiction to be unenforceable, the remainder of this Guaranty shall not be affected thereby, and each provision of this Guaranty shall be valid and enforceable to the fullest extent permitted by law.

15. <u>Entire Agreement; Amendments; Headings; Effectiveness</u>. This Guaranty constitutes the sole and entire agreement of each Guarantor and Obligor with respect to the subject matter hereof and supersedes all previous agreements or understandings, oral or written, between each Guarantor and Obligor with respect to such subject matter. No amendment or waiver of any provision of this Guaranty shall be valid and binding unless it is in writing and signed, in the case of an amendment, by both parties, or in the case of a waiver, by the party against which the waiver is to be effective. Section headings are for convenience of reference only and shall not define, modify, expand or limit any of the terms of this Guaranty. Delivery of this Guaranty by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed original of this Guaranty.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, Guarantor has executed this Guaranty as of the day and year first above written.

GUARANTOR:

Sworn to before me this ____ day of _____, ___, by _____

NOTARY PUBLIC

PRINT, TYPE OR STAMP NAME OF NOTARY PUBLIC

Personally Known _____ OR Produced Identification _____

Type of Identification Produced:

_____.

EXHIBIT A-3 – FORM OF LEASE ADDENDUM

[See attachment]

Lease Addendum

THIS LEASE ADDENDUM (this "A	Addendum") is made and entered into as of the day of
, 20, by and among	(" <u>Landlord</u> "),
	("Tenant"), and USL Super League, LLC, a Florida
limited liability company (the " <u>USL</u> ").	

WITNESSETH:

WHEREAS, Landlord and Tenant are parties to that certain	, dated
, 20 (the "Lease"), for the premises located at	(the " <u>Premises</u> ");

WHEREAS, Tenant and the USL are or will be parties to a Franchise Agreement (as amended, supplemented, renewed, or otherwise modified from time to time, the "Franchise Agreement"), pursuant to which the USL granted or will grant Tenant the right to operate a soccer team that participates in the professional soccer league currently known as the "USL Super League" (the "League");

WHEREAS, Landlord, Tenant and the USL, desire to provide the USL the opportunity to preserve the Premises as a venue for hosting USL soccer matches, and to assure Landlord that, if the USL exercises the option herein contained, any defaults of Tenant under the Lease will be cured by the USL before it takes possession of the Premises; and

WHEREAS, Tenant and Landlord desire to amend the Lease for the mutual benefit of both parties, as more specifically set forth herein;

NOW, THEREFORE, for and in consideration of the mutual promises and representations contained herein, and other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Collateral Assignment. Landlord acknowledges that Tenant intends to utilize the Premises 1. to host USL-sanctioned soccer matches and other related events, and that Tenant's rights to operate a soccer team that participates in the League and to use the League name, trademarks and service marks solely pursuant to the Franchise Agreement. Tenant's operations at the Premises are independently owned and operated. Landlord acknowledges that Tenant alone is responsible for all obligations under the Lease unless and until the USL or another franchisee expressly assumes such obligations and takes actual possession of the Premises. Notwithstanding any provisions of the Lease to the contrary, Landlord hereby consents, without payment of a fee and without the need for further Landlord consent, to (i) the collateral assignment of Tenant's interest in this Lease to the USL to secure Tenant's obligations to the USL under the Franchise Agreement, and/or (ii) the USL's succeeding to Tenant's interest in the Lease as a result of the USL's exercise of rights remedies under such collateral assignment or as a result of the USL's termination of, or exercise of rights or remedies granted in or under, any other agreement between the USL and Tenant, and/or (iii) Tenant's, the USL's and/or any other franchisee of the USL's assignment of the Lease to another franchisee of the USL with whom the USL has executed its then-standard franchise agreement. Landlord, Tenant and the USL agree and acknowledge that simultaneously with such assignment pursuant to the immediately preceding sentence, the USL shall be released from all liability under the Lease or otherwise accruing after the date of such assignment (in the event the USL is acting as the assignor under such assignment), but neither Tenant nor any other franchisee shall be afforded such release in the event Tenant/such franchisee is the assignor unless otherwise agreed by Landlord. Landlord further agrees that all unexercised renewal or extension rights shall not be terminated in the event of any assignment referenced herein, but shall inure to the benefit of the applicable assignee.

2. <u>Renewal of Lease</u>. Throughout the term of the Franchise Agreement and any renewals thereto, Tenant agrees that it shall elect and exercise all options to extend the term of or renew the Lease not less than 30 days prior to the last day that the option must be exercised, unless the USL otherwise agrees in writing. If the USL does not otherwise agree in writing, and upon failure of Tenant to so elect to extend or renew the Lease as aforesaid, Tenant hereby appoints the USL as its true and lawful attorney-in-fact to exercise such extension or renewal options in the name, place and stead of Tenant for the purpose of effecting such extension or renewal; <u>provided</u>, <u>however</u>, that the USL shall have no obligation to renew the Lease, or liability for renewing or failing to renew the Lease.

3. <u>No Further Modification of Lease</u>. Except as otherwise provided for in this Addendum, Tenant and Landlord agree that neither party shall renew or extend the term of the Lease or make any other modifications or alterations to the Lease without the prior written consent of the USL.

4. <u>Entry into Premises</u>. Tenant and Landlord agree that the USL shall have the right to enter the Premises to make any reasonable modifications or alterations necessary to protect the USL's interest in the proprietary marks or to cure any default under the Franchise Agreement or under the Lease, and Tenant and Landlord agree that the USL shall not be liable for trespass or any other crime or tort. If the Lease or Franchise Agreement expires or is terminated for any reason and the USL fails to exercise the rights under <u>Paragraph 1</u> above, Tenant agrees to de-identify the Premises as the home venue of Tenant's USL team and to promptly remove the USL's trademarks and trade dress from the Premises. The USL may enter upon the Premises without being guilty of trespass or tort to effect such de-identification if Tenant fails to effect de-identification within 10 days after receipt of written demand from the USL, following termination or expiration of the Franchise Agreement or Lease. Tenant shall reimburse the USL for its reasonable costs and expenses in effecting de-identification.

5. <u>Interference</u>. Notwithstanding the provisions of any part of the Lease, Landlord shall not: (i) interfere with the ingress or egress of the Premises; (ii) disrupt Tenant's business; (iii) reduce the usable area of the Premises; or (iv) reduce the number of parking spaces that currently exist for the Premises.

6. <u>Force Majeure</u>. Landlord and Tenant shall be excused for the period of delay in the performance of any of their respective obligations hereunder or under the Lease, excepting monetary obligations, and shall not be considered in default, when prevented from so performing due to a labor strike, riot, war, fire, flood or other casualty, or Acts of God so extensive as to prevent Tenant from conducting business or preventing Tenant or Landlord from complying with their obligations hereunder or under the Lease.

7. <u>Landlord Indemnity</u>. Except for the act, omission or negligence of Tenant or its agents, Landlord shall indemnify and hold harmless Tenant, its shareholders, officers, directors, members, agents, contractors, employees, and invitees (the "<u>Tenant Indemnitees</u>"), from and against any and all claims arising from Landlord's use or ownership of the Premises, or from other tenants' use of the Premises, and shall further indemnify and hold harmless the Tenant Indemnitees from and against any and all claims arising from any breach or default in the performance of any obligation on Landlord's part to be performed under the terms of the Lease, or arising from any act, omission or negligence of the Landlord, or any of Landlord's agents, contractors, or employees, and from and against all costs, attorney's fees, expenses and liabilities incurred in the defense of any such claim or any action or proceeding brought thereon; and in case of action or proceeding be brought against Tenant by reason of any such claim, Landlord, upon written notice from Tenant, shall defend the same at Landlord's expense.

8. <u>Notices</u>. Notwithstanding anything contained in the Lease to the contrary, all notices and demands required to be delivered hereunder will be deemed so delivered (i) at the time delivered, if delivered by hand; (ii) one business day after being placed on the hands of a commercial courier service for

next business day delivery; or (iii) three business days after placement in the United States Mail by registered or certified mail, return receipt request, postage prepaid and must be addressed to the party to be notified at the following addresses unless and until a different address has been designated by written notice to the other parties.

If directed to Tenant, the notice shall be addressed to:

Attention:

If directed to Landlord, the notice shall be addressed to:

Attention:

If directed to the USL, the notice shall be addressed to: USL Super League, LLC 1715 N. Westshore Blvd. Suite 825 Tampa, Florida 33607 Attention: Legal

9. <u>Third Party Beneficiary</u>. Landlord and Tenant each agree that the USL is an express third party beneficiary of the Lease, and that the USL may enforce its rights as third party beneficiary thereunder against Landlord and Tenant.

10. <u>Full Force and Effect</u>. Except as otherwise provided for herein, the terms and conditions of the Lease remain in full force and effect.

11. <u>Binding on Successors</u>. The covenants, agreements, terms, provisions and conditions contained in the Lease, as modified by this Addendum, shall bind and inure to the benefit of all parties hereto, and their respective successors and assigns.

[Signatures Appear on Following Page]

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Addendum the day and year first written above.

LANDLORD:

[INSERT ENTITY NAME]

By:_____

Name:_____

Title:_____

TENANT:

[INSERT ENTITY NAME]

By:		
Name:		
Title:		

USL: USL Super League, LLC

By:_____

Name:_____ Title:_____

EXHIBIT B – DIRECTOR OF STATE ADMINISTRATORS & AGENCIES

California

Department of Business Oversight 320 West 4th Street, Suite 750 Los Angeles, California 90013 (866) 275-2677

Hawaii

Department of Commerce & Consumer Affairs P.O. Box 40 Honolulu, Hawaii 96810 Telephone: (808) 586-2272

Illinois

Office of Attorney General 500 S. Second Street Springfield, Illinois 62706 (217) 782-4465

Indiana

Franchise Section Securities Division 302 W. Washington St., Room E-111 Indianapolis, Indiana 46204 (317) 232-6681

Maryland

Office of Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202 (410) 576-7042

Michigan

Consumer Protection Division Department of Attorney General 525 W. Ottawa St. P.O. Box 30213 Lansing, Michigan 48909 (517) 373-7117

Minnesota

Department of Commerce 85 7th Place East, Suite 500 St. Paul, Minnesota 55101-2198 (651) 296-4026

New York

Bureau of Investor Protection and Securities Department of Law 120 Broadway, 23rd Floor New York, New York 10271 (212) 416-8211

North Dakota

North Dakota Securities Department 600 East Blvd., Fifth Floor Bismarck, North Dakota 58505 (701) 328-4712

Rhode Island

Securities Division 233 Richmond Street, Suite 232 Providence, Rhode Island 02903-4232 (401) 222-3048

South Dakota

Department of Labor and Regulation Division of Securities 124 S. Euclid, Suite 104 Pierre, SD 57501 (605) 773-4823

Virginia

State Corporation Commission Division of Securities and Retail Franchising 1300 East Main Street, 9th Floor Richmond, Virginia 23219 (804) 371-9051

Washington

Department of Financial Institutions Securities Division 150 Israel Road, S.W. Tumwater, Washington 98501 (360) 902-8760

Wisconsin

Division of Securities Department of Financial Institutions 345 W. Washington, 4th Floor Madison, Wisconsin 53703 (608) 266-8559

EXHIBIT C – AGENT FOR SERVICE OF PROCESS

California

California Department of Business Oversight Commissioner of Business Oversight 320 West 4th Street, Suite 750 Los Angeles, California 90013-2344

Illinois

Illinois Attorney General 500 South Second Street Springfield, Illinois 62706

Indiana

Secretary of State of Indiana 201 State House 200 West Washington Street Indianapolis, Indiana 46204

Maryland

Maryland Securities Commissioner 200 St. Paul Place Baltimore, Maryland 21202-2020

Michigan

Michigan Department of Commerce Corporations and Securities Bureau 6546 Mercantile Way Lansing, Michigan 48910

Minnesota

Minnesota Commissioner of Commerce Department of Commerce 85 7th Place, Suite 500 St. Paul, Minnesota 55101

New York

Secretary of State of New York 99 Washington Avenue Albany, New York 12231

North Dakota

Securities Commissioner North Dakota Securities Department 600 East Boulevard Avenue State Capitol 5th Floor Department 414 Bismarck, ND 58505

Oregon

Department of Insurance and Finance 700 Summer Street, N.E. Suite 120 Salem, Oregon 97310

Rhode Island

Director of Department of Business Regulation 1511 Pontiac Avenue John O. Pastore Complex – Building 69-1 Cranston, Rhode Island 02920

South Dakota

Department of Labor and Regulation Division of Securities 124 S. Euclid, Suite 104 Pierre, SD 57501

Virginia

Clerk of the State Corporation Commission 1300 East Main Street, First Floor Richmond, Virginia 23219

Washington

Securities Administrator Washington State Department of Financial Institutions 150 Israel Road Tumwater, Washington 98501

Wisconsin

Wisconsin Commissioner of Securities 345 West Washington Avenue, 4th Floor Madison, Wisconsin 53703

EXHIBIT D – LIST OF CURRENT FRANCHISEES

As of December 31, 2022

Not applicable.

EXHIBIT E – FRANCHISES THAT HAVE TRANSFERRED CONTROLLING OWNERSHIP, BEEN CANCELLED OR TERMINATED, HAVE NOT BEEN RENEWED, BEEN ACQUIRED OR OTHERWISE CEASED TO DO BUSINESS

Not Applicable

EXHIBIT F-1 – FINANCIAL STATEMENTS

[See attachment]





CERTIFIED PUBLIC ACCOUNTANTS AND CONSULTANTS

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES

TAMPA, FLORIDA

CONSOLIDATED FINANCIAL STATEMENTS

DECEMBER 31, 2022 AND 2021



UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES

TAMPA, FLORIDA

CONSOLIDATED FINANCIAL STATEMENTS

DECEMBER 31, 2022 AND 2021

CONTENTS

	PAGE
Independent Auditor's Report	1
Consolidated Balance Sheets	3
Consolidated Statements of Income	4
Consolidated Statements of Members' Equity	5
Consolidated Statements of Cash Flows	6
Notes to Consolidated Financial Statements	7



INDEPENDENT AUDITOR'S REPORT

Board of Directors United Soccer Leagues, LLC and Subsidiaries Tampa, Florida

Opinion

We have audited the accompanying consolidated financial statements of United Soccer Leagues, LLC and subsidiaries (the "Company"), which comprise the consolidated balance sheets as of December 31, 2022 and 2021, and the related consolidated statements of income, changes in members' equity, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of the Company as of December 31, 2022 and 2021, and the results of their operations and their cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are available to be issued.

-1-

Saltmarsh, Cleaveland & Gund

saltmarshcpa.com | Since 1944 | (800) 477-7458

Board of Directors United Soccer Leagues, LLC and Subsidiaries

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements, including omissions, are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Saltmarsh Cleansland & bunk

Tampa, Florida February 15, 2023

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS DECEMBER 31, 2022 AND 2021

ASSETS

		2022	2021
Current Assets:			
Cash and restricted cash	\$	10,691,234	\$ 12,996,096
Accounts receivable, net		7,877,111	6,800,317
Prepaid expenses		391,321	228,367
Right of use asset - operating		242,449	-
Employee Retention Credit receivable		512,188	915,418
Total current assets		19,714,303	 20,940,198
Other Assets:			
Long-term receivables, net		4,529,500	2,856,793
Furniture and equipment, net		129,677	242,391
Intangible assets, net		1,263,567	1,274,317
Goodwill, net		203,315	304,973
Other assets		19,398	19,898
Total other assets		6,145,457	 4,698,372
Total Assets	\$	25,859,760	\$ 25,638,570
LIABILITIES AND MEMBERS' EQU	ITY		
Current Liabilities:			
Accounts payable	\$	1,219,515	\$ 735,893
Operating lease liability		307,483	-
Due to related party		-	25,075
Performance security deposits		1,835,659	1,863,714
Deferred revenue		4,589,632	4,437,647
Accrued liabilities		1,337,480	1,099,684
Total current liabilities		9,289,769	 8,162,013
Long Term Liabilities:			
Related party loan		5,761,518	5,761,518
Other liability		-	500,000
Total long term liabilities		5,761,518	 6,261,518
Total liabilities		15,051,287	14,423,531
Members' Equity		10,808,473	 11,215,039
Total Liabilities and Members' Equity	\$	25,859,760	\$ 25,638,570

The accompanying notes are an integral

part of these consolidated financial statements.

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME YEARS ENDED DECEMBER 31, 2022 AND 2021

	2022		2021
Revenue:			
Sales of initial franchises	\$	14,602,000	\$ 9,141,250
League participation fees		9,037,574	9,400,032
Other revenue		5,858,551	4,142,929
Total revenue		29,498,125	 22,684,211
Expenses:			
Payroll expense		9,596,184	6,995,218
Operating expenses		13,918,840	8,945,732
Broadcast and TV production		1,637,625	1,709,827
Legal and professional fees		731,900	565,065
Rent expense		452,834	424,803
Total expenses		26,337,383	 18,640,645
Income From Operations		3,160,742	 4,043,566
Other Income (Expense):			
Interest expense		(864,180)	(870,363)
Interest income		53,905	170,105
Forgiveness of Paycheck Protection Program loan		-	840,370
Employee Retention Credit		-	1,660,604
Other income		7,647	18,464
Total other income (loss)		(802,628)	 1,819,180
Net Income	\$	2,358,114	\$ 5,862,746

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF MEMBERS' EQUITY YEARS ENDED DECEMBER 31, 2022 AND 2021

	2022		 2021
Balance at Beginning of Year	\$	11,215,039	\$ 5,386,994
Net income		2,358,114	5,862,746
Distributions to members		(2,764,680)	 (34,701)
Balance at End of Year	\$	10,808,473	\$ 11,215,039

The accompanying notes are an integral part of these consolidated financial statements.

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS YEARS ENDED DECEMBER 31, 2022 AND 2021

		2022		2021
Cash Flows From Operating Activities:	^		<i>•</i>	
Net income	\$	2,358,114	\$	5,862,746
Adjustments to reconcile net income to net cash				
provided by operating activities:		220 515		210 170
Depreciation and amortization		229,515		219,179
Bad debt expense		3,802,581		261,548
Loss on disposal of furniture and equipment		5,757		4,083
Imputed interest - long-term receivables		(53,905)		(170,105)
Changes in -		(4,879,375)		(2,0.47,2.41)
Accounts receivable, net Prepaid expenses		(4,879,373) (162,954)		(2,047,341) 440,604
Employee Retention Credit receivable		403,230		(915,418)
Long-term receivables, net		(1,618,802)		
Right of use asset - operating		360,726		319,159
Other assets		500,720		(500)
		483,622		(300)
Accounts payable Operating lease liability		(447,065)		397,721
Due to related party		(447,003) (25,075)		(259,850)
Performance security deposits		(23,073) (28,055)		
Deferred revenue		(28,033)		(2,503) (304,232)
Accrued liabilities		389,169		(304,232) 93,129
Other liability		(500,000)		500,000
Net cash provided by operating activities		469,968		4,598,220
Net easil provided by operating activities		409,908		4,398,220
Cash Flows From Investing Activities:				
Purchases of furniture and equipment		(10,150)		(84,560)
Net cash used in investing activities		(10,150)		(84,560)
Cash Flows From Financing Activities:				
Repayments on line of credit		-		(1,000,000)
Member distributions		(2,764,680)		(34,701)
Net cash used in financing activities		(2,764,680)		(1,034,701)
Net Increase (Decrease) in Cash and Restricted Cash		(2,304,862)		3,478,959
Cash and Restricted Cash at Beginning of Year		12,996,096		9,517,137
Cash and Restricted Cash at End of Year	\$	10,691,234	\$	12,996,096
Supplemental Disclosure of Cash Flow Information: Cash paid during the year for interest	\$	864,180	\$	870,363
Supplemental Disclosure of Noncash Investing and Financing Activities: Recognition of right to use asset - operating upon adoption of ASC 842	\$	754,548	\$	

The accompanying notes are an integral part of these consolidated financial statements.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Nature of Business:

USISL, Inc., (the Corporate Predecessor) was incorporated on April 1, 1996. On August 27, 2009, NuRock Soccer Holdings, LLC ("NuRock") and Robert Hoskins (collectively "the Members") purchased all of the outstanding stock of the Corporate Predecessor and subsequently converted the Corporate Predecessor to United Soccer Leagues, LLC, ("USL"), a limited liability company. In February 2018, the Company amended and restated their operating agreement, effective January 1, 2018 amending the Member's profit and loss percentage interests to be 50.5% to NuRock, 0.5% to Robert Hoskins and 49% to Papadakis Family Trust, LLC. USL operates soccer leagues collectively known as United Soccer Leagues. USL sells and grants franchises to owners of soccer teams throughout the United States and Canada and manages league activities. USL is sanctioned by the United States Soccer Federation (the "Federation") and has been a member of the Federation for over twenty-five years.

The liability of the Members is limited to the amount of their invested capital and Members have no obligation to make additional capital contributions to USL. Net profits and losses are allocated among the Members based on the Members' respective profit and loss interests.

In January 2016, USL created a wholly owned subsidiary, RRFC Holdings, LLC. There was no activity until May 14, 2019 at which time the name was changed to ABQ Holdings, LLC ("ABQ"). ABQ was used to run the operations of a League Two team, Albuquerque Del Sol, for the 2019 season. There was no activity for the 2022 or 2021 season.

On January 1, 2017, USL created five Florida limited liability companies (USL Pro, LLC; USL Pro 2, LLC; USL Productions, LLC; Super Y League, LLC; and Premier Development League, LLC, collectively the "Newcos") as wholly owned subsidiaries of USL. USL restructured by contributing certain assets and assigning certain liabilities to the Newcos as set forth in the Operating and Contribution agreements.

In March 2019, USL created a wholly owned subsidiary, USL Kids, LLC ("USL Kids"). USL Kids was originally created to run a new soccer youth program. This program has been discontinued but USL Kids is still in existence.

On September 23, 2021, USL created a wholly owned subsidiary, USL W League, LLC. On October 8, 2021, USL created two wholly owned subsidiaries, USL Academy, LLC and USL Super League, LLC.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Principles of Consolidation:

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America and include the accounts of USL and its wholly owned subsidiaries (collectively known as the "Company"). Intercompany balances and transactions have been eliminated in consolidation.

Estimates:

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Accounts Receivable:

Accounts receivable are carried at original invoice amount less estimated allowances for doubtful accounts based on a review of all outstanding amounts. Management determines the allowance for doubtful accounts by regularly evaluating individual customer receivables and considering a customer's financial condition and credit history, and current economic conditions. Accounts receivable are written off when deemed uncollectible. Recoveries of accounts receivable previously written off are recorded when received. The allowance for doubtful accounts was \$525,374 and \$635,000 as of December 31, 2022 and 2021, respectively. Accounts receivable written off amounted to \$3,802,581 and \$261,548 for the years ended December 31, 2022 and 2021, respectively.

Furniture and Equipment:

Furniture and equipment are stated at cost. Depreciation expense is computed using the straight-line method over the estimated useful lives of the related assets, which range from 2 to 7 years. Depreciation charged to operations for 2022 and 2021 was \$117,108 and \$106,772, respectively. The cost value of furniture and equipment and associated accumulated depreciation are as follows:

	2022			2021
Furniture and equipment Less: accumulated depreciation	\$	743,099 613,422	\$	750,677 508,286
Furniture and equipment, net	\$	129,677	\$	242,391

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Furniture and Equipment (Continued):

When assets are retired or otherwise disposed of, the cost and related accumulated depreciation are removed from the accounts, and any resulting gain or loss is reflected in income for the period. The cost of maintenance and repairs is charged to expense as incurred; significant renewals and betterments are capitalized.

Intangible Assets:

Intangible assets are recorded at cost and consist of trademarks, which have indefinite lives, and franchise agreements, which are amortized using the straight-line method over their estimated useful life of 20 years. The Company determines the useful lives of intangible assets after considering the specific facts and circumstances related to each intangible asset. Factors the Company considers when determining useful lives include the contractual term of any agreement related to the asset, the historical performance of the asset, the Company's long-term strategy for using the asset, any laws or other local regulations which could impact the useful life of the asset, and other economic factors, including competition and specific market conditions.

The Company tests intangible assets determined to have indefinite useful lives for impairment annually, or more frequently if events or circumstances indicate that assets might be impaired. For indefinite-lived intangible assets, other than goodwill, if the carrying amount exceeds the fair value, an impairment charge is recognized in an amount equal to that excess. The Company believes that there has been no impairment of indefinite-lived intangible assets in 2022 and 2021.

Goodwill:

Goodwill consists of the excess of the purchase price of the Corporate Predecessor over the fair value of the net tangible and intangible assets acquired in the purchase. Goodwill is amortized on a straight-line basis for 10 years and analyzed for impairment at the entity level if a triggering event occurs.

Leases:

ASC 842, "*Leases*" ("ASC 842") requires a lessee to recognize a right-of-use asset and a lease liability for all leases with a term greater than 12 months on its balance sheet regardless of whether the lease is classified as financing or operating.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Leases (Continued):

The Company's only lessee arrangement is an operating lease for the Company's facility. Under this arrangement, the Company recorded the right-of-use asset and corresponding lease liability, based on the present value of the remaining lease payments discounted using the risk-free rate practical expedient allowable under ASC 842. The right-of-use asset is reported in noncurrent assets and the related lease liability is reported in current liabilities on the Balance Sheet. All leases are recorded on the Balance Sheet except for leases with an initial term less than 12 months for which the Company elected short-term lease recognition under ASC 842. Lease terms may contain renewal and extension options and early termination features. Lease expense is recognized on a straight-line basis over the lease term.

Revenue Recognition:

The Company's revenues consist of initial franchise fees, league participation fee, and other revenue such as special events, hotel rebates, sponsorships, merchandise sales, and miscellaneous revenue.

Accounting Standards Update No. 2014-09, *Revenue from Contracts with Customers (Topic 606)* ("ASC 606") provides that revenues are to be recognized when control of promised goods or services is transferred to a customer in an amount that reflects the consideration expected to be received for those goods or services. ASC 606 does not impact the Company's recognition of any of its revenue sources. Management has determined that all performance obligations related to the initial fees from a franchise sale are met on the date the agreement is signed. Accordingly, revenue related to the sale of initial franchises is recognized as income at a point in time. League participation fees are recognized as income during the year in which the season is played. Revenue from hotel rebates is recognized when the related teams' stays have been completed. Revenue from special events and sponsorships is recognized upon completion of the associated event. Merchandise sales are recorded when the merchandise is delivered to the customer.

Deferred revenue is recorded for transactions where cash has been received prior to the recognition of revenue. Deferred revenue included in the consolidated balance sheets relates to team pre-payments received in advance, and franchise fees received prior to all conditions of the franchise agreement being satisfied.

Income Taxes:

Under provisions of the Internal Revenue Code, limited liability companies that are treated as partnerships are not subject to income taxes, and any income or loss realized is taxed to the individual Members. Accordingly, no provision for income taxes is reflected in these financial statements. Taxable income is passed through to and reportable by the Members.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Income Taxes (Continued):

The Company evaluates its tax positions for uncertainty on an annual basis and believes it has no uncertain tax positions as of December 31, 2022 and 2021.

Recent Accounting Pronouncements:

In February 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2016-02 - *Leases (Topic 842)*. The guidance in this topic supersedes the requirements in ASC *Topic 840, Leases* ("ASC 840"). With the exception of short-term leases, the updated guidance requires lessees to recognize a lease liability representing the lessee's obligation to make lease payments arising from a lease, measured on a discounted basis, and a right-of-use asset representing the lessee's right to use, or control the use of, a specified asset for the lease term upon adoption. Lessor accounting was largely unchanged under the new guidance, except for clarification of the definition of initial direct costs which provided additional guidance on the timing of recognition of those costs. Subsequent to the issuance of this update, the FASB issued three additional ASUs that provide codification improvements and certain transition elections. The Company adopted the standard effective January 1, 2022, using the modified retrospective transition method permitted by ASU 2018-11. Upon adoption, the Company recognized an operating lease liability of \$754,548 based on the present value of the remaining minimum rental payments under current leasing standards for existing operating leases, as well as a corresponding right-of-use asset of \$603,175, the \$151,373 difference is attributable to the offset of the deferred rent existing at January 1, 2022. The Company has included newly applicable lease disclosures in Note 5.

Subsequent Events:

The Company has evaluated subsequent events through February 15, 2023, the date which the consolidated financial statements were available to be issued.

NOTE 2 - CASH AND RESTRICTED CASH

For the majority of the franchisees, the Company requires either a letter of credit or a cash deposit with the Company for a specified amount to cover any required future fees due from the franchisees and not paid to the Company as provided for in the franchise agreement. The cash deposits are recorded as a performance security deposit liability on the balance sheets with balances of \$1,835,659 and \$1,863,714 as of December 31, 2022 and 2021, respectively. The cash associated with the performance security deposit liabilities is maintained in separate restricted cash accounts.

NOTE 3 - LONG-TERM RECEIVABLES

Long-term receivables consist of initial franchise fees with extended terms and no stated interest rate. The Company has imputed a discount at 5.63%. At December 31, 2022 and 2021, the discounted balances of long-term receivables were \$4,529,500 and \$2,856,793. Based on the franchise agreements, the Company expects payments for the long-term receivables balance in 2024. The Company has evaluated long-term receivables and determined no allowance in relation to these balances are necessary.

NOTE 4 - INTANGIBLE ASSETS AND GOODWILL

Intangible assets are comprised of the following:

	Estimated	December 31,			
	Life		2022		2021
Franchise rights Less: accumulated amortization	20 Years	\$	215,000 139,750	\$	215,000 129,000
Definite-lived intangibles, net Trademarks	Indefinite		75,250 1,188,317		86,000 1,188,317
Intangible assets, net		\$	1,263,567	\$	1,274,317
Goodwill Less: accumulated amortization		\$	1,016,572 813,257	\$	1,016,572 711,599
Goodwill, net		\$	203,315	\$	304,973

Amortization expense related to the intangible assets and goodwill totaled \$112,407 for the years ended December 31, 2022 and 2021.

Estimated future amortization expense related to intangible assets and goodwill is as follows:

2023 2024	\$ 112,407 112,407
2025	 53,751
	\$ 278,565

NOTE 5 - LEASES

Operating Leases:

The Company leases its main facility from a third-party lender under a lease which expires in August 2023. As of December 31, 2022, the right-of-use asset related to the operating lease totaled \$242,449 and the lease liability totaled \$307,483. Operating lease expense totaled \$364,732 for 2022. The following table presents supplemental information pertaining to the operating lease as of and for the year ended December 31, 2022:

Operating cash flows from operating leases	\$ 451,071
ROU assets obtained in exchange for new operating lease liabilities	\$ 754,548
Weighted-average remaining lease term in years for operating leases	0.67
Weighted-average discount rate for operating leases	0.78%

The following table presents the maturities of the Company's operating lease liability and the present value discount at December 31, 2022:

2023	\$ 308,188
Less: present value discount	(705)
Total lease liability	\$ 307,483

NOTE 6 - PAYCHECK PROTECTION PROGRAM LOAN

On March 29, 2021, the Company received a second loan in the amount of \$840,370 under the Paycheck Protection Program ("PPP"), established as part of the Coronavirus Aid, Relief and Economic Security Act ("CARES Act"). The PPP loan's principal and accrued interest are forgivable to the extent that the proceeds are used for eligible purposes, subject to certain limitations. The loan proceeds were used for eligible purposes during the covered periods, and the loan was forgiven. Accordingly, the Company recognized \$840,370 as other income during the year ended December 31, 2021.

NOTE 7 - LINE OF CREDIT

In July 2020, the Company obtained a \$2,000,000 unsecured line of credit with a bank. Interest was charged at LIBOR plus 2.50% and was secured by substantially all assets of the Company. The line of credit was paid off in April of 2021, matured in September 2021 and was not renewed.

NOTE 8 - EMPLOYEE RETENTION CREDIT

The Company filed amended Form 941 payroll tax returns to apply for a refund of the Employee Retention Credit in the amount of \$1,660,604 for quarter four of 2020 and quarters one, two and three of 2021. Accordingly, the Company recognized \$1,660,604 as other income during the year ended December 31, 2021. During 2021, the Company received a refund for quarter four of 2020 and quarter two of 2021 in the amount of \$745,186. During 2022, the Company received a refund for quarter one of 2021 in the amount of \$403,230. The remaining refund of \$512,188 for quarter three of 2021 is reflected in current assets.

NOTE 9 - OTHER LIABILITY

As of December 31, 2021, the Company had an amount due under a Release Agreement in the amount of \$1,000,000. The portion due December 1, 2022, in the amount of \$500,000, is reflected in accounts payable at December 31, 2021 and the remaining \$500,000 is reflected in other liabilities. During 2022, the Company paid \$500,000. The remaining \$500,000 is due on December 1, 2023 and is included in accounts payable.

NOTE 10 - MEMBERS' EQUITY

In February 2018, the Company amended and restated their operating agreement, effective January 1, 2018. The amended agreement grants a 49% profit interest in United Soccer Leagues, LLC to the Papadakis Family Trust along with other certain privileges such as the ability to appoint managers. The granting of the profit interest is intended to be issued as defined by applicable revenue procedures issued under the Internal Revenue Code in connection with profit interests. In addition, the note payable - related party receives preferred returns, and shall be repaid prior to any distributions to the members unless approved by the majority members.

NOTE 11 - EMPLOYEE BENEFIT PLAN

The Company sponsors a Section 401(k) Profit Sharing Plan (the "Plan"). The Plan is available for employees who have completed 6 months of service and have attained age 18. The Company made contributions to the Plan totaling \$321,940 during the year ended December 31, 2022. The Company did not make contributions to the Plan during the year ended December 31, 2021. The Company moved the Plan from Securian to Fidelity during the year ended December 31, 2021. The Company reinstated the Company Safe Harbor contribution match effective January 1, 2022.

NOTE 12 - FRANCHISE AGREEMENTS

Information relating to franchise agreements and changes in the number of franchises are as follows:

	2022	2021
Franchises in operation at the beginning of the year	124	103
Franchise agreements signed, operations commenced	79	38
Franchise agreements terminated	14	17
Franchises in operation at the end of the year	189	124

As of December 31, 2022, there are twelve franchises with executed agreements that will not participate until after the 2023 season.

NOTE 13 - RELATED PARTY TRANSACTIONS

Note Payable - Related Party:

On January 1, 2017, USL and subsidiaries executed a secured promissory note payable to NuRock which is secured by substantially all assets. The note has a face amount of \$5,809,221 and arose in connection with the members' desire to distribute ownership capital plus a preferred return from the Company. The note bears interest at the greater of the current 10-year treasury rate plus one hundred thirty basis points, adjusted biannually, or 15% per annum. Interest is payable on or before the first day of each month. Principal payments can be made at the Company's discretion with any outstanding principal and interest balance due on January 1, 2047. USL has reduced the member capital contribution and recorded the note payable associated with this promissory note. Outstanding principal balance at December 31, 2022 and 2021, was \$5,761,518.

In 2022 and 2021, interest expense on this note payable amounted to \$864,180.

Management and Consulting Fee:

The Company is a party to a management and consulting agreement with NuRock which has a ten-year term that initially expired in October 2019 and provides for automatic five-year renewals. The agreement calls for a fee based on 3.5% of annual net revenues. The fee is only owed and payable in years in which the Company achieves positive net income. In 2022 and 2021, the Company recorded fees under this agreement of \$976,600 and \$785,076, respectively, and had a payable of \$25,075 outstanding as of December 31, 2021.

NOTE 14 - CONCENTRATIONS, COMMITMENTS AND CONTINGENCIES

Customer Concentrations:

Financial instruments which potentially subject the Company to concentrations of credit risk consist principally of accounts receivable. During the normal course of business, the Company extends credit to customers located primarily in the United States and Canada. The financial position, payment history and amount of performance security deposits of each customer have been considered in determining the allowance for doubtful accounts.

For the year ended December 31, 2022, three franchisees accounted for approximately 48% of the total revenues of the Company. At December 31, 2022, three franchisees accounted for approximately 92% of the outstanding receivable balance.

For the year ended December 31, 2021, three franchisees accounted for approximately 37% of the total revenues of the Company. At December 31, 2021, three franchisees accounted for approximately 67% of the outstanding receivable balance.

Cash Concentration:

The Company's cash balances held at financial institutions are insured by the Federal Deposit Insurance Corporation ("FDIC") up to certain limits. As of December 31, 2022, the Company had cash balances of \$11,725,027 held by financial institutions in excess of insured limits.

Litigation:

The Company is a party to litigation in the normal course of operations. Management does not believe that the impact of any outstanding legal matters will have a material adverse impact on the Company's results of operations or financial position.

Other Commitments:

During 2019, the Company entered into an agreement with a third party which requires the Company to pay \$170,000 annually for a platform fee and mobile application fee from 2020 through 2023.

NOTE 15 - SUBSEQUENT EVENT

In January 2023, the Company received the remaining Employee Retention Credit refund in the amount of \$512,188.





CERTIFIED PUBLIC ACCOUNTANTS AND CONSULTANTS

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES

TAMPA, FLORIDA

CONSOLIDATED FINANCIAL STATEMENTS

DECEMBER 31, 2021 AND 2020



UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES

TAMPA, FLORIDA

CONSOLIDATED FINANCIAL STATEMENTS

DECEMBER 31, 2021 AND 2020

CONTENTS

	PAGE
Independent Auditor's Report	1
Consolidated Balance Sheets	3
Consolidated Statements of Operations	4
Consolidated Statements of Members' Equity	5
Consolidated Statements of Cash Flows	6
Notes to Consolidated Financial Statements	7



INDEPENDENT AUDITOR'S REPORT

Board of Directors United Soccer Leagues, LLC and Subsidiaries Tampa, Florida

Opinion

We have audited the accompanying consolidated financial statements of United Soccer Leagues, LLC and subsidiaries (the "Company"), which comprise the consolidated balance sheets as of December 31, 2021 and 2020, and the related consolidated statements of operations, changes in members' equity, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of the Company as of December 31, 2021 and 2020, and the results of their operations and their cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are available to be issued.

-1-

Saltmarsh, Cleaveland & Gund

saltmarshcpa.com | Since 1944 | (800) 477-7458

Board of Directors United Soccer Leagues, LLC and Subsidiaries

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements, including omissions, are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Salfmansh Cleansland & Gund

Tampa, Florida March 2, 2022

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS DECEMBER 31, 2021 AND 2020

ASSETS

		2021	 2020
Current Assets:			
Cash and restricted cash	\$	12,996,096	\$ 9,517,137
Accounts receivable, net		6,800,317	5,014,524
Prepaid expenses		228,367	668,971
Employee Retention Credit receivable		915,418	
Total current assets		20,940,198	 15,200,632
Other Assets:			
Long-term receivables, net		2,856,793	3,005,847
Furniture and equipment, net		242,391	268,686
Intangible assets, net		1,274,317	1,285,067
Goodwill, net		304,973	406,630
Other assets		19,898	 19,398
Total other assets		4,698,372	 4,985,628
Total Assets	\$	25,638,570	\$ 20,186,260
LIABILITIES AND MEMBERS' EC	UIT	Y	
Current Liabilities:			
Borrowings under line of credit	\$	-	\$ 1,000,000
Accounts payable		735,893	138,172
Due to related party		25,075	284,925
Performance security deposits		1,863,714	1,866,217
Deferred revenue		4,437,647	4,741,879
Accrued liabilities		1,099,684	 1,006,555
Total current liabilities		8,162,013	 9,037,748
Long Term Liabilities:			
Related party loan, less current maturities		5,761,518	5,761,518
Other liability		500,000	-
Total long term liabilities		6,261,518	 5,761,518
Total liabilities		14,423,531	14,799,266
Members' Equity		11,215,039	 5,386,994
Total Liabilities and Members' Equity	\$	25,638,570	\$ 20,186,260

The accompanying notes are an integral part of these consolidated financial statements.

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS YEARS ENDED DECEMBER 31, 2021 AND 2020

	2021		2020
Revenue:			
Sales of initial franchises	\$ 9,141,250	\$	2,659,227
League participation fees	9,400,032		4,445,581
Other revenue	4,142,929		1,094,467
Total revenue	 22,684,211		8,199,275
Expenses:			
Payroll expense	6,995,218		5,424,105
Operating expenses	8,945,732		4,271,126
Broadcast and TV production	1,709,827		1,330,143
Legal and professional fees	565,065		614,778
Rent expense	424,803		407,787
Total expenses	 18,640,645		12,047,939
Income (Loss) From Operations	 4,043,566		(3,848,664)
Other Income (Expense):			
Interest expense	(870,363)		(865,430)
Interest income	170,105		305,550
Forgiveness of Paycheck Protection Program loan	840,370		861,900
Employee Retention Credit	1,660,604		-
Other income	 18,464	_	-
Total other income	 1,819,180		302,020
Net Income (Loss)	\$ 5,862,746	\$	(3,546,644)

The accompanying notes are an integral part of these consolidated financial statements.

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF MEMBERS' EQUITY YEARS ENDED DECEMBER 31, 2021 AND 2020

	 2021	 2020
Balance at Beginning of Year	\$ 5,386,994	\$ 11,525,779
Net income (loss)	5,862,746	(3,546,644)
Distributions to members	 (34,701)	 (2,592,141)
Balance at End of Year	\$ 11,215,039	\$ 5,386,994

UNITED SOCCER LEAGUES, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS YEARS ENDED DECEMBER 31, 2021 AND 2020

	2021		2020	
Cash Flows From Operating Activities:				
Net income (loss)	\$	5,862,746	\$	(3,546,644)
Adjustments to reconcile net income (loss) to net cash				
provided by operating activities:				
Depreciation and amortization		219,179		216,524
Bad debt expense		261,548		392,188
Loss on disposal of furniture and equipment		4,083		-
Imputed interest - long-term receivables		(170,105)		(305,543)
Changes in -				
Accounts receivable, net		(2,047,341)		2,451,116
Prepaid expenses		440,604		(353,193)
Other receivable		(915,418)		-
Long-term receivables, net		319,159		1,844,399
Other assets		(500)		77,500
Accounts payable		597,721		(95,238)
Due to related party		(259,850)		(552,063)
Performance security deposits		(2,503)		(115,076)
Deferred revenue		(304,232)		571,270
Accrued liabilities		93,129		(240,138)
Other liability		500,000		-
Net cash provided by operating activities		4,598,220		345,102
Cash Flows From Investing Activities:				
Purchases of furniture and equipment		(84,560)		(3,069)
Net cash used in investing activities		(84,560)		(3,069)
Cash Flows From Financing Activities:				
Borrowings under line of credit		-		1,000,000
Repayments on line of credit		(1,000,000)		-
Member distributions		(34,701)		(2,592,141)
Net cash used in financing activities		(1,034,701)		(1,592,141)
Net Increase (Decrease) in Cash and Restricted Cash		3,478,959		(1,250,108)
Cash and Restricted Cash at Beginning of Year		9,517,137		10,767,245
Cash and Restricted Cash at End of Year	\$	12,996,096	\$	9,517,137
Supplemental Disclosure of Cash Flow Information: Cash paid during the year for interest	\$	870,363	\$	865,430
Supplemental Disclosure of Noncash Investing and Financing Activities: Exchange of note receivable for franchise rights	\$		\$	5,380,496

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Nature of Business:

USISL, Inc., (the Corporate Predecessor) was incorporated on April 1, 1996. On August 27, 2009, NuRock Soccer Holdings, LLC ("NuRock") and Robert Hoskins (collectively "the Members") purchased all of the outstanding stock of the Corporate Predecessor and subsequently converted the Corporate Predecessor to United Soccer Leagues, LLC, ("USL"), a limited liability company. In February 2018, the Company amended and restated their operating agreement, effective January 1, 2018 amending the Member's profit and loss percentage interests to be 50.5% to NuRock, 0.5% to Robert Hoskins and 49% to Papadakis Family Trust, LLC. USL operates soccer leagues collectively known as United Soccer Leagues. USL sells and grants franchises to owners of soccer teams throughout the United States and Canada and manages league activities. USL is sanctioned by the United States Soccer Federation (the "Federation") and has been a member of the Federation for over twenty-five years.

The liability of the Members is limited to the amount of their invested capital and Members have no obligation to make additional capital contributions to USL. Net profits and losses are allocated among the Members based on the Members' respective profit and loss interests.

In January 2016, USL created a wholly owned subsidiary, RRFC Holdings, LLC. There was no activity until May 14, 2019 at which time the name was changed to ABQ Holdings, LLC ("ABQ"). ABQ was used to run the operations of a League Two team, Albuquerque Del Sol, for the 2019 season. There was no activity for the 2020 or 2021 season.

On January 1, 2017, USL created five Florida limited liability companies (USL Pro, LLC; USL Pro- 2, LLC; USL Productions, LLC; Super Y League, LLC; and Premier Development League, LLC, collectively the "Newcos") as wholly owned subsidiaries of USL. USL restructured by contributing certain assets and assigning certain liabilities to the Newcos as set forth in the Operating and Contribution agreements.

In March 2019, USL created a wholly owned subsidiary, USL Kids, LLC ("USL Kids"). USL Kids was originally created to run a new soccer youth program. This program has been discontinued but USL Kids is still in existence.

On September 23, 2021, USL created a wholly owned subsidiary, USL W League, LLC. On October 8, 2021, USL created two wholly owned subsidiaries, USL Academy, LLC and USL Super League, LLC.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Principles of Consolidation:

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America and include the accounts of USL and its wholly owned subsidiaries (collectively known as the "Company"). Intercompany balances and transactions have been eliminated in consolidation.

Estimates:

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Accounts Receivable:

Accounts receivable are carried at original invoice amount less estimated allowances for doubtful accounts based on a review of all outstanding amounts. Management determines the allowance for doubtful accounts by regularly evaluating individual customer receivables and considering a customer's financial condition and credit history, and current economic conditions. Accounts receivable are written off when deemed uncollectible. Recoveries of accounts receivable previously written off are recorded when received. The allowance for doubtful accounts was \$635,000 and \$600,000 as of December 31, 2021 and 2020, respectively.

Furniture and Equipment:

Furniture and equipment are stated at cost. Depreciation expense is computed using the straight-line method over the estimated useful lives of the related assets, which range from 2 to 7 years. Depreciation charged to operations for 2021 and 2020 was \$106,772 and \$104,117, respectively. The cost value of furniture and equipment and associated accumulated depreciation are as follows:

	 2021	 2020
Furniture and equipment Less: accumulated depreciation	\$ 750,677 508,286	\$ 690,623 421,937
Furniture and equipment, net	\$ 242,391	\$ 268,686

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Furniture and Equipment (Continued):

When assets are retired or otherwise disposed of, the cost and related accumulated depreciation are removed from the accounts, and any resulting gain or loss is reflected in income for the period. The cost of maintenance and repairs is charged to expense as incurred; significant renewals and betterments are capitalized.

Intangible Assets:

Intangible assets are recorded at cost and consist of trademarks, which have indefinite lives, and franchise agreements, which are amortized using the straight-line method over their estimated useful life of 20 years. The Company determines the useful lives of intangible assets after considering the specific facts and circumstances related to each intangible asset. Factors the Company considers when determining useful lives include the contractual term of any agreement related to the asset, the historical performance of the asset, the Company's long-term strategy for using the asset, any laws or other local regulations which could impact the useful life of the asset, and other economic factors, including competition and specific market conditions.

The Company tests intangible assets determined to have indefinite useful lives for impairment annually, or more frequently if events or circumstances indicate that assets might be impaired. For indefinite-lived intangible assets, other than goodwill, if the carrying amount exceeds the fair value, an impairment charge is recognized in an amount equal to that excess. The Company believes that there has been no impairment of indefinite-lived intangible assets in 2021 and 2020.

Goodwill:

Goodwill consists of the excess of the purchase price of the Corporate Predecessor over the fair value of the net tangible and intangible assets acquired in the purchase. Goodwill is amortized on a straight-line basis for 10 years and analyzed for impairment at the entity level if a triggering event occurs.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Revenue Recognition:

The Company's revenues consist of initial franchise fees, league participation fee, and other revenue such as special events, hotel rebates, sponsorships, merchandise sales, and miscellaneous revenue.

Accounting Standards Update No. 2014-09, *Revenue from Contracts with Customers (Topic 606)* ("ASC 606") provides that revenues are to be recognized when control of promised goods or services is transferred to a customer in an amount that reflects the consideration expected to be received for those goods or services. ASC 606 does not impact the Company's recognition of any of its revenue sources. Management has determined that all performance obligations related to the initial fees from a franchise sale are met on the date the agreement is signed. Accordingly, revenue related to the sale of initial franchises is recognized as income at a point in time. League participation fees are recognized as income during the year in which the season is played. Revenue from hotel rebates is recognized when the related teams' stays have been completed. Revenue from special events and sponsorships is recognized upon completion of the associated event. Merchandise sales are recorded when the merchandise is delivered to the customer.

Deferred revenue is recorded for transactions where cash has been received prior to the recognition of revenue. Deferred revenue included in the consolidated balance sheets relates to team pre-payments received in advance, and franchise fees received prior to all conditions of the franchise agreement being satisfied.

Income Taxes:

Under provisions of the Internal Revenue Code, limited liability companies that are treated as partnerships are not subject to income taxes, and any income or loss realized is taxed to the individual Members. Accordingly, no provision for income taxes is reflected in these financial statements. Taxable income is passed through to and reportable by the Members.

The Company evaluates its tax positions for uncertainty on an annual basis and believes it has no uncertain tax positions as of December 31, 2021 and 2020.

Upcoming Accounting Pronouncements:

In February 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update No. 2016-02, Leases (Topic 842) ("ASU 2016-02"). The guidance in this topic supersedes the requirements in Accounting Standards Codification Topic 840, Leases. The update will require business entities to recognize lease assets and liabilities on the balance sheet and to disclose key information about leasing arrangements. A lessee would recognize a liability to make lease payments and a right-of-use asset representing its right to use the leased asset for the lease term. ASU 2016-02 is effective for fiscal years beginning after December 15, 2021. Management is currently assessing the impact of ASU 2016-02 on the Company's financial statements.

NOTE 1 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Subsequent Events:

The Company has evaluated subsequent events through March 2, 2022, the date which the consolidated financial statements were available to be issued.

NOTE 2 - CASH AND RESTRICTED CASH

For the majority of the franchisees, the Company requires either a letter of credit or a cash deposit with the Company for a specified amount to cover any required future fees due from the franchisees and not paid to the Company as provided for in the franchise agreement. The cash deposits are recorded as a performance security deposit liability on the balance sheets with balances of \$1,863,714 and \$1,866,217 as of December 31, 2021 and 2020, respectively. The cash associated with the performance security deposit liabilities is maintained in separate restricted cash accounts.

NOTE 3 - LONG-TERM RECEIVABLES

Long-term receivables consist of initial franchise fees with extended terms and no stated interest rate. The Company has imputed a discount at 5.63%. At December 31, 2021 and 2020, the discounted balances of long-term receivables were \$2,856,793 and \$3,005,847. The Company has evaluated long-term receivables and determined no allowance in relation to these balances are necessary.

The table below summarizes the expected payments based on the franchise agreements:

2023	\$ 2,873,666
2024	 12,000
	 2,885,666
Less: discount on long-term receivables	28,873
	\$ 2,856,793

NOTE 4 - INTANGIBLE ASSETS AND GOODWILL

Intangible assets are comprised of the following:

	Estimated	Decen	nber 3	1,
	Life	 2021		2020
Franchise rights Less: accumulated amortization	20 Years	\$ 215,000 129,000	\$	215,000 118,250
Definite-lived intangibles, net		 86,000		96,750
Trademarks	Indefinite	 1,188,317		1,188,317
Intangible assets, net		\$ 1,274,317	\$	1,285,067
Goodwill		\$ 1,016,572	\$	1,016,572
Less: accumulated amortization		 711,599		609,942
Goodwill, net		\$ 304,973	\$	406,630

Amortization expense related to the intangible assets and goodwill totaled \$112,407 for the years ended December 31, 2021 and 2020.

Estimated future amortization expense related to intangible assets and goodwill is as follows:

\$ 112,407
112,407
112,407
 53,752
\$ 390,973

NOTE 5 - LINE OF CREDIT

During the year ended December 31, 2020, the Company obtained a \$2,000,000 unsecured line of credit with a bank. Interest is charged at LIBOR plus 2.50% and was secured by substantially all assets of the Company. The outstanding balance on the line of credit was \$1,000,000 as of December 31, 2020. The line of credit was paid off in April of 2021, matured in September 2021 and was not renewed.

NOTE 6 - PAYCHECK PROTECTION PROGRAM LOAN

On May 1, 2020, the Company received loan proceeds in the amount of \$861,900 under the Paycheck Protection Program ("PPP"), established as part of the Coronavirus Aid, Relief and Economic Security Act ("CARES Act"). The Company received a second PPP loan in the amount of \$840,370 on March 29, 2021. The PPP loan's principal and accrued interest are forgivable to the extent that the proceeds are used for eligible purposes, subject to certain limitations. The loan proceeds were used for eligible purposes during the covered periods, and both loans were forgiven. Accordingly, the Company recognized \$840,370 and \$861,900 as other income during the years ended December 31, 2021, and 2020, respectively.

NOTE 7 - EMPLOYEE RETENTION CREDIT

The Company filed amended Form 941 payroll tax returns to apply for a refund of the Employee Retention Credit for quarter four of 2020 and quarters one, two and three of 2021. The Company received a refund for quarter four of 2020 and quarter two of 2021 in the amount of \$745,186. An additional refund is expected in the amount of \$915,418 for quarter one and three of 2021. Accordingly, the Company recognized \$1,660,604 as other income during the year ended December 31, 2021.

NOTE 8 - OTHER LIABILITY

As of December 31, 2021, the Company has an amount due under a Release Agreement in the amount of \$1,000,000. A payment of \$500,000 is due on December 1, 2022 and is included in accounts payable. The remaining \$500,000 is due on December 1, 2023 and is included in other liability at December 31, 2021.

NOTE 9 - EQUITY

In February 2018, the Company amended and restated their operating agreement, effective January 1, 2018. The amended agreement grants a 49% profit interest in United Soccer Leagues, LLC to the Papadakis Family Trust along with other certain privileges such as the ability to appoint managers. The granting of the profit interest is intended to be issued as defined by applicable revenue procedures issued under the Internal Revenue Code in connection with profit interests. In addition, the note payable - related party receives preferred returns, and shall be repaid prior to any distributions to the members unless approved by the majority members.

NOTE 10 - EMPLOYEE BENEFIT PLAN

The Company sponsors a Section 401(k) Profit Sharing Plan (the "Plan"). The Plan is available for employees who have completed 6 months of service and have attained age 20½. The Company made contributions to the Plan totaling \$97,516 during the year ended December 31, 2020. The Company did not make contributions to the Plan during the year ended December 31, 2021. The Company moved the Plan from Securian to Fidelity during the year ended December 31, 2021. The Company reinstated the Company Safe Harbor contribution match effective January 1, 2022.

NOTE 11 - FRANCHISE AGREEMENTS

Information relating to franchise agreements and changes in the number of franchises are as follows:

	2021	2020
Franchises in operation at the beginning of the year	102	111
Franchise agreements signed, operations commenced	38	2
Franchise agreements terminated	17	11
Franchises in operation at the end of the year	123	102

As of December 31, 2021, there are three franchises with executed agreements that will not participate until after the 2022 season.

NOTE 12 - RELATED PARTY TRANSACTIONS

Note Payable - Related Party:

On January 1, 2017, USL and subsidiaries executed a secured promissory note payable to NuRock which is secured by substantially all assets. The note has a face amount of \$5,809,221 and arose in connection with the members' desire to distribute ownership capital plus a preferred return from the Company. The note bears interest at the greater of the current 10-year treasury rate plus one hundred thirty basis points, adjusted biannually, or 15% per annum. Interest is payable on or before the first day of each month. Principal payments can be made at the Company's discretion with any outstanding principal and interest balance due on January 1, 2047. USL has reduced the member capital contribution and recorded the note payable associated with this promissory note. Outstanding principal balance at December 31, 2021 and 2020, was \$5,761,518.

In 2021 and 2020, interest expense totaled \$864,180.

NOTE 12 - RELATED PARTY TRANSACTIONS (Continued)

Management and Consulting Fee:

The Company is a party to a management and consulting agreement with NuRock which has a ten-year term that initially expired in October 2019 and provides for automatic five-year renewals. The agreement calls for a fee based on 3.5% of annual revenues. The fee is only owed and payable in years in which the Company achieves positive net income. In 2021 and 2020, the Company recorded fees under this agreement of \$785,076 and \$284,928, respectively, and had a payable of \$25,075 and \$284,925, respectively, outstanding as of December 31, 2021 and 2020.

NOTE 13 - CONCENTRATIONS, COMMITMENTS AND CONTINGENCIES

Customer Concentrations:

Financial instruments which potentially subject the Company to concentrations of credit risk consist principally of accounts receivable. During the normal course of business, the Company extends credit to customers located primarily in the United States and Canada. The financial position, payment history and amount of performance security deposits of each customer have been considered in determining the allowance for doubtful accounts.

For the year ended December 31, 2021, three franchisees accounted for approximately 37% of the total revenues of the Company. At December 31, 2021, three franchisees accounted for approximately 67% of the outstanding receivable balance.

For the year ended December 31, 2020, one franchisee accounted for approximately 12% of the total revenues of the Company. At December 31, 2020, three franchisees accounted for approximately 92% of the outstanding receivable balance.

Cash Concentration:

The Company's cash balances held at financial institutions are insured by the Federal Deposit Insurance Corporation ("FDIC") up to certain limits. As of December 31, 2021, the Company had cash balances of \$13,563,899 held by financial institutions in excess of insured limits.

Litigation:

The Company is a party to litigation in the normal course of operations. Management does not believe that the impact of any outstanding legal matters will have a material adverse impact on the Company's results of operations or financial position.

NOTE 13 - CONCENTRATIONS, COMMITMENTS AND CONTINGENCIES (Continued)

Operating Leases:

The Company leases office space and certain equipment under operating leases. During 2018, the Company entered into a modification of their lease agreement to include additional floor space in their current location as well as provide for an extension of their existing lease through September 2023. The new lease agreement provides for annual rental increases and for the payment of taxes, insurance, utilities, repairs and maintenance expenses. As rent is recorded on the straight-line basis, the Company had \$151,373 and \$188,143 in deferred rent for the years ended December 31, 2021 and 2020, respectively, which is included in accrued liabilities on the accompanying consolidated balance sheets.

Future minimum lease payments under the operating leases are as follows:

2022 2023		\$ 489,412 334,386
	-	\$ 823,798

Rent expense totaled \$424,803 and \$407,787 for the years ended December 31, 2021 and 2020, respectively.

Other Commitments:

During 2019, the Company entered into an agreement with a third party which requires the Company to pay \$170,000 annually for a platform fee and mobile application fee from 2020 through 2023.

NOTE 11 - COMMITMENTS AND CONTINGENCIES (Continued)

Operating Leases:

The Company leases office space and certain equipment under operating leases. During 2018, the Company entered into a modification of their lease agreement to include additional floor space in their current location as well as provide for an extension of their existing lease through September 2023. The new lease agreement provides for annual rental increases and for the payment of taxes, insurance, utilities, repairs and maintenance expenses. As rent is recorded on the straight-line basis, the Company had \$188,143 and \$201,613 in deferred rent for the years ended December 31, 2020 and 2019, respectively, which is included in accrued liabilities on the accompanying consolidated balance sheets.

Future minimum lease payments under the operating leases are as follows:

2021	\$ 405,738
2022	451,071
2023	 308,188
	\$ 1,164,997

Rent expense totaled \$407,787 and \$423,388 for the years ended December 31, 2020 and 2019, respectively.

Other Commitments:

During 2019, the Company entered into an agreement with a third party which requires the Company to pay \$170,000 annually for a platform fee and mobile application fee from 2020 through 2023.

EXHIBIT F-2 – GUARANTY

[See attachment]

GUARANTEE OF PERFORMANCE

For value received, United Soccer Leagues, LLC, a Florida limited liability company (the "Guarantor"), located at 1715 N. Westshore Blvd., Suite 825, Tampa, Florida, 33607 absolutely and unconditionally guarantees to assume the duties and obligations of USL Super League, LLC, a Florida limited liability company located at 1715 N. Westshore Blvd., Suite 825, Tampa, Florida, 33607 (the "<u>Franchisor</u>"), under its franchise registration in each state where the franchise is registered, and under its Franchise Agreement identified in its 2023 Franchise Disclosure Document, as it may be amended, and as that Franchise Agreement may be entered into with franchisees and amended, modified or extended from time to time. This guarantee continues until all such obligations of the Franchisor under its franchise registrations and the Franchise Agreement are satisfied or until the liability of Franchisor to its franchise Agreement has been completely discharged, whichever first occurs. The Guarantor is not discharged from liability if a claim by a franchisee against the Franchisor remains outstanding. Notice of acceptance is waived. The Guarantor does not waive receipt of notice of default on the part of the Franchisor. This guarantee is binding on the Guarantor and its successors and assigns.

The Guarantor signs this guarantee in Tampa, Florida, as of the 15th day of February, 2023.

Guarantor: United Soccer Leagues, LLC

By: /s/ Alec Papadakis

Name: Alec Papadakis

Title: Chief Executive Officer

EXHIBIT G – STATE ADDENDA TO FRANCHISE DISCLOSURE DOCUMENT

STATE ADDENDA TO FRANCHISE DISCLOSURE DOCUMENT

Exhibit G.1	California
Exhibit G.2	Illinois
Exhibit G.3	Minnesota
Exhibit G.4	New York
Exhibit G.5	North Dakota
Exhibit G.6	Rhode Island
Exhibit G.7	Washington

EXHIBIT G.1 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF CALIFORNIA

In recognition of the requirements of the California Franchise Investment Law, California Corporations Code §§ 31000 - 31516, and the California Franchise Relations Act, California Business and Professions Code §§ 20000 - 20043, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of California shall be amended to include the following:

1. California Corporations Code § 31125 requires USL Super League, LLC to give you a disclosure document, in a form containing the information that the Commissioner of Corporations of the California Department of Corporations may by rule or order require, prior to a solicitation or a proposed material modification of an existing franchise.

2. The California Franchise Investment Law requires that a copy of all proposed agreements relating to the sale of the franchise to be delivered together with the disclosure document.

3. Item 3, "Litigation," shall be amended by the addition of the following language:

Neither USL Super League, LLC d/b/a USL Super League, nor any person or franchise broker identified in Item 2 of the Franchise Disclosure Document, is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Act of 1934, 15 U.S.C.A. 78a *et seq.*, suspending or expelling such persons from membership in this association or exchange.

4. Item 17, "Renewal, Termination, Transfer, and Dispute Resolution," shall be amended by the addition of the following language:

The regulations of the California Department of Corporations require that the following information concerning provisions of the franchise agreement be disclosed to you:

California Business and Professions Code sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or nonrenewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control.

The franchise agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law, 11 U.S.C.A. §§ 101, *et seq*.

The franchise agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

The franchise agreement contains a covenant not to compete which extends beyond the term of the franchise agreement. This provision may not be enforceable under California law.

The franchise agreement requires the application of the laws of Florida. This provision may be unenforceable under California law.

The franchise agreement requires binding arbitration. The arbitration will occur in Tampa,

Florida. This provision may not be enforceable under California law.

The franchise agreement requires you to sign a general release of claims upon renewal or transfer of the franchise agreement. California Corporations Code § 31512 provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of that law or any rule or order is void.

5. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

6. OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF CORPORATIONS. ANY COMPLAINTS CONCERNING THE CONTENT OF THE WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF BUSINESS OVERSIGHT at www.dbo.ca.gov.

EXHIBIT G.2 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF ILLINOIS

In recognition of the requirements of the Illinois Franchise Disclosure Act of 1987, 815 Illinois Compiled Statutes §§ 705/1 - 705/44, the Franchise Disclosure Document for USL Super Legaue, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of Illinois shall be amended to include the following:

1. The "Summary" section of Item 17(v), entitled "Choice of forum," shall be amended by the addition of the following language:

However, any provision in the franchise agreement that designates jurisdiction or venue in a forum outside of the State of Illinois is void under Section 4 of the current Illinois Franchise Disclosure Act of 1987 (as amended), although the franchise agreement may provide for arbitration in a forum outside of the State of Illinois.

2. The "Summary" section of Item 17(w), entitled "Choice of law," shall be amended by the addition of the following language:

However, except for federal law, Illinois law applies if the jurisdiction requirements of the Illinois Franchise Disclosure Act of 1987 (as amended) are met.

3. Each provision of this Addendum to the Franchise Disclosure Document shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Illinois Franchise Disclosure Act of 1987 are met independently, without reference to this Addendum to the Franchise Disclosure Document, and only to the extent such provision is a then valid requirement of the statute.

EXHIBIT G.3 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF MINNESOTA

In recognition of the requirements of the Minnesota Franchise Act, Minnesota Statutes §§ 80C.01 - 80C.30, and of the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce, Minnesota Rules §§ 2860.0100 - 2860.9930, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of Minnesota shall be amended to include the following:

1. The Risk Factors set forth on the State Cover Page shall be amended by the addition of the following paragraph:

MINNESOTA STATUTES § 80C.21 AND MINNESOTA RULES § 2860.4400J PROHIBIT US FROM REQUIRING LITIGATION TO BE CONDUCTED OUTSIDE OF THE STATE OF MINNESOTA. IN ADDITION, NOTHING IN THE FRANCHISE DISCLOSURE DOCUMENT OR FRANCHISE AGREEMENT CAN ABROGATE OR REDUCE ANY OF YOUR RIGHTS AS PROVIDED FOR IN THE MINNESOTA FRANCHISE ACT, OR YOUR RIGHTS TO ANY PROCEDURE, FORUM, OR REMEDIES PROVIDED FOR BY THE LAWS OF THE JURISDICTION.

MINNESOTA RULES § 2860.4400J PROHIBITS US FROM REQUIRING YOU TO CONSENT TO LIQUIDATED DAMAGES.

2. Item 5, "Initial Fees," shall be amended by the addition of the following language:

All payment of expansion fees shall be deferred until the franchise opens.

3. Item 6, "Other Fees," shall be amended by the addition of the following language:

Minnesota Rules § 2860.4400J prohibits us from requiring you to consent to liquidated damages.

4. Item 7, "Estimated Initial Investment," shall be amended by the addition of the following language:

All payment of expansion fees shall be deferred until the franchise opens.

5. Item 13, "Trademarks," shall be amended by the addition of the following language:

We will protect the franchisee's right to use our trademarks, service marks, trade names, logotypes or other commercial symbols or will indemnify the franchisee from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the name.

6. Item 17, "Renewal, Termination, Transfer, and Dispute Resolution," shall be amended by the addition of the following paragraphs:

With respect to franchisees governed by Minnesota law, we will comply with Minnesota Statutes § 80C.14, Subsections 3, 4, and 5 which require, except in certain specified cases, that a franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice of

non-renewal of the franchise agreement, and that consent to the transfer of the franchise not be unreasonably withheld.

Pursuant to Minnesota Rules § 2860.4400D, any general release of claims that you or a transferor may have against us and our subsidiaries and affiliates, and our respective officers, directors, agents and employees, including without limitation, claims arising under federal, state, and local laws and regulations shall exclude claims you or a transferor may have under the Minnesota Franchise Act, and the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce.

Minnesota Statutes § 80C.21 and Minnesota Rules § 2860.4400J prohibit us from requiring litigation to be conducted outside of the State of Minnesota. In addition, nothing in the Franchise Disclosure Document or franchise agreement can abrogate or reduce any of your rights as provided for in the Minnesota Franchise Act, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.

7. Each provision of this Addendum to the Franchise Disclosure Document shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Minnesota Franchise Act or the Franchise Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce are met independently, without reference to this Addendum to the Franchise Disclosure Document, and only to the extent such provision is a then valid requirement of the statute.

IN WITNESS WHEREOF, the undersigned acknowledges receipt of the foregoing Addendum to the USL Super League, LLC Franchise Disclosure Document for the State of Minnesota.

FRANCHISEE:

(Name of Entity)

By:_____ Signature

Name

Title

Date

EXHIBIT G.4 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF NEW YORK

In recognition of the requirements of Article 33 of the General Business Law of the State of New York, §§ 680 - 695, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of New York shall be amended to include the following:

1. The Risk Factors set forth on the State Cover Page shall be amended by the addition of the following paragraphs:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT B OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY A STATE DOES NOT MEAN THAT THE STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY.

THIS FRANCHISE DISCLOSURE DOCUMENT IS PROVIDED FOR YOUR OWN PROTECTION AND CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THIS FRANCHISE DISCLOSURE DOCUMENT AND ALL CONTRACTS OR AGREEMENTS SHOULD BE READ CAREFULLY IN THEIR ENTIRETY FOR AN UNDERSTANDING OF ALL RIGHTS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

ALTHOUGH THESE FRANCHISES HAVE BEEN ACCEPTED FOR FILING, SUCH FILING UNDER ARTICLE 33 OF THE GENERAL BUSINESS LAW OF THE STATE OF NEW YORK DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE NEW YORK STATE DEPARTMENT OF LAW THAT THE INFORMATION PROVIDED HEREIN IS TRUE. THE DEPARTMENT'S REVIEW DID NOT INCLUDE A DETAILED EXAMINATION OF THE MATERIALS SUBMITTED. A FALSE, INCOMPLETE, INACCURATE OR MISLEADING STATEMENT MAY CONSTITUTE A VIOLATION OF BOTH FEDERAL AND STATE LAW, AND SHOULD BE REPORTED TO BOTH THE FEDERAL TRADE COMMISSION, WASHINGTON D.C. 20580 AND THE NEW YORK STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION AND SECURITIES, 120 BROADWAY, NEW YORK, NEW YORK 10271.

WE MAY NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, WE CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON YOU TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. Item 3, "Litigation," shall be amended by the addition of the following language:

Neither USL Super League, LLC, its predecessor, a person identified in Item 2, nor any affiliate offering franchises under USL Super League, LLC's principal trademark:

(a) has an administrative, criminal or civil action pending against that person alleging: a felony; a violation of a franchise, antitrust or securities law; fraud, embezzlement, fraudulent conversion, misappropriation of property; unfair or deceptive practices or comparable civil or misdemeanor allegations.

(b) has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the ten-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud or securities law; fraud, embezzlement, fraudulent conversion or misappropriation of property, or unfair or deceptive practices or comparable allegations.

(c) is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a federal, State or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. Item 4, "Bankruptcy," is deleted in its entirety and replaced with the following:

Neither USL Super League, LLC, its affiliates, its predecessor, officers or general partner, during the 10-year period immediately before the date of the Franchise Disclosure Document: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after the officer or general partner of USL Super League, LLC held this position in the company or partnership.

4. Item 7, "Estimated Initial Investment," shall be amended by the addition of the following language:

THERE ARE NO OTHER DIRECT OR INDIRECT PAYMENTS TO USL SUPER LEAGUE, LLC IN CONJUNCTION WITH THE PURCHASE OF THE FRANCHISE.

5. The "Summary" section of Item 17(d), entitled "Termination by franchisee," shall be amended by the addition of the following language:

You may terminate your franchise agreement on any grounds available by law.

6. The "Summary" section of Item 17(j), entitled "Assignment of contract by franchisor," shall be amended by the addition of the following language:

However, no assignment will be granted except to an assignee who in our good faith judgment is willing and able to assume our obligations.

7. The "Summary" sections of Item 17(c), entitled "Requirements for franchisee to renew or extend"

Item 17(m), entitled "Conditions for franchisor approval of transfer," shall each be amended by the addition of the following language:

The transferor must execute a general release, in a form prescribed by us, of any and all claims against USL Super League, LLC and our subsidiaries and affiliates, and our respective officers, directors, agents and employees; provided, however, that all rights enjoyed by you and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force, it being the intent of this provision that the non-waiver provisions of §§ 687.4 and 687.5 of Article 33 of the General Business Law of the State of New York be satisfied.

8. The "Summary" section of Item 17(w), entitled "Choice of law," shall be amended by the addition of the following language:

The foregoing choice of law should not be considered a waiver of any right conferred upon us or you by Article 33 of the General Business Law of the State of New York.

9. Each provision of this Addendum to the Franchise Disclosure Document shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of Article 33 of the General Business Law of the State of New York are met independently, without reference to this Addendum to the Franchise Disclosure Document, and only to the extent such provision is a then valid requirement of the statute.

10. USL SUPER LEAGUE, LLC REPRESENTS THAT THIS FRANCHISE DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

EXHIBIT G.5 ADDENDUM TO USL SUPER LEGAUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF NORTH DAKOTA

In recognition of the requirements of the North Dakota Century Code and ND Franchise Investment Law Chapter 51-19, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of North Dakota shall be amended to include the following:

1. Item 17(c) of the Franchise Disclosure Document and Section 3 of the Franchise Agreement requires the franchise to sign a general release upon renewal of the franchise agreement. The Commissioner has determined this to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Delete this provision each place it appears in the agreement.

2. Item 17(i) of the Franchise Disclosure Document and Section 12.8 of the Franchise Agreement requires the franchisee to consent to termination or liquidated damages. The Commissioner has determined this to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Delete this provision each place it appears in the agreement.

3. Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota.

4. Item 17(r) of the Franchise Disclosure Document and Section 12.22 of the Franchise Agreement discloses the existence of certain covenants restricting competition to which franchises must agree. Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota.

5. Item 17(u) of the Franchise Disclosure Document and Section 13 of the Franchise Agreement provides that the franchisee must agree to the arbitration or mediation of disputes, such arbitration or mediation to be held in Florida. The Commissioner has determined that franchise agreements, which provide that parties agree to the arbitration or mediation of disputes at a location that is remote from the site of the franchisee's business, are unfair, unjust, or inequitable within the intent of Section 51-19-09 if the North Dakota Investment Law. Delete this provision each place it appears in the agreement. Any litigation, arbitration or mediation under the agreement shall be conducted in North Dakota.

6. Item 17(v) of the Franchise Disclosure Document and Section 13.2 of the Franchise Agreement provides that franchisees must consent to the jurisdiction of courts in Florida. The Commissioner has held that requiring franchisees to consent to the jurisdiction of courts outside of North Dakota is unfair, unjust or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. North Dakota law will govern this agreement. Delete this provision each place it appears in the agreement. Any litigation, arbitration or mediation under the agreement shall be conducted in North Dakota.

7. Item 17(w) of the Franchise Disclosure Document and Section 13.1 of the Franchise Agreement provides that the agreement shall be construed according to the laws of the State of Florida. Apart from civil liability as set forth in Section 51-19-12 N.D.C.C., which is limited to violations of the North Dakota Franchise Investment Law (registration and fraud) the liability of the franchisor to a franchise is based largely on contract law. Despite the fact that those provisions are not contained in the franchise investment law, those provisions contain substantive rights intended to be afforded to North Dakota residents, and is

unfair to franchise investors to require them to waive their rights under North Dakota Law.

The Commissioner has held that franchise agreements, which specify that they are to be governed by the laws of a state other than North Dakota, are unfair, unjust or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. If the agreement requires a state law other that North Dakota, the requirement is void. North Dakota law will govern the agreement.

8. Section 13.7 of the Franchise Agreement requires the franchise to consent to a waiver of trial by jury. The Commissioner has determined this to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Delete this provision each place it appears in the agreement.

9. Section 13.4 of the Franchise Agreement requires the franchisee to consent to a waiver of exemplary and punitive damages. The Commissioner has determined this to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Delete this provision each place it appears in the agreement.

10. Section 13 of the Franchise Agreement requires the franchisee to consent to a limitation of claims within one year. The Commissioner has determined this to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. Limitations of claims will be governed by North Dakota Law.

11. It is the policy of the USL Super League, LLC to hold all franchise fees received in an account named Franchise Deposit Liability Account, and to recognize fees held in the account as revenue only when the franchisee commences operations. Also, administrative fees paid by said franchisees are posted in a deferred revenue account until such time as the revenue is earned.

Franchisor: USL Super League, LLC d/b/a USL Super League

Signature of authorized officer

Title of authorized officer

Date

Franchisee:

Authorized signatory

Name of Franchisee

Date

EXHIBIT G.6 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF RHODE ISLAND

In recognition of the requirements of the Rhode Island Franchise Investment Act, Rhode Island Code §§ 19-28.1-1 - 19-28.1-34, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of Rhode Island shall be amended to include the following:

1. Item 17, "Renewal, Termination, Transfer, and Dispute Resolution," shall be amended by the addition of the following language:

§ 19-28.1-14 of the Rhode Island Franchise Investment Act provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

2. Each provision of this Addendum to the Franchise Disclosure Document shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Rhode Island Franchise Investment Act are met independently, without reference to this Addendum to the Franchise Disclosure Document, and only to the extent such provision is a then valid requirement of the statute.

EXHIBIT G.7 ADDENDUM TO USL SUPER LEAGUE, LLC FRANCHISE DISCLOSURE DOCUMENT FOR THE STATE OF WASHINGTON

The State of Washington has an act, the Washington Franchise Investment Protection Act, Revised Code of Washington §§ 19.100.010 - 19.100.940, which may supersede the franchise agreement in your relationship with us including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with us including the areas of termination.

In recognition of the requirements of the Washington Franchise Investment Protection Act, the Franchise Disclosure Document for USL Super League, LLC d/b/a USL Super League in connection with the offer and sale of franchises for use in the State of Washington shall be amended to include the following:

1. Item 17, "Renewal, Termination, Transfer, and Dispute Resolution," shall be amended by the addition of the following paragraphs:

In any arbitration involving a franchise purchased in Washington, the arbitration site shall be either in the State of Washington or in a place as mutually agreed upon at the time of the arbitration, or as determined by the arbitrator.

If any of the provisions in the Franchise Disclosure Document or franchise agreement are inconsistent with the relationship provisions of Revised Code of Washington § 19.100.180 or other requirements of the Washington Franchise Investment Protection Act, the provisions of the Washington Franchise Investment Protection Act will prevail over the inconsistent provisions of the Franchise Disclosure Document and franchise agreement with regard to any franchise sold in Washington.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act shall prevail.

Any release or waiver of rights executed by you will not include rights under the Washington Franchise Investment Protection Act except if you execute the release or waiver as part of a negotiated settlement after your franchise agreement is in effect and where we are both represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act and/or rights or remedies under the Washington Franchise Investment Protection Act, such as a right to a jury trial, may not be enforceable.

We may collect the transfer fees if they reflect our reasonable estimated or actual costs in effecting your transfer.

2. Each provision of this Addendum to the Franchise Disclosure Document shall be effective only to the extent, with respect to such provision, the jurisdictional requirements of the Washington Franchise Investment Protection Act are met independently, without reference to this Addendum to the Franchise Disclosure Document, and only to the extent such provision is a then valid requirement of the statute.

EXHIBIT H – MANUALS (TABLES OF CONTENTS)

[see attachment]



Broadcast Manual

Draft *accurate at time of filing and subject to change*



A. GENERAL1
B. MINIMUM PRODUCTION STANDARDS
C. TRANSMISSION
D. DIGITAL BROADCASTS
E. LINEAR BROADCASTS
1. Supremacy of National Linear Broadcast Agreements
2. Production of National Linear Broadcasts
3. Broadcast Territory
F. CLUB COMMERCIAL INVENTORY
1. Games Of The Week
2. Pre-Game and Post-Game Shows
G. LOCAL BROADCAST PARTNER PROMOTION
1. Local Broadcast Partner Bug
2. Station or Upcoming Program Promotion
H. SATELLITE OR DIGITAL RADIO
I. LEAGUE APPROVAL
J. GAME OF THE WEEK PACKAGE
K. MUSIC AND GRAPHICS
1. Music
2. Graphics
L. STATISTICS / OPTA
1. API to Ross Xpression
M. RUN-OF-SHOW/HALF-TIME
N. PLAYOFFS AND CUP FINAL
O. ARCHIVE
P. PACKAGED CONTENT
Q. FINES AND ENFORCEMENT
R. CONTACT INFORMATION
S. APPENDICES



APPENDICES

- A. Minimum Production Standards
- B. Game of the Week Schedule
- C. Run-of-Show / Broadcast Clock
- D. Commercial Guidelines
- E. Schedule of Fines
- F. VISTA Game Procedure
- G. Producing Soccer on Television
- H. Directing Soccer on Television
- I. Announcing USL
- J. Digital Archiving
- K. Public Relations Procedures
- L. Broadcast Agreement Template
- M. Broadcast Addendum





Operations Manual

Draft *accurate at time of filing and subject to change*



TABLE OF CONTENTS

DEFINITIONS		
Section I: Membership		
1. Fran	chise Governance	
А.	Governance	
B.	Authority of the League	
C.	Territories	
D.	Reference to Franchise Agreement	
E.	Restrictions on Ownership	13
F.	Restrictions on Agents	
G.	Winter Summit	13
Н.	Regional/League Meetings	14
I.	Rules of Order	14
J.	Failure to Attend	14
К.	Conditions for Termination of Franchise	14
L.	Withdrawal or Termination of a Team	14
М.	League Review, Appeals & Discipline (RAD)	14
2. Fran	chise Requirements	
А.	Liability & Insurance	17
В.	Letter of Credit	17
C.	League Administrative Fee	
D.	Fines	
E.	Team Names & Logos	
F.	Trademarks	
G.	Senior Management	19
Н.	Telephone	19
I.	E-mail and Website	19
J.	Stadium Contract	19
К.	League Sponsorship Obligations	19
L.	Compliance	19
М.	Exemptions to League Regulations & Standards	20
N.	Financial Statements	20
О.	Confidentiality Regarding Union Negotiations	20
Р.	Tickets.com	20
3. Gove	erning Bodies and Committees	
А.	President	20
B.	League Directors	21
C.	Executive Committee	21
D.	Board of Governors	



E.	Competition Strategy Committee	1
F.	Standing Committees	1
Section	II: Discipline & Conduct	2
1.	League Disciplinary Power	22
2.	Authorities	22
A	. The League Disciplinary Committee	2
B.	. Independent Disciplinary Panel	2
3.	Disciplinary Process	23
4.	Suspensions	23
A	. The League's Discretionary Power on Suspensions	3
B.	. Serving Suspensions During the Regular Season	3
C.	. Exhibition Game Send-off24	4
D	. Final Regular Season Game Send-off24	4
E.	Suspension Parameters & Restrictions	4
F.	Extending a Suspension2	5
G	. Suspension during the Playoffs	5
H	. Cross League and Competition Suspensions	5
I.	USSF Discipline for Serious Incidents	5
5.	Penalty Points and Accumulation	25
A	. League Disciplinary Point System	5
B.	. Penalty Points	5
C.	. Team Penalty Point Accumulation	5
D	. Player and Staff Caution Accumulation	5
E.		
F.	Staff Dismissal Accumulation	6
6.	Team Conduct	
A	. Major Game Misconduct	6
B.	. Points of Emphasis	7
C.	. Game Official Assault or Abuse	7
D	. Coach/Staff Dismissal	7
E.	Player/Coach or Non-Player Behavior Before or After Caution, Send-off or Dismissal 2	8
F.	Disparaging Comments	8
G	. Intentional Throwing of Games	8
H	. Illegal Incentives for Winning a Game	8
I.	Gifts to Referees or Assistant Referees	
J.	Betting on Soccer Games	9
K	-	
L.		9
М	-	



N.	Zero-Tolerance Policy		
О.	Other Misconduct		
7. Appe	eals Process		30
А.	Red Card and Second Yellow Card Appeal		
B.	Suspension and Fine Appeal		
C.	Appeals to the Independent Disciplinary Panel		
Section III: T	eam Operations		
1. Stadi	ium and Playing Field		35
А.	Seating Capacity		
В.	Entrance		
C.	Press Box		
D.	Press Box Communications		
E.	Public Address (P.A.) System		
F.	Scoreboard and Game Clock		
G.	Playing Surface Guidelines		
H.	Field Dimensions		
I.	Field Markings		
J.	Goals		
К.	Lighting		
L.	Dressing Rooms		
М.	Showers		
2. Adva	anced Game Preparation		38
A.	Home Team Travel Information Sheet		
B.	Visiting Team Travel Information Sheet		
C.	Visiting Team Guide		
D.	Competition Timeline		
Е.	Training Facilities		
F.	Uniforms		
G.	Uniform Sets	40	
H.	Team Logo	40	
I.	Uniform Numbers	40	
J.	Player Numbers	40	
K.	Player Names	40	
L.	Uniform Selection	41	
М.	Goalkeeper Jersey	41	
N.	Replacement Jersey	41	
О.	Warm-Ups	41	
Р.	Sponsorships	41	
Q.	Practical Equity	42	



3. Centr	ralized Travel Program	
А.	Air Travel Policy	
B.	Air Travel Booking Deadline	
C.	Air Travel Considerations	
D.	Ground Transportation Policy	
Е.	Ground Transportation Billing	
F.	Ground Transportation Contracting	43
G.	Itineraries	43
Н.	Itinerary Submittal Violation	43
I.	Hotel Policy	43
J.	Hotel Billing	44
К.	Hotel Rooming Lists	45
L.	Rooming List Submittal Violation	45
М.	Hotel Menus	45
N.	Menu Submittal Violation	45
О.	Audio Visual	45
Р.	Hotel Confirmation	45
Q.	Multi-City Road Trips	45
R.	Opt-Outs	45
4. USL 5	Super League Final Playoff Travel Pool	
А.	Travel Pool Policy	45
B.	Emergency Travel Fund	46
5. Tech	nical Staff	
А.	Head Coach	46
В.	Assistant Coach	47
C.	Coaching Staff	47
6. Playe	er Medical Protocols	
А.	Head Injury Policy	47
B. He	ad Injury/Concussion Staffing and Communication	47
C. Er	ntry and Exit Physical	
Section IV: G	ame Day Operations	
1. Game	e Day Staff	
А.	Game Day Staff	
B.	Director of Operations	
C.	Home Team Liaison	
D.	Visiting Team Liaison	
Е.	Visiting Team Locker Room Attendant	
F.	Referee Liaison	
G.	Game Day Security Personnel	



	H.	Certified Athletic Trainer	51
	I.	Team Physician, Ambulance Crew, and Stretcher Crew	51
	J.	Ball Retrievers & Ball Retrievers' Supervisor	52
	K.	Public Address Announcer	52
	L.	Scorekeeper/Statisticians/Spotters	52
	M.	Timekeeper	52
	N.	Ticket Manager	52
	0.	Ticket Sellers	52
	P.	Ticket Takers	53
	Q.	Game Activities and Promotions	53
	R.	Hospitality	53
	S.	Live Stream Onsite Production Contact	53
	Т.	Match Observer	53
2.	Game Ope	erations	53
	A.	Complimentary Tickets	53
	B.	Game Day Contact with Visiting Team	53
	C.	Arrival to Stadium	54
	D.	Exchange of Official Game Day Lineup Sheet	54
	E.	Game Day Lineup/Removal of Injured Player	54
	F.	Minimum Number of Players Required	54
	G.	Dressing Room Preparation and Contents	54
	H.	Competition Timeline and Clock	55
	I.	Ice Baths	56
	J.	Ice, Water, and Electrolyte Beverage	56
	К.	League Policies	56
	L.	Team Warm-up Period	56
	M.	Warm-Up and Training Balls	56
	N.	Team and Referee Introductions	56
	О.	Locker Room Exit Times	57
	P.	Walk-Out Procedure	57
	Q.	Halftime Length	57
	R.	Second Half Preparation	57
	S.	Bench Personnel	57
	Т.	Field Level Personnel	58
	U.	Electronic Device Policy	58
	V.	Spiideo	58
	W.	Warm-Up Areas	58
	X.	End of the Game	58
3.	Game Pre	sentation	58



	A.	Game Day Presentation	58
	B.	Game Day Signage	
	C.	Field Evaluation	
	D.	Coaches and Reserve Players Entry	59
	E.	National Flag	
	F.	National Anthem	59
	G.	Bench Dress Code	60
	H.	Basic Compulsory Equipment (Including Undergarments)	60
	I.	PA Announcements	60
	J.	Background Sound	61
	K.	Sideline Presentation	61
4.	Equipmer	nt and Supplies	61
	A.	Match Ball	61
	B.	Warm-Up and Practice Balls	61
	C.	Athletic Trainer Supplies	61
	D.	Scorekeeper/Fourth Official's Table	61
	E.	Benches	61
5.	Security a	nd Safety Procedures	62
	A.	Security Staff Procedures	62
	B.	Law Enforcement Officers	62
	C.	Post-Game Security Form	62
	D.	Barriers	62
	E.	Prohibited Items	62
	F.	Metal Detection	62
	G.	Alcohol Management	63
	H.	Standard Response Guidelines	63
	I.	Supporters Groups	63
	J.	Travelling Supporters Groups	64
	К.	Signage and Tifo	64
	L.	Smoke Devices	65
	M.	Credentials	66
	N.	Approaching Game Officials	67
	О.	Entering the Field	67
	P.	Ejected Players & Coaches Removal Policy	67
	Q.	Forfeit and Non-Performance	67
	R.	Non-Performance by Either or Both Teams	68
	S.	Game Delays and Postponements	68
	Т.	Serious Incident Protocol	70
	U.	Cooling Breaks (Formerly Known as Hydration Breaks)	72



V.	Video Replay	
W.	Fan Code of Conduct	73
Х.	Stadium Announcements	74
6. Post	-Game Reporting	
А.	Game Statistics	74
В.	Minimum Standards Compliance Form	74
C.	Coach's Evaluation of Referees	74
D.	Game Day Ticket Report	74
Е.	Total Ticket Report	75
F.	Sales Rep Report	75
G.	Sponsorship/Corporate Sales Report	75
Н.	Merchandise Game Day Report	75
I.	Total Merchandise Report	75
Section V: R	egistration	
General	Registration Rules	
Regist	tration Procedure	
Online	e Player Registration	77
Player	r Announcements	77
Player Cl	lassifications	
Profes	ssional Player	77
Acade	my Players	77
Regist	tered Player	77
Eligibl	le Player	
Injure	d Reserve Player	
Ineligi	ible Player	
Illegal	l Player	
Intern	national Player	
Tamp	ering 80	
NCAA	80	
Master R	loster Regulations	80
23-Man (Game Day Player Pool	
Master R	loster Freeze	
Playoff E	ligibility	
Registra	tion Deadline	
Internati	ional Clearance Process	
ITC Prog	ression	
Releases		
Releas	ses; Generally	
Unilat	eral Termination by the Club for Just Cause	



Unilateral Termination by the Player for Just Cause	
Mutual Releases.	
Contract Dispute Resolution	
Loans	84
Generally. 84	
Loaning a Player Out	
Accepting a Player on Loan	
Loan Players in the Playoffs	
Active and Inactive Loans	
Player Transfers	85
Transfers to a Professional Team	
Transfers to an Amateur Team	
Academy Players	
Professional Contract Options	86
Professional Contract Options	
-	
Section VI: In-Season Operations Forms	87 88
Section VI: In-Season Operations Forms Section VII: Appendices	87
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards Appendix 3 – Betting and Integrity Policy	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards Appendix 3 – Betting and Integrity Policy Appendix 4 – Consumer Products Licensing Program Description	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards Appendix 3 – Betting and Integrity Policy Appendix 4 – Consumer Products Licensing Program Description Appendix 5 – Merchandise Buyers' Guide	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards Appendix 3 – Betting and Integrity Policy Appendix 4 – Consumer Products Licensing Program Description Appendix 5 – Merchandise Buyers' Guide Appendix 6 – Sport Betting Marketing Rules	
Section VI: In-Season Operations Forms Section VII: Appendices Appendix 1 – Uniform Design Requirements Appendix 2 – Advertising Standards Appendix 3 – Betting and Integrity Policy Appendix 4 – Consumer Products Licensing Program Description Appendix 5 – Merchandise Buyers' Guide Appendix 6 – Sport Betting Marketing Rules Appendix 7 – Trademark Requirements	



2023 USL Super League Competition Manual

Performance Introduction

This 2023 USL Super League Competition Manual (the "**Competition Manual**") includes certain rules and procedures that must be followed by teams participating in the League during the 2023 Season (each, a "**Team**").

Compliance with these rules and procedures is paramount to the successful operation of the League and its Teams. Failure to comply with the rules and procedures set forth in the Competition Manual will result discipline from the League. Please note that all fines and suspensions listed in this Competition Manual are minimums. Except where otherwise specified in this Manual, the League has discretion to render discipline beyond those minimums on a case-by-case basis.

The Competition Manual is a confidential document and is intended for the solely for the internal use of the League and its Teams. Under no circumstances should this Competition Manual be shared externally.

This is a draft accurate at the time of filing and is subject to change.



Table	of	Contents
1	•••	0 0 11 0 0 11 0 0

Introduct	ion1	L
Staff Dir	ectory2	2
Definitio	ns	3
Section I	: League Structure	5
1. A	lignment5	5
2. G	overning Bodies and Committees	5
А.	President	5
B.	League Directors	5
3. P	laying Rules	5
4. P	laying Season	5
5. L	eague Schedule	5
6. P	oint System	5
7. L	eague Standings/Tie-Breakers	5
8. S	ubstitutions	7
А.	Substitution for Exhibition Games	7
B.	Substitution Passes	7
C.	Substitution Number Set	7
D.	Goalkeeper Substitution	7
E.	Head Injury Substitution	7
9. E	xtra Time7	7
10.	Playoffs and USL Championship Final	3
11.	U.S. Open Cup)
12.	Exhibition and Scrimmage Matches)
13.	International Matches)
А.	Procedures to follow when hosting an international match / tournament)
B.	Procedures necessary when desiring to travel abroad including pre-season events9)
14.	Forfeit and Non-Performance)
А.	Forfeit10)
B.	Non-Performance by Both Teams10)
Section I	I: Registration	1
1. G	eneral Registration Rules	1
А.	Registration Procedure11	1
B.	Online Player Registration	1
C.	Player Announcements	1
2. P	layer Classifications	2



Professional Player	. 12
Academy Players	. 12
Registered Player	. 12
Eligible Player	. 12
Injured Reserve Player	. 13
Ineligible Player	. 13
Illegal Player	. 14
International Player	. 14
Tampering	. 15
NCAA	. 15
Master Roster Regulations	. 16
3-Man Game Day Player Pool	. 16
Master Roster Freeze	. 17
Ainimum Number of Players Required	. 18
Playoff Eligibility	. 18
Registration Deadline	. 18
nternational Clearance Process	. 18
Proof of Entry Prior to 10	. 18
First Registration	. 18
Minors Process	. 18
ITC Request	. 19
ITC Progression	. 19
Releases	. 19
Releases; Generally	. 19
Unilateral Termination by the Team for Just Cause.	. 19
Unilateral Termination by the Player for Just Cause.	. 20
Mutual Releases.	. 20
Contract Dispute Resolution.	. 20
Loans	. 20
Generally	. 20
Loaning a Player Out.	. 20
Accepting a Player on Loan	. 21
Loan Guidelines	. 21
Loan Players in the Playoffs	. 21
Active and Inactive Loans	. 21
Player Transfers	. 22
	3-Man Game Day Player Pool Master Roster Freeze Minimum Number of Players Required Playoff Eligibility Registration Deadline



COMPETITION MANUAL

А.	Transfers to a Professional Team	
B.	Transfers to an Amateur Team	
C.	Academy Players	
14. Prof	essional Contract Options	
А.	To Exercise an Option:	
B.	To Decline an Option:	
Section III: Fie	eld & Game Equipment	
1. USL C	hampionship Stadium Field Dimensions	
2. Playing	g Surface	
3. Field I	Dimensions	
4. Field N	/larkings	
А.	Bench & Technical Area	
B.	Penalty Area	
C.	Goals & Goal Area	
D.	Halfway Mark & and Center Circle	
E.	Corner Area & Flags	
F.	Field Marking	
G.	Commercial Advertising/Logos	
5. Equipr	nent	
А.	Goals	
B.	Match Ball	
C.	Warm-Up and Practice Balls	
D.	Ice, Water, and Electrolyte Beverage	
Ε.	Uniforms	
F.	Uniform Sets	
G.	Uniform Selection	
Н.	Goalkeeper Jersey	
I.	Replacement Jersey	
J.	Warm-Ups	
К.	Player Appearances and Travel Dress Code	
L.	Basic Compulsory Equipment (Including Undergarments)	
М.	Dressing Rooms	
N.	Showers	
О.	Ice Baths	
Р.	Dressing Room Preparation and Contents	
Section IV: Le	eague Policies and Gameday Procedures	



1.	Technica	l Staff	. 31
	A.	Head Coach	. 31
	B.	Assistant Coach	. 31
	C.	Coaching Staff	. 31
2.	Gameday	Policies	. 31
	А.	Electronic Device Policy	. 31
	B.	Smoking, Vaping & Tobacco Policy	. 31
	C.	Hydration Breaks	. 32
	D.	Approaching Game Officials	. 32
	E.	Entering the Field	. 33
	F.	Ejected Players & Coaches Removal Policy	. 33
3.	Advance	d Game Preparation	. 33
	A.	Training Facilities	. 33
	B.	Practical Equity	. 33
4.	Gameday	Presentation	. 34
	А.	Game Day Contact with Visiting Team	. 34
	B.	Arrival to Stadium	. 34
	C.	Competition Timeline	. 34
	D.	Field Evaluation	. 34
	E.	Exchange of Official Game Day Lineup Sheet	. 34
	F.	Team Warm-up Period	. 35
	G.	Warm-Up and Training Balls	. 35
	H.	Locker Room Exit Times	. 35
	I.	Coaches and Reserve Players Entry	. 36
	J.	Team and Referee Introductions	. 36
	К.	Putting a New Ball into Play	. 36
5.	Gameday	Personnel	. 36
	A.	Bench Personnel	. 36
	B.	Field Level Personnel	. 37
	C.	Bench Dress Code	. 37
6.	Player M	edical Protocols	. 37
	А.	Head Injury Policy	. 37
	B.	Entry and Exit Physical	. 37
7.	Halftime		. 37
	А.	Halftime Length	. 37
	B.	Second Half Preparation	. 38



8. Post-	Game Reporting	
А.	Game Statistics	
B.	Coach's Evaluation of Referees	
Section V: G	ame Delays & Rescheduling	
1. Game	e Delays and Postponements	
А.	Advanced Preparation	
B.	Communication Surrounding Delay or Postponement	
C.	Restart	
D.	Canceling/Postponing a Game Prior to Start	
E.	Authority to Delay or Postpone a Game	
F.	Grace Period Before Abandoning a Game	
G.	Game Cancellation / Postponement Criteria	
Н.	Referee "No Show"	
I.	Incomplete Game	
J.	Rescheduling a Postponed Game	
Κ.	Rescheduling Considerations	
L.	Stadium Clearances are Final	
М.	Cost from Delayed or Postponed Matches	
Section VI: N	Match Officials	
1. Playin	ng Rules	
2. Matel	h Officials	
А.	Referee	
В.	Assistant Referees	
C.	Fourth Official	
3. USL	Championship Match Officials	
4. Exhib	pition/Scrimmage Officials	
Section VII:	Discipline & Conduct	
1. Leagu	ue Disciplinary Power	
2. Authorithe	orities	
А.	The League Disciplinary Committee	
В.	Independent Disciplinary Panel	
3. Disci	plinary Process	
4. Suspe	ensions	
А.	The League's Discretionary Power on Suspensions	
B.	Serving Suspensions During the Regular Season	
C.	Exhibition Game Send-off	



D. Final Regular Season Game Send-off	
E. Suspension Parameters & Restrictions	
F. Extending a Suspension	
G. Suspension during the Playoffs	
H. Cross League and Competition Suspensions	
I. USSF Discipline for Serious Incidents	
5. Penalty Points and Accumulation	
A. League Disciplinary Point System	
B. Penalty Points	
C. Team Penalty Point Accumulation	
D. Player and Staff Caution Accumulation	
E. Good Behavior Incentive	
F. Staff Dismissal Accumulation	
6. Technical Staff Conduct	
A. Warning Offenses	
B. Cautionable Offenses	
C. Send-Off Classifications	
7. Team Conduct	
A. Cautionable Offenses	
B. Send-Off Classifications	
C. Major Game Misconduct	
D. Points of Emphasis	
E. Game Official Assault or Abuse	
F. Coach/Staff Dismissal	
G. Player/Coach or Non-Player Behavior Before or After Caution,	
H. Disparaging Comments	
I. Intentional Throwing of Games	
J. Illegal Incentives for Winning a Game	
K. Gifts to Referees or Assistant Referees	
L. Betting on Soccer Games	
M. Scandalous Conduct	
N. Moral Turpitude	
O. Vandalism/Destruction of Property	
P. Zero-Tolerance Policy	
Q. Other Misconduct	
 	



EXHIBIT I – ACKNOWLEDGMENT OF RECEIPT

[See attachment]

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If USL SUPER LEAGUE, LLC offers you a franchise, it must provide this disclosure document to you:

- (a) 14 calendar-days before you sign a binding agreement with, or make a payment to, USL SUPER LEAGUE, LLC or an affiliate in connection with the proposed franchise sale, or
- (b) Under New York and Rhode Island law, if applicable, at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship, or
- (c) Under Michigan, Oregon and Washington at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If USL SUPER LEAGUE, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency listed on Exhibit B.

The franchisor is USL SUPER LEAGUE, LLC, located at 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607. Its telephone number is (813) 963-3909.

Issuance date: February 15, 2023

The franchise seller is (check one): □ Jake Edwards □ Garrison Mason □ Alec Papadakis □ Justin Papadakis □ Amanda Vandervort

He/she is located at 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607. His/her telephone number is (813) 963-3909.

USL SUPER LEAGUE, LLC authorizes the respective state agencies identified on <u>Exhibit C</u> to receive service of process for it in the particular state.

I have received a Franchise Disclosure Document dated February 15, 2023 (the effective date in certain states is listed in the State Effective Date page) that included the following Exhibits:

- A. Franchise Agreement
- B. Director of State Administrators & Agencies
- C. Agent for Service of Process
- D. List of Current Franchisees
- E. List of Franchisees Who Have Left the System

Date

- F. Financial Statements
- G. State Addenda to Franchise Disclosure Document
- H. Manuals Table of Contents
- I. Receipt

Prospective Franchisee

Name of Signor and Title
(if a corporation or limited liability company)

Address:

Please sign this copy of the receipt, date your signature and keep for your records.

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If USL SUPER LEAGUE, LLC offers you a franchise, it must provide this disclosure document to you:

- (a) 14 calendar-days before you sign a binding agreement with, or make a payment to, USL SUPER LEAGUE, LLC or an affiliate in connection with the proposed franchise sale, or
- (b) Under New York and Rhode Island law, if applicable, at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship, or
- (c) Under Michigan, Oregon and Washington at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If USL SUPER LEAGUE, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency listed on Exhibit B.

The franchisor is USL SUPER LEAGUE, LLC, located at 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607. Its telephone number is (813) 963-3909.

Issuance date: February 15, 2023.

The franchise seller is (check one): \Box Jake Edwards \Box Garrison Mason \Box Alec Papadakis \Box Justin Papadakis \Box Amanda Vandervort

He/she is located at 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607. His/her telephone number is (813) 963-3909.

USL SUPER LEAGUE, LLC authorizes the respective state agencies identified on <u>Exhibit C</u> to receive service of process for it in the particular state.

I have received a Franchise Disclosure Document dated February 15, 2023 (the effective date in certain states is listed in the State Effective Date page) that included the following Exhibits:

- A. Franchise Agreement
- B. Director of State Administrators & Agencies
- C. Agent for Service of Process
- D. List of Current Franchisees
- E. List of Franchisees Who Have Left the System

Date

- F. Financial Statements
- G. State Addenda to Franchise Disclosure Document
- H. Manuals Table of Contents
- I. Receipt

Prospective Franchisee

]	Name of Signor and Title	
((if a corporation or limited liability c	ompany)

Address:

You may return the signed receipt either by signing, dating, and mailing it to USL SUPER LEAGUE, LLC, 1715 N. Westshore Blvd. Suite 825 Tampa, FL 33607, or by scanning and emailing a copy of the signed and dated receipt to legal@uslsoccer.com.